Essays on the roles of families, firms, location, and criminal records

Erika Forsberg



The Institute for Evaluation of Labour Market and Education Policy (IFAU) is a research institute under the Swedish Ministry of Employment, situated in Uppsala. IFAU's objective is to promote, support and carry out scientific evaluations. The assignment includes: the effects of labour market and educational policies, studies of the functioning of the labour market and the labour market effects of social insurance policies. IFAU shall also disseminate its results so that they become accessible to different interested parties in Sweden and abroad.

IFAU also provides funding for research projects within its areas of interest. The deadline for applications is October 1 each year. Since the researchers at IFAU are mainly economists, researchers from other disciplines are encouraged to apply for funding.

IFAU is run by a Director-General. The institute has a scientific council, consisting of a chairman, the Director-General and five other members. Among other things, the scientific council proposes a decision for the allocation of research grants. A reference group including representatives for employer organizations and trade unions, as well as the ministries and authorities concerned is also connected to the institute.

Postal address: P O Box 513, 751 20 Uppsala Visiting address: Kyrkogårdsgatan 6, Uppsala Phone: +46 18 471 70 70 Fax: +46 18 471 70 71 ifau@ifau.uu.se www.ifau.se

Dissertation presented at Uppsala University to be publicly examined in Lecture Hall 2 Ekonomikum, Kyrkogårdsgatan 10, Uppsala, Friday, 20 September 2024 at 13:15 for the degree of Doctor of Philosophy.

Essay III has been published by IFAU as working paper 2024:14 and Swedish report 2024:15.

ISSN 1651-4149

Economic Studies 217

Erika Forsberg Labor-market inequality

Department of Economics, Uppsala University

Visiting address:	Kyrkogårdsgatan 10, Uppsala, Sweden
Postal address:	Box 513, SE-751 20 Uppsala, Sweden
Telephone:	+46 18 471 00 00
Telefax:	+46 18 471 14 78
Internet:	http://www.nek.uu.se/

ECONOMICS AT UPPSALA UNIVERSITY

The Department of Economics at Uppsala University has a long history. The first chair in Economics in the Nordic countries was instituted at Uppsala University in 1741.

The main focus of research at the department has varied over the years but has typically been oriented towards policy-relevant applied economics, including both theoretical and empirical studies. The currently most active areas of research can be grouped into six categories:

- * Labour economics
- * Public economics
- * Macroeconomics
- * Microeconometrics
- * Environmental economics
- * Housing and urban economics

Erika Forsberg

Labor-market inequality

Essays on the roles of families, firms, location, and criminal records



UPPSALA UNIVERSITET Dissertation presented at Uppsala University to be publicly examined in Lecture Hall 2 Ekonomikum, Kyrkogårdsgatan 10, Uppsala, Friday, 20 September 2024 at 13:15 for the degree of Doctor of Philosophy. The examination will be conducted in English. Faculty examiner: Professor Matthew Lindquist (SOFI, Stockholms universitet).

Abstract

Forsberg, E. 2024. Labor-market inequality. Essays on the roles of families, firms, location, and criminal records. *Economic studies* 217. 218 pp. Uppsala: Department of Economics, Uppsala University. ISBN 978-91-506-3059-6.

Essay 1 Individuals working in larger labor markets tend to earn more than those working in smaller labor markets, but the reason for this is still unclear. This paper studies whether larger cities provide better occupational skill matches by combining machine learning techniques with data on individuals' productive skills matched with employer data to construct a novel measure of match quality. I show that occupational skill-match quality is higher for individuals living in large local labor markets. Conditional on skills, differences in match quality explain around 30 percent of the city-size wage gap. The higher match quality in larger labor markets is related to a more diversified occupation structure and more learning possibilities in these markets.

Essay 2 (with Martin Nybom and Jan Stuhler) To what extent does the sorting of workers across firms contribute to intergenerational persistence and why? We show that socioeconomic disparities in firm pay premia account for about one third of the intergenerational elasticity of income in Sweden. Firm pay gaps open already at career start, implying that children from more privileged backgrounds find more favorable entry points to the labor market. Their pay advantage widens further in their early careers as they climb the firm pay ladder faster, switch firms more frequently, and secure higher pay gains conditional on switching. Skill sorting explains most of the divergence over the career, but not the initial pay gaps at the career start.

Essay 3 (with Akib Khan and Olof Rosenqvist) Family background shapes outcomes across the life cycle. While the importance of family background varies across countries, less is known about heterogeneities across social groups within a country. Using Swedish data, we compare sibling correlations in skills, schooling, and earnings across fine-grained socioeconomic status (SES) groups. The result from the study shows that sibling correlations decline with parental socioeconomic status. This pattern holds for skills, schooling, and earnings.

Essay 4 (with Hans Grönqvist, Susan Niknami and Mårten Palme) We investigate the effect of being included in Sweden's first online criminal database, which facilitates anonymous and free name-based searches for individuals charged with a crime. Leveraging administrative rules that restricted the identification of individuals charged before specific dates, we estimate the effects by comparing outcomes of exposed and non-exposed individuals. We find significant adverse effects of exposure on earnings but not on employment or criminal recidivism. However, there are significantly stronger detrimental effects on both labor market outcomes and recidivism in defendant subgroups such as those with at least a high school degree, acquitted individuals, and those living in areas with a relatively low concentration of ex-criminals. Our results suggest that stigma is a potentially important but previously unappreciated mechanism explaining responses to criminal justice interactions.

Keywords: Inequality, match quality, local labor markets, intergenerational mobility, firms, criminal records

Erika Forsberg, Department of Economics, Box 513, Uppsala University, SE-75120 Uppsala, Sweden.

© Erika Forsberg 2024

ISSN 0283-7668 ISBN 978-91-506-3059-6 URN urn:nbn:se:uu:diva-535607 (http://urn.kb.se/resolve?urn=urn:nbn:se:uu:diva-535607)

To Anton and Otto

Acknowledgements

Completing this thesis would not have been possible without the support and encouragement of numerous individuals and institutions.

First, I want to express my gratitude to my main supervisor, Martin Nybom, for his guidance and support throughout this journey. His dedication to discussing any challenges I encountered has been invaluable. I also want to thank my secondary supervisors, Hans Grönqvist and Lena Hensvik, whose feedback and expertise have greatly helped me improve the thesis. During my PhD studies, I got the opportunity to work together on projects with Martin and Hans. That experience has given me valuable knowledge, which has been of great use for me to be able to start my own project.

I am also grateful to my other co-authors, Jan Stuhler, Susan Niknami, Mårten Palme, Akib Khan, and Olof Rosenqvist, for their collaboration and for sharing their expertise, which has enriched my understanding in numerous ways. I also want to thank Jan Stuhler, for inviting me to a short visit to Universidad Carlos III de Madrid.

I want to thank everybody who has taken the time to read and give feedback on my thesis. I especially want to thank the discussant of my licentiate seminar, Rasmus Landersø, the discussant on my final seminar, Raoul van Maarseveen, and the internal discussants Arizo Karimi and Georg Graetz, for their insightful comments and valuable feedback.

I am happy to have done the PhD at the Department of Economics at Uppsala University. During my studies there, I have met a friendly and supportive environment and encouragement to continue my work. I want to thank the faculty for fostering such an atmosphere and the administrative staff for making everything run so smoothly. Thanks to all fellow PhD students for meeting on lunches, walks, and fikas, and making the PhD studies an enjoyable experience. A special thanks to the Phd students in my cohort Rinni, Gabriella, Hanfeng, Akib, Daniel, Hiep, Qingyan, Lovisa, Malin and Zunyuan. I also want to give a special thanks to Rinni and Tram, for making my Phd studies such a memorable time.

I spent a large part of the PhD program at IFAU, where I received tremendous support and a welcoming atmosphere. My sincere thanks to all my colleagues, with special gratitude to my office mates at IFAU Henning, Olle and Zariab. I am also thankful to the administrative staff at Ifau, for all your help and support.

I also want to thank the participants and organizers of the Uppsala Labor Group and Uppsala Urban Lab. These groups have helped me advance my work, both by providing valuable feedback on my own work and by hearing about and being inspired by others' work.

Thanks to the Jan Wallander och Tom Hedelius foundation for funding my exchange to the Queen Mary University of London and to my hosts Sang Yoon (Tim) Lee and Anna Raute for inviting me. I am also very grateful to all the faculty and Phd students at Queen Mary University of London for making my exchange there an exciting and fun experience.

I also want to thank my friends outside academia. A special thanks to my friends My, Alice, Elin, Amanda, Frida, Caroline and Yash who enrich my free time and help me remember that there exists a life outside of work and academia as well. I want to thank my family, who always stand by my side and support me during my work. Thanks to my mother, Tina, who has always encouraged me to study, learn, and follow my dreams. Thanks to my sister Julia, who constantly challenges me and gets me to grow as a person. To my mormor Anne-Liss, morfar Åke, farmor Tellervu, I am grateful to have you as my grandparents and for all you have done for me during my childhood and today. I am also thankful for the support and love from my aunts, uncles, and cousins. I also want to thank Olle and Lisa and my boyfriend's family, his parents Maria and Per-Olof, and his siblings Jonas and Hedvig with their families, for their unwavering support.

Finally, thanks to my boyfriend Anton. For being there and supporting me during my thesis work and for enriching my life in numerous ways. I dedicate this thesis to Anton and my son Otto, who bring me boundless joy and inspiration.

> Uppsala, August 2024 Erika Forsberg

Contents

Introdu	ction	11
Ref	erences	16
Essav 1	· Labor market size and multidimensional skill-mismatch	19
1	Introduction	20
2	Conceptual framework	
3	Data	25
4	Empirical approach	27
5	Wage differences and labor market size	30
6	Results	31
7	Mechanisms	39
8	Robustness	46
9	Conclusion	50
Ref	erences	52
App	pendix A: Additional figures and tables	56
Essay 2	2: Labor-market Drivers of Intergenerational Earnings Persistenc	e 65
1	Data and specifications	71
2	The contribution of firms to intergenerational persistence	74
3	Firm pay gradients over the lifecycle	80
4	The role of sorting across firms by skill	92
5	Do firm pay reflect compensating differentials?	97
6	Conclusions	103
Ref	erences	105
App	pendix A: Additional figures and tables	109
Essay 3	B: Do sibling correlations in skills, schooling, and earnings vary	
by soci	oeconomic background? Insights from Sweden	127
1	Introduction	128
2	Conceptual framework	132
3	Data	134
4	Estimation	136
5	Main results: sibling correlations by parental earnings	137
6	Robustness: sibling correlations by parental education	139
7	Discussion	142
8	Conclusion	143
Ref	erences	146

Appendix A: Additional figures and tables		
Essay 4: Making Background Salient: The Effects of Open Access		
Criminal Databases on Offender Behavior	163	
1 Introduction	164	
2 Background	169	
3 Data and sample construction	175	
4 Research design	176	
5 Results	177	
6 Concluding Remarks	189	
References	191	
Appendix A: Additional figures and tables	197	
Appendix B: Data	218	

Introduction

It is a well-documented fact that differences in human capital are one explanation for wage differences, where individuals with more education and skills have been shown to get higher wages (see, for example Card, 1999; Lindqvist and Vestman, 2011). However, individual outcomes in the labor market may not solely depend on their skills. Instead, it is possible that equally productive workers end up having different labor market outcomes due to imperfections in labor markets and the specific labor market conditions they encounter. The labor market could affect individuals in different ways and through different channels, and this thesis explores various dimensions of how the labor market influences individuals' labor market outcomes.

One way in which the labor market might affect individuals is that different individuals meet different geographical labor markets. Different geographical labor markets might provide different labor market conditions and career opportunities. If individuals are perfectly mobile, we might expect them to move to the labor market with the best opportunities. However, in practice, that might not be the case. Individuals are likely to have preferences of where to live (see for example Moretti, 2011). For example, they may prefer to live close to their friends and family and stay in the city where they were born. Thus, individuals with similar skills who are born in different places might face different labor markets and thus might have different labor market opportunities.

Geographical labor markets might differ in several dimensions. One dimension of the labor markets that has been shown to be important is the size of the labor market, where individuals who work in larger labor markets have been shown to, on average, earn more than individuals who work in smaller labor markets (see for example Rosenthal and Strange, 2008; Papageorgiou, 2022; Eliasson and Westerlund, 2022). Chapter 1 examines more about how geographical labor markets affect labor market possibilities by studying how labor market size is related to match quality on the labor market, where match quality is high if individuals work in an occupation where their skills are valuable.

In a perfectly competitive labor market, workers who lose their jobs can, without cost, find an identical job (Manning, 2011). However, in practice, this might not be the case. Jobs vary in attributes, and workers may find themselves better off in one firm compared to another. One dimension of how firms differ from each other is how much they pay their workers. Research has shown that some firms pay their workers more, while others pay less (Abowd et al., 1999).

Thus, if workers are able to enter different types of firms, their earnings might be affected, even if the worker possesses the same human capital. Differences in firm pay have been shown to be an important factor for inequality (see for example Card et al., 2013).

In chapter 2, we examine the role of firms, where we look at the role of firms to explain persistence in income across generations. That is, we study if the fact that children from high-income families earn more themselves is explained by the fact that they are more likely to sort into high-paying firms. However, the role of firms might not be unrelated to the role of human capital. It might be the case that individuals with higher human capital are able to enter better-paying firms. In Chapter 2, we therefore also examine how the sorting of children from high-income families to better-paying firms is related to skills.

While research has shown that family background shapes individuals' outcomes, it is possible that family background does not matter equally everywhere. In Chapter 3, we examine how the importance of family background, measured as siblings' correlations, varies depending on family background.

In a world with perfect information, the employees could observe all relevant information about the individual and choose the individual best suited for the job. However, in practice, this might not be the case. There might be information frictions where not all information is available to the employee, or there is a cost of obtaining the information (see Stigler, 1962 for an early discussion about this). Thus, in the case of imperfect information, what information is available in the labor market might affect the worker's job possibilities.

With technological change, information about the worker that was previously unavailable might suddenly become available (see for example Autor, 2001). Depending on what information becomes more available, the worker might either benefit or be harmed by its viability. Chapter 4 studies more about the effect of new information becoming available by studying how the introduction of an online database, which reveals the criminal history of the individuals, affects their labor market possibilities.

Thus, the labor market conditions the individual faces might affect their career opportunities. Different dimensions of the labor market, such as geographical labor markets, firms, and information access, might matter for individuals' labor market outcomes. Below is a short summary of the different chapters in the thesis.

Essay 1: Labor market size and multidimensional skill-mismatch

While earlier research has provided evidence that individuals who work in larger labor markets earn more than individuals who work in smaller labor markets (see, for example Rosenthal and Strange, 2008; Papageorgiou, 2022; Eliasson and Westerlund, 2022), the mechanism behind the higher wage pre-

mium in larger labor markets still remain unclear. In Chapter 1, I examine one mechanism behind the city-size wage premium by studying whether larger cities provide better occupational skill matches.

To study how match quality differs depending on city size, I construct a match quality measure using Swedish data on eight different types of skills from the military enlistment test. The tests include both cognitive (inductive, verbal, spatial, and technical ability) and non-cognitive skills (social maturity, intensity, psychological energy, and emotional stability). To measure how useful different skills are in different occupations, I estimate the return to skills in different occupations. Instead of assuming any functional form I estimate the return to skills using a random forest. I also use a random forest to estimate the return to skills on the whole market. The match quality measure is then constructed as the estimated return to skills in the occupation where the individual works minus the market return to skills. Thus, the intuition behind the match quality measure is that individuals are well-matched if they work in occupations with high returns to their skills compared to what they could receive on average.

The results show that match quality is higher in larger labor markets than in smaller ones. Conditional on skills, the difference in match quality explains around 30 percent of the city-size wage gap. The city-size match-quality gap is small for young individuals but increases with age. This result is consistent with initial uncertainty about the optimal match that decreases with experience (see Guvenen et al., 2020; Fredriksson et al., 2018). The increase in match quality over the life-cycle seems to come from both more frequent, and better, occupation switches in larger labor markets. When examining mechanisms for higher match quality in larger labor markets, the analysis suggests that the higher match quality in larger labor markets is driven by both more occupation diversity and by more learning possibilities in larger labor markets.

It should be noted that the finding that match quality is higher in larger labor markets does not necessarily mean individuals would be better off by moving to large labor markets since the utility of the individual is also affected by the cost of living and their preferences about where to live. Thus, while the result here highlights the limited ability to find a good match in smaller labor markets, research is needed on how to best solve this problem and increase match quality in smaller labor markets.

Essay 2: Labor-market Drivers of Intergenerational Earnings Persistence (with Martin Nybom and Jan Stuhler)

Children from high-income families, on average, earn more than children from low-income families. Most of the literature that examines the mechanism behind the persistence in earnings across generations has focused on the role of human capital (see for example Becker and Tomes, 1979; Solon, 2004; Heck-

man and Mosso, 2014).¹ In Chapter 2, we instead focus on the role of labor market factors, and more precisely, firms, to explain the persistence of income between generations.

We show that sorting to higher-paying firms explains an important part of the persistence in earnings across generations, explaining almost 30 percent of the intergenerational earnings correlation. The contribution of firms rises to 38% if we incorporate differences in returns to experience across firm types in our analysis.

However, part of the sorting to higher-paying firms is still related to skill sorting. Using data on the estimated individual fixed effects, education, and skills, we show that around half of the SES gradient in firm premia is explained by skill sorting. However, even conditional on skills, 50 percent of the SES gradient in skill premia remains, indicating that investments in human capital do not seem to be enough to equalize income for children from families of different incomes.

Moreover, we study the life cycle dynamics of the firm premia depending on family background. The result in Chapter 2 shows that children from highincome families start off at better-paying firms already career start, and the difference in firm premiums continues to widen at the beginning of the career. While most of the widening in skill premium can be explained by skill sorting, most of the initial gap in firm premia cannot be explained by skill sorting. The results indicate that while human capital seems to be an important explanation for why children from high-income families climb further in the job ladder, coming from families with high income might give better starting positions in the career for reasons unrelated to the skill of the child.

Essay 3: Do sibling correlations in skills, schooling, and earnings vary by socioeconomic background? Insights from Sweden (with Akib Khan and Olof Rosenqvist)

In Chapter 3, we examine how the importance of family background, measured as siblings' correlations, varies depending on family background. The sibling correlation shows the correlation in outcome between a pair of siblings. We use Swedish register data to compare sibling correlations in skills, schooling, and earnings across fine-grained groups defined by parental socioeconomic status.

The result from the study shows that sibling correlations decline with parental socioeconomic status. This pattern holds for skills, schooling, and earnings.

For the sibling correlation in income and education, the decline is driven by an increase in within-family variation by parental SES, indicating that siblings become less similar to each other in terms of income and education in families

¹However, see Dobbin and Zohar (2023) and Engzell and Wilmers (2024) for some notable expectations of studies that also focus on labor market drivers for intergenerational mobility.

with higher socioeconomic status. In contrast, the decline in sibling correlations of skills is driven by a decline in between-family variation. Thus, the increase in within-family variation, in income and education, by parents' socioeconomic background does not seem to be driven by differences in skills. A potential explanation is that high-ability children from low SES families cannot reach their full potential in terms of earnings and educational attainment but instead end up closer to their lower-ability siblings (as also suggested by Papageorge and Thom, 2020 and Ronda et al., 2022).

Essay 4: Making Background Salient: The Effects of Open Access Criminal Databases on Offender Behavior (with Hans Grönqvist, Susan Niknami and Mårten Palme)

Technological advances have led to the creation of large online criminal databases that are easily accessible to the public. However, the impact of this accessibility on the offenders in these databases remains unclear. In chapter 4, we investigate the effect of being included in Sweden's first online criminal database, which facilitates anonymous and free name-based searches for individuals charged with a crime.

To estimate the causal effect of being included in the database, we use the fact that legally, courts were only obliged to provide criminal charges five years back in time, giving a cutoff date for when the company was able to get access to courts. This allows us to compare outcomes between two groups: those charged after the specified cutoff date, whose criminal records are consequently exposed online, and those of non-exposed offenders who were charged prior to the cutoff date and consequently do not have their criminal history online. We compare outcomes for these two groups using a difference-in-difference design.

The result from the study shows that being included in the database has negative effects on earnings but not on employment or criminal recidivism. However, there are significantly stronger detrimental effects on both labor market outcomes and recidivism in defendant subgroups such as those with at least a high school degree, acquitted individuals, and those living in areas with a relatively low concentration of ex-criminals. Our results suggest that stigma is a potentially important but previously unappreciated mechanism explaining responses to criminal justice interactions.

References

- Abowd, John M., Francis Kramarz, and David N. Margolis (1999). "High Wage Workers and High Wage Firms". *Econometrica* 67.2, 251–333.
- Autor, David H. (Mar. 2001). "Wiring the Labor Market". Journal of Economic Perspectives 15.1, 25–40.
- Becker, Gary S. and Nigel Tomes (1979). "An Equilibrium Theory of the Distribution of Income and Intergenerational Mobility". *Journal of Political Economy* 87.6, 1153–1189.
- Card, David (1999). "Chapter 30 The Causal Effect of Education on Earnings". Ed. by Orley C. Ashenfelter and David Card. Vol. 3. Handbook of Labor Economics. Elsevier, 1801–1863.
- Card, David, Jörg Heining, and Patrick Kline (May 2013). "Workplace Heterogeneity and the Rise of West German Wage Inequality*". *The Quarterly Journal of Economics* 128.3, 967–1015. eprint: https://academic.oup.com/ qje/article-pdf/128/3/967/30630708/qjt006.pdf.
- Dobbin, Caue and Tom Zohar (2023). "Quantifying the Role of Firms in Intergenerational Mobility". Unpublished manuscript.
- Eliasson, Kent and Olle Westerlund (Oct. 2022). "The urban wage premium and spatial sorting on observed and unobserved ability". *Journal of Economic Geography* 23.3, 601–627. eprint: https://academic.oup.com/joeg/article-pdf/23/3/601/50408541/lbac029.pdf.
- Engzell, Per and Nathan Wilmers (2024). "Firms and the Intergenerational Transmission of Labor Market Advantage".
- Fredriksson, Peter, Lena Hensvik, and Oskar Nordström Skans (Nov. 2018). "Mismatch of Talent: Evidence on Match Quality, Entry Wages, and Job Mobility". *American Economic Review* 108.11, 3303–38.
- Guvenen, Fatih, Burhan Kuruscu, Satoshi Tanaka, and David Wiczer (Jan. 2020). "Multidimensional Skill Mismatch". *American Economic Journal: Macroeconomics* 12.1, 210–44.
- Heckman, James J and Stefano Mosso (2014). "The Economics of Human Development and Social Mobility". *Annual Review of Economics* 6, 689–733.
- Lindqvist, Erik and Roine Vestman (Jan. 2011). "The Labor Market Returns to Cognitive and Noncognitive Ability: Evidence from the Swedish Enlistment". *American Economic Journal: Applied Economics* 3.1, 101–28.
- Manning, Alan (2011). "Chapter 11 Imperfect Competition in the Labor Market". Ed. by David Card and Orley Ashenfelter. Vol. 4. Handbook of Labor Economics. Elsevier, 973–1041.

- Moretti, Enrico (2011). "Chapter 14 Local Labor Markets". Ed. by David Card and Orley Ashenfelter. Vol. 4. Handbook of Labor Economics. Elsevier, 1237–1313.
- Papageorge, N. W. and K. Thom (2020). "Genes, education, and labor market outcomes: evidence from the health and retirement study". *Journal of the European Economic Association* 18.3, 1351–1399.
- Papageorgiou, Theodore (July 2022). "Occupational Matching and Cities". *American Economic Journal: Macroeconomics* 14.3, 82–132.
- Ronda, V., E. Agerbo, D. Bleses, P. B. Mortensen, A. Børglum, O. Mors, et al. (2022). "Family disadvantage, gender, and the returns to genetic human capital". *The Scandinavian Journal of Economics* 124.2, 550–578.
- Rosenthal, Stuart S. and William C. Strange (2008). "The attenuation of human capital spillovers". *Journal of Urban Economics* 64.2, 373–389.
- Solon, Gary (2004). "A Model of Intergenerational Mobility Variation over Time and Place". Generational Income Mobility in North America and Europe, 38–47.
- Stigler, George J. (1962). "Information in the Labor Market". *Journal of Political Economy* 70.5, 94–105.

Essay 1: Labor market size and multidimensional skill-mismatch

Acknowledgements: I thank my supervisors, Martin Nybom, Lena Hensvik, and Hans Grönqvist, for their valuable comments. This paper has also benefited from discussions with Jorge De la Roca, Enrico Moretti, Mitchell Downey, Raoul van Maarseveen, Georg Graetz, Anna Raute, Simon Franklin, Marco Manacorda, Sang Yoon Tim Lee, Felipe Gonzalez, Rafael Lalive, Tom Zohar, Oskar Nordström Skans, Luca Repetto, Peter Fredriksson, Matz Dahlberg, Adam Gill, Federica Meluzzi, Petter Berg and Michael Simmons. I have also received valuable comments from participants at presentations at: the poster session at HCEO-FAIR 2022 Summer School on Socioeconomic Inequality Bergen, seminar at Queen Mary University of London, Uppsala brown bag, Uppsala labor group, the Uppsala Urban lab workshop, the 5th Queen Mary University of London Economic and Finance workshop, seminar at Linnaeus University and seminar at Umeå University.

1 Introduction

A large empirical literature has shown that individuals who work in larger labor markets earn more than individuals who work in smaller labor markets (see for example Rosenthal and Strange, 2008; Papageorgiou, 2022; Eliasson and Westerlund, 2022). This city-size wage premium should partly be explained by productivity differences (see for example Glaeser and Maré, 2001), otherwise firms in the tradable sector would increase profits by relocating to smaller areas where both wage and land costs are lower (Moretti, 2011). From the workers' perspective, productivity differences could remain due to mobility frictions, such as higher cost of living in larger cities or individuals having preferences to reside in certain areas.

To be able to address geographical inequalities, it is important to understand the mechanism behind the city-size wage differential. Theoretically, competing mechanisms such as increased opportunities for individuals to augment human capital and higher match quality in larger labor markets have been suggested as explanations for the wage difference (see for example Puga, 2010). While research has provided evidence on learning and city size (see for example De La Roca and Puga, 2016), the empirical evidence on how match quality differs between large and small labor markets is still limited. The main reason is the challenges involved in measuring match quality. Most of the existing literature instead relies on indirect evidence. For instance, Wheeler (2008), Bleakley and Lin (2012), and Korpi and Clark (2019), show job change patterns that are consistent with higher match quality in larger cities.

In this study, I use rich population-based data to construct a novel measure of occupational skill-match quality that allows me to provide direct evidence of the importance of this mechanism in explaining the city-size wage differential. The measure draws from advances in the literature on multidimensional mismatch (see Guvenen et al., 2020; Fredriksson et al., 2018), which highlights that different combinations of skills might be useful in different jobs. To construct the match quality measure, I use Swedish data on eight different types of skills from the military enlistment test. The tests are conducted at age 18 or 19 and include both cognitive (inductive, verbal, spatial, and technical ability) and non-cognitive skills (social maturity, intensity, psychological energy, and emotional stability). As a proxy for skill requirements in each occupation, I estimate the return to skills in different occupations. Because theory gives little guidance on how to model the relationship between skills and wages, I pursue a non-parametric approach to estimation. Specifically, I use recent advancements in machine learning to improve on the earlier match quality measure and estimate a random forest model trained on tenured workers with data on the eight skills.¹ Analogously, I use a random forest to estimate the return to skills on the whole market. The match quality measure is

¹See recent work by Almgren et al. (2022) who also use a random forest to proxy for skill requirements in occupations.

then constructed as the estimated return to skills in the occupation where the individual works, minus the market return to skills. Thus, individuals are wellmatched if they work in occupations with high returns to their skills compared to what they could receive on average based on their skills.

My results show that match quality is higher in larger compared to smaller labor markets. This result holds conditional on skills and in a subset of workers who move across labor markets of different sizes. Conditional on skills, the difference in match quality explains around 30 percent of the city-size wage gap. The magnitude is in line with the result of Papageorgiou (2022), who uses a calibrated model and shows that occupational match quality explains around 35 percent of the city-size wage premium in the United States. The difference in match quality between individuals in larger and smaller labor markets is especially large for high-skilled workers, measured either by their cognitive skills, non-cognitive skills, or education. This is well in line with the fact that the city-size wage premium is higher for highly skilled individuals (see Bacolod et al., 2009).

The city-size match-quality gap is small for young individuals but increases with age. This result is consistent with initial uncertainty about the optimal match that decreases with experience (see Guvenen et al., 2020; Fredriksson et al., 2018). The increase in match quality over the life-cycle seems to come from both more frequent, and better, occupation switches in larger labor markets. The finding of an increase in match quality over the life-cycle is also in line with the finding by Eckert et al. (2022). The authors find that refugees placed in a large labor market initially have similar wages as refugees placed in smaller labor markets and experience faster wage growth through sorting to better (urban) jobs.

I examine two mechanisms behind the higher match quality in larger cities: occupation diversity and learning possibilities. If a more diverse set of occupations exists in large labor markets, this could give individuals more occupations to choose from and, therefore, could increase the likelihood that individuals find a good match (see Papageorgiou, 2022). Theoretically, higher learning possibilities in larger cities could come from a lower cost of switching occupations in larger cities (Wheeler, 2008). This allows individuals to explore different occupations and learn about what their optimal match is. Moreover, working in larger cities gives workers more valuable knowledge (Glaeser and Maré, 2001; De La Roca and Puga, 2016), which in some cases might be needed to be able to enter high-match quality occupations.

The analysis gives suggestive evidence that both of these mechanisms are at work. By constructing a Herfindahl-Hirschman Index (HHI) of occupation concentration in the labor markets, I show that larger labor markets have a more diversified occupation structure. Moreover, the difference in match quality in large and small labor markets partly seems to be explained by the difference in occupation diversity. Following the method in De La Roca and Puga (2016), I also show that experience obtained in larger labor markets is more valuable for future match quality than experience obtained in smaller labor markets. In addition, prior experience from large labor markets is valuable for individuals who work in small labor markets, suggesting that learning might be an important mechanism for higher match quality in larger labor markets.

This paper is related to several strands of literature. First, this paper is related to the strand of research that employs AKM models (see Abowd et al., 1999) to investigate mismatch between individuals and firms in relation to city size. These studies estimate assortative matching based on one-dimensional skills and tend to find that larger cities have a higher degree of assortative matching (see for example Dauth et al., 2022; Andersson et al., 2007; Leknes et al., 2022; Card et al., 2021).² While these studies have focused on the mismatch between workers and firms, I shift the focus to occupation mismatch. Occupational match quality plays a vital role in explaining workers' earnings (Guvenen et al., 2020). Consequently, investigating the role of occupation match quality in explaining the city-size wage premium is crucial. Furthermore, while prior studies employing AKM models have examined matching in terms of one-dimensional skills, my research utilizes data on eight distinct types of skills and estimates a multidimensional match quality measure. The match quality measure allows for both horizontal dimensions of mismatch, in terms of workers being overqualified and underqualified for their jobs, and vertical dimensions of mismatch, in terms of workers having the wrong set of skills for their jobs.

Secondly, this paper complements the literature exploring the mismatch between individuals' education and the educational requirements of their jobs, which tends to find higher education mismatch in smaller cities (see Abel and Deitz, 2015; Berlingieri, 2018; Boualam, 2014; Koster and Ozgen, 2021). Moreover, smaller cities seem to have more mismatch in terms of earlier industry experience (Harmon, 2013). However, using survey questions Andini et al. (2013) finds small effects of density on match quality. In contemporaneous work, Moretti and Yi (2024) show that after displacement, workers in larger labor markets are more likely to find a job with higher match quality, as indicated by the fact that they are more likely to work in an industry relevant to their college major, be employed in the same industry as before, and that the new job is likely to last longer.

In contrast to earlier studies, this paper looks at mismatch in terms of skills that are measured at age 18 or 19, when the individuals are unlikely to have entered the labor market. Unlike education choices, which may be influenced by local labor market conditions, pre-labor market skills are unlikely to be shaped by local labor market conditions. Consequently, this study enriches

²In contrast Mion and Naticchioni (2009) find a negative association between market size and assortative matching in Italy and Figueiredo et al. (2013) find limited evidence on more associative matching with more firm clustering within the same industry in Portugal.

the existing literature by shedding light on match quality of pre-determined skills, and suitable occupations in markets of different sizes. Moreover, I provide novel evidence on the life-cycle dynamics of match quality in relation to labor market size and direct evidence on how the city-size match quality gap is related to learning possibilities and occupation diversity.

The paper is structured as follows: Section 2 presents the conceptual framework and section 3 describes the data. In section 4, the empirical approach is explained, and section 5 presents background about how wage difference and labor market size look in the Swedish setting. The main results are presented in section 6 and the mechanisms behind the higher match quality in larger labor markets are examined in section 7. Section 8 explores the robustness of the results and the conclusions are discussed in section 9.

2 Conceptual framework

This section presents a simple framework to illustrate the idea behind the difference in match quality between large and small labor markets. Before entering the labor market, the individual has a set of different skills that are useful in different occupations. The more similar the skills obtained by the individual are to the skills valued in the occupation, the higher the match quality. Higher match quality increases productivity, which is shared with the worker in terms of higher wages.

In a world without frictions, individuals would move to labor markets with higher match quality until the match quality between different labor markets equalizes. However, in the presence of mobility frictions, this might not happen. To get the idea behind the mobility friction, consider a utility function similar to Moretti (2011).

$$U_{iL} = W(MQ, X)_{iL} - r_L + A_L + e_{iL}$$

The utility for individual (*i*) living in the local labor market (*L*) depends on the wage (*W*), the cost of living (*r*), amenities (*A*) and idiosyncratic preferences e_{iL} . The idiosyncratic preferences e_{iL} show how much the individual values the labor market holding the wage and amenity constant. For example, Moretti (2011) claims that being born in a city or having relatives in the city might give a higher value of e_{iL} . If e_{iL} is high, regional mobility will be low. Yagan (2019) shows that individuals do not seem to migrate to new places to any great extent as a response to economic shocks, indicating that mobility frictions might be high in practice.

Match quality (MQ) is assumed to have an impact on utility by affecting the wage. Productive skills also depend on a vector of other worker characteristics X, which, for example, can be the age of the worker. If workers get direct utility from working in an occupation that matches their skill set, match quality can also enter the utility function through the amenity. If idiosyncratic preferences e_{iL} are high, individuals might not move even in the case of large differences in real wages and amenities. Thus, the implication from the equation above is that in the presence of mobility frictions, match quality might differ between different labor markets.

In the presence of mobility frictions, the possibility of forming good matches in the labor market has been theorized to be higher in large labor markets (Puga, 2010). There are at least three mechanisms for why match quality might be higher in larger labor markets: occupation diversity, frictions, and learning possibilities.

2.1 Occupation diversity

Larger labor markets have been shown to have a more diversified occupation structure in the sense that more different occupations exist in large labor markets (Papageorgiou, 2022; Korpi, 2007), where especially scarce specialist occupations are more common in larger cities (Duranton and Jayet, 2011). A more diversified occupation structure gives the individual more occupations to choose from which could increase the likelihood that individuals find the optimal match given their skills.

2.2 Search frictions

The cost of switching jobs and occupations has been theorized to be lower in larger cities since larger cities have more job openings (Puga, 2010; Bleakley and Lin, 2012). With more job openings, individuals could apply to many jobs at the same time and be able to change jobs when they like instead of waiting for the job opening. Thus, lower search frictions speed up the matching process.

2.3 Learning possibilities

Earlier research about match quality has shown that work experience and job switches are important for learning about the optimal match (Guvenen et al., 2020; Fredriksson et al., 2018). A lower cost of switching occupations in larger labor markets will allow the individual to change occupations more and thus explore more different occupations (Wheeler, 2008). Thus, less search frictions in larger labor markets might give individuals the opportunity to learn about what occupations are a good match for them given their skills. Moreover, working in larger cities has been shown to provide larger human capital learning possibilities compared to working in a smaller city (Glaeser and Maré, 2001; De La Roca and Puga, 2016). Thus, experience from working in larger cities might give workers valuable knowledge, which might open up

possibilities to enter some new occupations with higher match quality. Finally, larger labor markets have more education possibilities (Frenette, 2006). For some occupations, a university education might be necessary for entering the occupation. Thus, more learning through university education could open doors to be able to enter more occupations.

Taken together, from theory, we could expect higher match quality in larger labor markets, driven by more occupation diversity in larger cities, lower search frictions, and more learning possibilities. To implement policies to increase match quality in small labor markets, it is vital to know why match quality differs between labor markets of different sizes. Thus, section 7 tries to separate between the mechanisms. If occupation diversity drives the result, controlling for a measure of occupation diversity is expected to reduce the estimated city-size match quality gap. If learning possibilities in larger labor markets are driving the result, we would expect that the gain in match quality from working in a large labor market should not be immediate; instead, match quality is expected to increase over time when working in a large compared to a small labor market.

3 Data

I use administrative wage data collected by Statistics Sweden. Occupation data are available for a large part of the Swedish population, covering almost 50 percent of all private and public sector workers.³ Skills are measured with scores from tests during the military enlistment for those who enlisted between the years 1969 and 1994 when almost all males participated in the military drafting. The enlistment is done at age 18 or 19. Since military enlistment was only mandatory for males, the data is limited to only include males.

The cognitive measures include four cognitive skills: inductive, verbal, spatial, and technical ability. The non-cognitive measures also capture the four skills: social maturity, intensity, psychological energy, and emotional stability. The evaluation of the non-cognitive score is done by a psychologist in a 20-minute interview (see Mood et al., 2012); before the interview, the psychologist has access to a form where the tested has answered questions about friends and family. In some years, the cognitive tests are graded on a scale of 0-25, and in others years, a scale of 0-40. The non-cognitive scores are measured on a scale of 1-5. To make the test scores comparable, all test scores have been standardized within each year of enlistment to have a mean of 0 and a standard deviation of 1.

³The wage and occupation information is collected during a measurement week (in September–November) each year. To be included in the sampling, the individual needs to be employed for at least one hour during the sampling week. Sampling is stratified by firm size and industry, and small firms in the private sector are underrepresented.

Occupation data is used on the three-digit level for the years 1996-2013, with the ssyk96 definition, in turn based on the international isco88. Some individuals have multiple occupation observations in a given year (around 4 percent of the sample). For these individuals, only the main occupation is used, where the main occupation is defined as the occupation with the highest wage expressed in full-time equivalent wages. In the sample, there exist 111 occupations. ⁴ Wage is measured as full-time equivalent wages. Furthermore, the data is linked to data on education, firms, and the municipality the individual lives in.

To define labor markets, I rely on Statistics Sweden's definition of local labor markets defined from commuting patterns. The local labor market definition 2013 is used, and thus, municipalities are assumed to belong to the same local labor market during the whole sample period. This definition gives 73 local labor markets. In the main specification, the log of the number of inhabitants in the local labor market is used to measure labor market size. In some cases, a categorical definition of the local labor market is used. The labor market is then divided into three size categories. Large labor markets are defined as labor markets with more than 500,000 inhabitants, consisting of three labor markets are defined as labor markets are defined as labor markets are defined as labor markets with more than 100,000 and less than 500,000 inhabitants, consisting of 20 local labor markets. The small labor markets are labor markets. The number of inhabitants used is the number of inhabitants in the local labor markets with less than 100,000 inhabitants, consisting of 50 local labor markets. The number of inhabitants used is the number of inhabitants in the local labor market in 2013, defined by SCB.

To be able to use the military and occupation data, the sample is restricted to men born between 1951 and 1976, with non-missing occupation data from 1996 to 2013. The sample is then restricted to those with data on the military test scores, which is available for 80 percent of the sample. Of this sample, 17 percent miss test scores data on any of the cognitive or non-cognitive sub-tests. The sample further excludes those individuals. To train the random forest, 20 percent of all tenured individuals are used, where tenured are defined as individuals who have worked in the occupation for at least three years. The test data consists of the remaining 80 percent of the tenured individuals and all individuals who never have tenure. The results in the paper use the individuals in the test data.

Table 1 summarizes the data by labor market size. As can be seen in the table, even if most labor markets are small, the number of observations is larger in large labor markets. Thus, a large part of the population lives in large labor markets. In panel B, population characteristics are shown divided by labor market size. The population in large labor markets is slightly younger.

⁴In total, there exist 113 occupations at the three-digit level. However, two occupations are excluded because they have less than one tenured worker (the excluded occupations are photo models and street market salespersons).

Cognitive skills, non-cognitive skills, and education levels are higher in larger labor markets, in line with individuals with high skills receiving a higher citysize premium (Bacolod et al., 2009; Andersson et al., 2014; Koster and Ozgen, 2021; Carlsen et al., 2016; Neves Jr et al., 2017) and individuals with high skills being more likely to migrate to large labor markets (see Bacolod et al., 2021). The summary statistics also show that descriptively, wages and match quality are highest in large labor markets and smallest in small labor markets.

	Large labor markets	Medium sized labor markets	Small labor markets
	(1)	(2)	(3)
Panel A. Labor markets			
Number of labor markets	3	20	50
Number of observations	2,660,719	2,407,657	937,253
	(44%)	(40%)	(15%)
Panel B. Population characteristics			
Age	41.85	41.90	42.49
Cognitive skill	0.29	0.09	-0.036
Non-cognitive skills	0.14	0.07	-0.01
Share University education	49.40	37.70	28.86
LN Wage	10.35	10.22	10.18
Match quality	0.097	-0.06	-0.11

Table 1. Summary statistics

Notes: This table provides summary statistics separate by the size of the local labor market. I rely on Statistics Sweden's definition of local labor markets based on commuting patterns. The sample includes males born between 1951 and 1976, with outcomes for the years 1996-2013. The skill and match quality measure is standardized to have mean 0 and standard deviation 1.

4 Empirical approach

This section describes how the match quality measure is constructed and presents the method for studying the difference in match quality in labor markets of different sizes.

4.1 Match quality measure

To construct the match quality measures, I use data on wages and eight different types of skills, both cognitive (inductive, verbal, spatial, and technical ability) as well as non-cognitive skills (social maturity, intensity, psychological energy, and emotional stability). The skills measures have been shown to be an important predictor of wages (Lindqvist and Vestman, 2011). The idea behind the match quality measure is that different combinations of skills might be useful in different occupations, and individuals have high match quality if they work in an occupation where their skills are highly valued. I follow Fredriksson et al. (2018) and use tenured workers to proxy for skill requirements. The conclusions remain similar when using O^*Net data to proxy for skill requirements in the occupation.⁵

The match quality measure is then constructed by estimating returns to skills in each occupation. The functional form of the match quality measure is theoretically unclear. I thus deviate from the linear match quality measure used in Fredriksson et al. (2018) and instead use recent advancement in the machine learning literature and estimate the match quality measure using a random forest (in similarity with Almgren et al., 2022). ⁶

The random forest is trained on data on each of the eight skills and residualized wages, where log wages are residualized on age and year-fixed effects. The training sample is constructed from a random sample of 20 percent of all tenured individuals. Individuals from the training sample are then excluded from the test data. All results shown in the paper only include individuals in the test data. The random forest is trained with residualized wages as the outcome and the eight skills as the independent variables. Thus, the random forest predicts wages based on skills. Separate random forest algorithms are estimated for each occupation, allowing different interactions of skills to be differently important for returns in each occupation. Appendix A.1 shows the relationship between predicted wages based on skills from the random forest for the test data and actual residualized wages. As can be seen in Figure A1 there is a clear positive relationship between the predicted wages from the random forest and actual wages.

Highly skilled individuals will have higher returns in all occupations. To take away the direct impact of skill, I construct the match quality measure as the return to skill for the occupation the individual works in minus the market return to skill. The idea is that the match quality is higher if an individual works in an occupation with higher returns to skills than they could receive on average on the labor market. To estimate the market return, I train the random forest on all workers in the training sample. Match quality is then defined as predicted earnings based on skills in the occupation the individual works in minus market returns:

⁵Section 8.3 provides results where the match quality measure instead are constructed using O*NET data to proxy for skill requirement in each occupation, following Guvenen et al. (2020). ⁶Fredriksson et al. (2018) use two different match quality measures, one using the skills of tenured workers and one using the estimated return to skills. The match quality measure in this paper is more similar to the estimated return to skills match quality measure, where match quality is defined as $MQ_{ij} = \sum_{s=1}^{8} (\hat{\beta}_{js} - \hat{\beta}_s) X_{is}$, where $\hat{\beta}_{js}$ is the estimated return to skill s in job j, $\hat{\beta}_s$ is the estimated return to skill s on the market, and X_{is} is the skill level of skill *s* for individual *i*. In difference to Fredriksson et al. (2018), this study focuses on match quality on the occupation level instead of job level and uses a random forest instead of assuming any functional form of the match quality measure.

$MQ_{io} = Return_{io} - Return_{iM}$

where $Return_{io}$ is the predicted return estimated with the random forest based on the skills for individual *i* working in occupation *o*, and $Return_{iM}$ is the estimated market return for individual *i*. To ease interpretation, the match quality measure is then normalized to have a mean zero and standard deviation equal to one. Figure A2 in appendix A.1 shows match quality plotted against skills. The figure shows a negative relationship between match quality and skills, indicating that higher match quality in larger cities can not be driven by higher skill levels in larger cities. Instead it seems like individuals with higher skill levels have a harder time finding a good match compared to individuals with lower skill levels.

Since match quality is predicted using only skills, this implies that two individuals with the same skills who work in the same occupation will have the same match quality, even if they live in different labor markets. Moreover, the only way for an individual to change match quality is by changing occupation.

4.2 Methodology

To estimate the difference in match quality depending on labor market size, I estimate equation (1) below, using OLS.

$$MQ_{ito} = \beta Log(Citysize)_{itL} + X_{it} + \varepsilon_{it} (1)$$

The dependent variable of interest is MQ_{it} , the match quality for individual i in time period t working in occupation o. The independent variable of interest is $Log(Citysize)_{itL}$, the log of the number of inhabitants in the local labor market. Appendix A.2 presents results with labor market size as categorical variables, by including dummy variables for living in medium or large labor markets compared to the omitted category living in a small labor market.

In the main specification, X_{it} includes age-fixed effects, year-fixed effects, and second-order polynomials in each of the eight skills to compare individuals with similar skill levels. Furthermore, alternative specifications are presented where X_{it} controls for additional individual characteristics such as education, and, in some specifications also individual fixed effects. It should be noted that, while controlling for education allows comparing more similar individuals, more education possibilities in larger labor markets could also be a mechanism for why individuals in larger labor markets can obtain higher match quality. Individual fixed effects control for all factors that are constant within the individual over time. However, when using individual fixed effects, the effect of local labor market size on match quality is identified from individuals who move between local labor markets of different sizes. If individuals dislike to move, we can expect that individuals who still choose to move to this for a reason, for example, to improve their occupation match. Thus, we can expect the estimate for the sub-population of movers to differ from the effect identified for all individuals.

5 Wage differences and labor market size

Research has shown that individuals living in larger labor markets earn more than those living in smaller labor markets. This pattern has been shown to hold in the United States (Papageorgiou, 2022), and in different European countries (see for example Rosenthal and Strange, 2008) including Sweden (see Eliasson and Westerlund, 2022). Table 2 provides evidence on wage premiums in larger labor markets, also in the setting studied here, by estimating regression 1 with log wage as the dependent variable. Conditional on only year and age fixed effects, the city-size wage elasticity is 0.053, indicating that a 1 percent larger city is associated with 0.05 percent higher wages. Adding controls for skills, the wage elasticity is still 0.039, which means that conditional on skills wages are 12 percent higher in large compared to small labor markets (see table A2 in appendix). Adding controls for education in addition to skills only reduces the estimate slightly, and the wage elasticity of city size is still 0.036.

The magnitude of the effect changes somewhat when individual fixed effects are included, in line with the effect now being identified from movers which might differ from the population. However, even when including individual fixed effects, wages are still higher in larger compared to smaller labor markets.

	(1)	(2)	(3)	(4)
VARIABLES	Ln Wage	Ln Wage	Ln Wage	Ln Wage
Log population size	0.053***	0.039***	0.036***	0.015***
	(0.000)	(0.000)	(0.000)	(0.000)
Year Fe	Yes	Yes	Yes	Yes
Age FE	Yes	Yes	Yes	Yes
Control skills	No	Yes	Yes	No
Education control	No	No	Yes	No
Individual FE	No	No	No	Yes
Observations	6,005,629	6,005,629	6,002,946	6,005,629

Notes: This table provides results estimating equation 1, with log wage as the dependent variable, estimating the difference in log wages for individuals who live in labor markets of different sizes. The sample includes males born between 1951 and 1976, with outcomes for the years 1996-2013. In column 4, where individual fixed effects are included, the age dummies are normalized to be constant between 45 and 54 to avoid multicollinearity between age and year-fixed effects. Standard errors are adjusted for 659,206 clusters at the individual level.*** p < 0.01, ** p < 0.05, * p < 0.1

Figure 1 below shows a map of the distribution of labor market size in Sweden. As can be seen in the map, smaller and larger labor markets exist in all parts of Sweden.



Figure 1. Labor market size Sweden

Notes: The figure plots the log size of local labor markets in Sweden. I rely on Statistics Sweden's definition of local labor markets based on commuting patterns. Darker colors represent larger labor markets.

6 Results

This section presents the key results on the city-size match quality gap. Section 6.1 presents the main results on how match quality differs depending on labor market size estimated with regression 1. It is possible that not all individuals gain equally in terms of match quality from living in a large labor market. Section 6.2, therefore presents how match quality differs with labor market size separately depending on an individual's skill level. Section 6.3 presents life-cycle dynamics on how match quality develops over the life-cycle for individuals who live in small, medium, and large labor markets.

6.1 Main results

Table 3 presents results for how match quality varies with local labor market size. In column 1, we see that individuals living in larger labor markets have

higher match quality than individuals in smaller labor markets. Column 2 controls for second-order polynomials in each of the eight skills. Controlling for skills has a small and positive impact on the estimate, indicating that the difference in match quality depending on labor market size is not driven by different skill compositions of workers depending on local labor market size. When controlling for skills, the estimates suggest that an increase in city size by 10 percent is associated with 0.007 standard deviations higher match quality. Thus, the results in the table are in line with the theory of higher match quality in larger labor markets.

Column 3 in addition controls for education. The estimate falls somewhat, indicating that one mechanism for higher match quality in larger labor markets might be more education possibilities, which might increase the potential to realize the optimal match. However, even conditional on education match quality is higher in larger local labor markets. Column 4 includes individual fixed effects, and thus the estimated effect of local labor market size on match quality comes from individuals who move between local labor markets of different sizes. When controlling for individual fixed effects the effect decreases substantially. As discussed before, this could be because the individual fixed effect captures individual characteristics that affect match quality. On the other hand, it could also be because the effect is now identified by movers. If individuals dislike to move, they might, for example, only choose to move if they receive higher match quality. Nevertheless, when including individual fixed effects, match quality is higher in larger compared to smaller local labor markets.

Table A2 in Appendix B shows results with labor market size as a categorical variable, where match quality is divided into large, medium, and small labor markets. The conclusion from this specification is similar: larger and medium-sized labor markets have higher match quality than small labor markets. From Figure A4 it is clear that the relationship between match quality and city size is relatively linear, where match quality gradually increases with log city size.

The result in Table 3 speaks to the role of match quality as one explanation for why larger labor markets have higher wages. However, the question still remains if match quality is an important contribution to the city-size wage premium or not. In Table 4 I quantify how much of the wage premium in large local labor markets can be attributed to match quality. This is done by estimating the city-size wage premium regression and comparing the coefficient for local labor market size with and without controls for match quality. Thus, the coefficient in column 2 shows how much of the city-size wage premium remains when match quality is held constant, and the difference between the coefficients in columns 1 and 2 shows how much of the wage gap between large and small labor markets can be attributed to match quality. Note that controls for skills are included in both columns 1 and 2. Thus, the difference between the coefficient in columns 1 and 2 shows how much match quality

	(1)	(2)	(3)	(4)
VARIABLES	Match	Match	Match	Match
	quality	quality	quality	quality
Log population size	0.064***	0.070***	0.059***	0.009***
	(0.001)	(0.001)	(0.001)	(0.001)
Year Fe	Yes	Yes	Yes	Yes
Age FE	Yes	Yes	Yes	Yes
Control skills	No	Yes	Yes	Yes
Control education	No	No	Yes	No
Individual FE	No	No	No	Yes
Observations	6,005,629	6,005,629	6,002,946	6,005,629

Table 3. Main result

Notes: This table provides result estimating equation 1, estimating the difference in match quality for individuals who live in labor markets of different sizes. The sample includes males born between 1951 and 1976, with outcomes for the years 1996-2013. In column 4, where individual fixed effects are included, the age dummies are normalized to be constant between 45 and 54 to avoid multicollinearity between age and year-fixed effects. Standard errors are adjusted for 659,206 clusters at the individual level. *** p<0.01, ** p<0.05, * p<0.1

contributes to the conditional wage premium, where the fact that the population in small and large labor markets have different skill composition has already been taken into account.

Comparing the coefficients in columns 1 and 2 suggests that match quality explains around 30 percent of the city-size wage gap. Occupational match quality thus explains an important part of the wage difference between smaller and larger labor markets. The magnitude is in line with the result of Papageorgiou (2022), who uses a calibrated model and shows that occupational match quality explains around 35 percent of the city-size wage premium in the United States.

When using match quality as a control instead of as the dependent variable, it is important to get as accurate measure of match quality as possible. If the match quality measure is estimated with error, controlling for match quality might not capture the true match quality effect, and the contribution of match quality to the city-size wage gap might be underestimated. When using the linear version of match quality as a control instead of the non-parametric match quality measure estimated with the random forest, match quality is estimated to explain around 10 percent, instead of 30 percent of the city-size wage gap (see table A1 in appendix). Thus, even if the conclusion of higher match quality in larger labor markets holds with the linear match quality measure (see table 10), using the recent advancements in machine learning to estimate a non-parametric version of the match quality measure seems to allow for a more accurate quantification of the importance of match quality to explain the city-size wage gap.

	(1)	(2)	(3)	(4)
VARIABLES	Ln Wage	Ln Wage	Ln Wage	Ln Wage
Log population size	0.039***	0.028***	0.053***	0.043***
	(0.000)	(0.000)	(0.000)	(0.000)
Match quality		0.169***		0.163***
		(0.000)		(0.000)
Year Fe	Yes	Yes	Yes	Yes
Age FE	Yes	Yes	Yes	Yes
Control skills	Yes	Yes	No	No
Observations	6,005,629	6,005,629	6,005,629	6,005,629

Table 4. Quantify the contribution of match quality to city-size wage premium

Notes: This table provides results estimating equation 1, with log wage as the dependent variable adding match quality as a control comparing the coefficient for local labor market size with and without controls for match quality. The sample includes males born between 1951 and 1976, with outcomes for the years 1996-2013. Standard errors are adjusted for 659,206 clusters at the individual level. *** p<0.01, ** p<0.05, * p<0.1

6.2 Heterogeneity analysis

Earlier research has shown that the city-size wage premium is higher among high-skilled individuals (see for example Bacolod et al., 2009). For policy makers who may want to decrease the city-size match quality gap, understanding if the city-size match quality gap exists for everybody or also is concentrated to highly skilled individuals might be important to implement policy effectively. Table 5 therefore explores whether the city-size match quality gap is also higher among high-skilled individuals. Heterogeneity analysis is presented by the level of cognitive skills, non-cognitive skills and education.

There is a positive association between match quality and living in a larger labor market across the skill distribution. However, the correlation between labor market size and match quality is especially large for highly skilled individuals, a pattern that holds when measuring skills in all dimensions: cognitive skills, non-cognitive skills, and education. Table 6 quantifies how much the city-size wage premium can be explained by match quality separately for all skill groups. Match quality is one important explanation behind the citysize wage premiums for all groups and explains between 24% to 32% of the city-size wage premium. Match quality explains a slightly higher part of the wage premium for high-skilled workers, indicating that one potential reason for higher wage premiums for high-skilled individuals might be that high-skill individuals benefit more in terms of match quality from living in a large compared to a small local labor market. Nevertheless, it should be noted that both individuals with higher and lower skill levels seem to benefit in terms of higher match quality from living in larger labor markets.
Dependent variable: Match quality						
	(1) Cognitive ability under median	(2) Cognitive ability over median	(3) Non- cognitive ability under	(4) Non- cognitive ability over	(5) High school or less	(6) University
			median	median		
Log population size	0.041*** (0.001)	0.099*** (0.001)	0.046*** (0.001)	0.094*** (0.001)	0.042*** (0.001)	0.085*** (0.002)
Year Fe	Yes	Yes	Yes	Yes	Yes	Yes
Age FE	Yes	Yes	Yes	Yes	Yes	Yes
Control skills	Yes	Yes	Yes	Yes	Yes	Yes
Observations	3,004,065	3,001,564	3,068,475	2,936,693	3,409,100	2,594,733

Table 5. Heterogeneity analysis

Notes: This table provides result estimating equation 1, estimating the difference in match quality for individuals who live in labor markets of different sizes, separately for different skill groups. The sample includes males born between 1951 and 1976, with outcomes for the years 1996-2013. Standard errors are clustered at the individual level. *** p<0.01, ** p<0.05, * p<0.1

	(1)	(2)	(3)	(4)	(5)	(6)
	Cognitive	Cognitive	Non-	Non-	High	University
	ability	ability	cognitive	cognitive	school	
	under	over	ability	ability	or less	
	me-	me-	under	over		
	dian	dian	me-	me-		
			dian	dian		
Panel A.						
Log population size	0.026***	0.053***	0.028***	0.050***	0.025***	0.053***
	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
Panel B.						
Log population size	0.019***	0.036***	0.021***	0.035***	0.019***	0.038***
	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
Match quality	0.168***	0.168***	0.169***	0.167***	0.148***	0.171***
	(0.001)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
Year Fe	Yes	Yes	Yes	Yes	Yes	Yes
Age FE	Yes	Yes	Yes	Yes	Yes	Yes
Control skills	Yes	Yes	Yes	Yes	Yes	Yes
Observations	3,004,065	3,001,564	3,068,475	2,936,693	3,409,100	2,594,733
Panel C. Shar	e of city siz	e wage pre	mium expl	lained by n	natch quali	ity
Share explained	27%	32%	25%	30%	24%	28%

Table 6. Contribution to city-size wage premium by skill group

Dependent variable: Ln Wage

Notes: This table provides results estimating equation 1, separately by skill group, with log wage as the dependent variable adding match quality as a control comparing the coefficient for local labor market size with and without controls for match quality. The sample includes males born between 1951 and 1976, with outcomes for the years 1996-2013. Panel A shows results for the city-size wage premium without controls for match quality, Panel B shows results for the city-size wage premium with controls for match quality and Panel C shows how much match quality contributes to the city-size wage premium. Standard errors are clustered at the individual level.*** p < 0.01, ** p < 0.05, * p < 0.1

6.3 Life-cycle pattern

The previous section showed that individuals living in large labor markets have higher match quality than individuals in smaller labor markets, and that occupational match quality explains a substantial part of the wage difference between labor markets of different sizes. This section explores the life-cycle dynamics of match quality depending on labor market size, by showing new evidence on how match quality evolves over the life cycles depending on labor market size.

Figure 2 plots match quality against age separately depending on labor market size, where the categorical definition of labor market sizes is used and la-





(a) Match quality

(b) Match quality stayers

Notes: The figures plot match quality against age separately for individuals living in small, medium, and large labor markets. The sample includes males born between 1951 and 1976, with outcomes for the years 1996-2013. Figure (a) includes all individuals and figure (b) only includes individuals who always stay in the local labor market they were born in.

bor markets are divided into small, medium and large labor markets. Figure 2a plots this for all individuals after where they live at that age and figure 2b plots this for individuals who always stay in the local labor market they were born in. At age 25, match quality is similar in local labor markets of different sizes. Match quality then increases over age in all labor markets. However, the increase in match quality over the life cycle is, on average, larger in large compared to small labor markets. The increase in match quality is in line with the idea that individuals have imperfect information about their optimal match but learn about their match quality over time as they work (see Guvenen et al., 2020; Fredriksson et al., 2018).

However, the increase could also be in line with an occupation diversity explanation. If entry-level jobs are available in most cities but some highskilled jobs are only available in larger cities, this could also explain the pattern observed in the data. At the beginning of the lice-cycle match quality is similar between different labor markets, and when individuals in larger labor markets continue to better matches in occupations that only exist in the larger labor markets this would result in a larger increase in match quality in larger labor markets.

Since Figure 2a shows match quality after where the individual lives at that age, the pattern could also be affected by a changed composition of individuals in large and small labor markets over the life cycle, where individuals move to local labor markets of different sizes over the life cycle. However, the pattern looks similar in figure 2b, which only includes individuals who always remain in the local labor market they were born in, indicating that the life-cycle dynamics in match quality are not driven by the changed composition of individuals in different labor markets.

Since the mean education level is higher in larger labor markets (see Table 1), individuals in larger labor markets might enter the labor market later, which could give different life-cycle patterns for individuals in large and small labor markets. To take this into account, Figure A5 in Appendix A.3 shows the results using potential experience instead of age.⁷ The results look similar when using potential experience, indicating that the increase in match quality over the life cycle in large labor markets compared to small labor markets is not driven by the fact that individuals in large and small labor markets enter the labor markets at different ages.

To examine the dynamics behind the life-cycle pattern, I provide evidence for how the likelihood of changing occupation as well as moving up in the occupation match-quality distribution is related to labor market size. This is done by running equation (1) with occupation change and higher occupation match quality as the dependent variable. Changing occupation is defined as working in another occupation than the individual was observed in last. Results are presented for all occupation changes, and separately for changes within and between firms. An occupation change within the firm is defined as a change of occupation where the firm remains the same and an occupation change between firms is defined as a change of both occupation and firm. Higher occupation match quality is constructed as a dummy variable equal to one if the individual changes to an occupation with higher match quality than before and zero if the individual changes to an occupation with lower match quality than before. When estimating the result for higher occupation match quality, the sample is limited to individuals who change occupation.

Table 7 shows that individuals in larger labor markets are more likely to change occupations. The result of more occupation switches is in line with the theory of lower costs of changing occupation in larger labor markets (see Puga, 2010; Bleakley and Lin, 2012). The increased likelihood of changing occupation in large labor markets is equally driven by within and between firm changes. Thus, the lower cost of changing occupations in larger labor markets could both come from more job openings, making it easier to change firms and more career opportunities within the firm and thus a lower cost of changing occupations without having to change employers. For individuals in medium-sized labor markets, the higher likelihood of changing occupation compared to small labor markets is entirely driven by within-firm changes.

Furthermore, Table 7 shows that individuals in larger labor markets are more likely to move up in the occupation match quality distribution when they change occupation. For individuals in large compared to small labor markets, the move up the occupation match-quality distribution is mainly driven by between-firm occupation changes. However, for individuals in medium-sized local labor markets, the pattern looks different. The increase in match quality

⁷Potential experience is defined as age minus (years of education+ six), where six is the age the individual usually starts school.

	(1)	(2)	(3)	(4)	(5)	(6)	
VARIABLES	Occupation switch			Switch to	Switch to higher match quality		
	All	Within firm	Between firm	All	Within firm	Between firm	
Lance Johan montrate	0.019***	0.000***	0.000***	0.017***	0.002	0.014***	
Large labor markets	(0.001)	(0.009^{+++})	(0.009^{+++})	$(0.01)^{++++}$	(0.002)	$(0.0014^{+0.04})$	
Medium labor markets	0.002***	0.003***	-0.000	0.008***	0.009***	-0.001	
	(0.001)	(0.000)	(0.000)	(0.001)	(0.001)	(0.001)	
Year Fe	Yes	Yes	Yes	Yes	Yes	Yes	
Age FE	Yes	Yes	Yes	Yes	Yes	Yes	
Control skills	Yes	Yes	Yes	Yes	Yes	Yes	
Observations	5,346,423	5,346,423	5,346,423	790,525	790,525	790,525	

Table 7. Match quality dynamics

Notes: This table provides results estimating regressions with changing occupation and good switch as the dependent variables, and dummy variables for living in a large and medium-sized labor market compared to the omitted category living in a small labor market as the independent variable of interest. The sample includes males born between 1951 and 1976, with outcomes for the years 1996-2013. Changing occupation is defined as working in another occupation than the individual did in the last observation. Switch to higher match quality is defined as a dummy variable equal to one for individuals who change occupation to an occupation with higher match quality and zero for individuals who change occupation to an occupation with worse match quality. When estimating the regression for the outcome switching to a higher match quality occupation, the sample is limited to individuals who switch occupations. Standard errors are clustered at the individual level.*** p<0.01, ** p<0.05, * p<0.1

for medium-sized compared to small local labor markets is entirely driven by within-firm changes.

Thus, the analysis shows that the improvement of match quality over the life cycle in larger compared to smaller labor markets comes from both more occupation switches and better occupation switches conditional on switching occupations.

7 Mechanisms

Section 6 showed that match quality is higher in larger labor markets. This section examines two different theoretical explanations for higher match quality in larger labor markets: more occupation diversity and more learning possibilities about match quality in larger labor markets.

7.1 Occupation diversity

One theoretical mechanism behind the difference in match quality between

large and small local labor markets is occupation diversity. If there are mobileity costs, and the occupation where the individual would have the best match quality is not available in the labor market, the individual might not be able to reach the best occupation match. Thus, a more diversified occupation structure gives individuals more occupations to choose from and increases the likelihood that individuals find an occupation that matches their skill set.

To study if occupation diversity is a mechanism behind higher match quality in larger labor markets, I construct a Herfindahl-Hirschman Index (HHI). The HHI index is a measure widely used to measure employment concentration (see for example Thoresson, 2024; Benmelech et al., 2022). Here I construct a HHI measure of occupation concentration in the labor market, defined as: $HHI_L = \sum_{o=1}^{o} (s_o)^2$, where L denotes the labor market, s_o is the employment share in the labor market of occupation o. Thus, the HHI index is calculated by squaring the employment share of each occupation in the labor market and summing the number over all occupations. The HHI index can take values between 0 and 1, where 1 means there is only one occupation in the labor market and zero means less employment concentration and, thus a more diversified occupation structure.

Furthermore, to study if the occupations that don't exist are relevant, I study if the occupation where the individual has the highest match quality exists in the labor market the individual lives in. Where the occupation is defined as existing if anybody in the occupation sample, in the relevant labor market, work in the occupation.



Notes: The HHI index is a measure of occupation concentration in the labor market, defined as: $HHI_L = \sum_{o=1}^{o} (s_o)^2$, where *L* denote the labor market, s_o is the employment share in the labor market of occupation o. The HHI index can take values between 0 and 1, where 1 means it is only one occupation in the labor market and zero means less employment concentration and thus a more diversified occupation structure. Figure (a) plots the HHI index against labor market size and figure (b) plots the HHI index against match quality.

Figure 3a plots occupation diversity, defined with the HHI index, against labor market size. The figure shows a negative relationship between labor market size and the HHI index. Thus, larger labor markets seem to have less occupation concentration and more occupation diversity. The conclusions of higher occupation diversity in larger labor markets remain similar if occupation diversity is instead defined as the share of all occupations that exist in the local labor market following Papageorgiou (2022) (see Table A6b and A7b in Appendix A.5). The finding of more occupation diversity in larger labor markets is in line with the result of more occupation diversity in larger labor markets in the United States (see Papageorgiou, 2022) and more industry diversity in larger labor markets in Sweden (see Korpi, 2007).

Figure 3b shows that there is a positive association between occupation diversity and match quality. Thus, the descriptive pattern supports the theory of occupation diversity as one mechanism for why match quality is higher in larger local labor markets.

To study how much occupation diversity contributes to the higher match quality in larger labor markets, I run equation (1) and include a control for the HHI index. Column 1 in Table 8 shows the main result for how match quality is related to labor market size, and column 2 shows the same result but controls for the HHI index. From the table, it is clear that the HHI index is negatively correlated to match quality (and thus, occupation diversity is positively correlated with match quality). The coefficient on labor market size becomes smaller when including controls for occupation diversity, indicating that occupation diversity might explain part of the city-size match quality gap. The coefficient on labor market size is reduced by approximately 23 percent when controls for occupation diversity are included. Column 3 shows that the occupation in which the individual would have the highest match quality is more likely to exist in larger compared to smaller labor markets. Thus, the results in this section suggest that one reason for higher match quality in larger labor markets.

	(1)	(2)	(3)
VARIABLES	Match quality	Match quality	Best match exist
			in labor market
log population size	0.070***	0.054***	0.003***
	(0.001)	(0.002)	(0.000)
Herfindahl-Hirschman Index		-2.370***	
		(0.209)	
Year Fe	Yes	Yes	Yes
Age FE	Yes	Yes	Yes
Control skills	Yes	Yes	Yes
Observations	6,005,629	6,005,629	6,005,629

Table 8. Occupation diversity

Notes: This table provides result estimating equation 1, estimating the difference in match quality for individuals who live in labor market of different sizes, column 2 includes controls for the HHI. The HHI is a measure of occupation concentration in the labor market, defined as: $HHI_L = \sum_{o=1}^{o} (s_o)^2$, where *L* denote the labor market, s_o is the employment share in the labor market of occupation o. The HHI can take values between 0 and 1, where 1 means it is only one occupation in the labor market and zero means less employment concentration and thus a more diversified occupation structure. Standard errors are clustered at the individual level. *** p<0.01, ** p<0.05, * p<0.1.

7.2 Match quality and learning

Learning and match quality have been discussed as different mechanisms behind the city-size wage premium (Puga, 2010), and research has shown that more learning possibilities in larger cities are one important explanation for the city-size wage premium (De La Roca and Puga, 2016; D'Costa and Overman, 2014). However, learning and match quality are not necessarily disconnected explanations. Since individuals seem to be unsure about their match quality from the beginning, one possible explanation for higher match quality in larger labor markets is more learning opportunities about which occupations are a good match in larger labor markets. As noted by Wheeler (2008) a lower cost of switching jobs might allow the worker to explore more different jobs, making it easier to find the job where they are the most productive.

To estimate if learning is one channel for higher match quality in larger labor markets, I follow De La Roca and Puga (2016) and estimate the return to experience in labor markets of different sizes but with match quality instead of wages as the dependent variable:

$$MQ_{it} = \alpha_L + \lambda_i + \sum_{k=1}^{j=3} \mu_{kL} \theta_{ikt} + \rho_{iLt}$$
(2)

In equation (2) α_L are separate dummy variables for living in a large or medium labor market, with the omitted category living in a small labor market, λ_i is individual fixed effects, θ_{ikt} is years of experience working in a labor

market of size *k* for worker *i* and time period *t*. Years of experience are calculated as years of working in a labor market of size *L* since age 20.⁸ The value of experience is allowed to vary depending on both in which labor market the experience was obtained (*k*) and where the individual lives now (*L*). The specification also includes second-order polynomial terms of experience. Since experience in labor markets of different sizes now are included, α_L captures the static match quality premium of moving to labor market size *L*.

The idea behind the exercise is that if learning is one important mechanism behind higher match quality in larger labor markets, not only the static premium should matter. Instead, match quality should increase gradually and years of experience in large labor markets should be important. Moreover, if learning is the primary mechanism driving the difference in match quality, the knowledge of what is a good match should also be portable. Thus, if learning is important, we could expect years of experience in a large city to also be beneficial for individuals who now live in a small city.

The learning mechanism about match quality could either be driven by labor market explanations or more education possibilities in larger cities. Thus, it is also possible that the learning possibilities in larger cities are connected to learning through a more direct education channel. By moving to a large city, the worker can get access to a university education, which could increase the knowledge of the individual and give access to more occupations. To examine if the learning channel about match quality seems to be driven by education possibilities or labor market factors, I estimate regression (2) separately for individuals with and without a university education. The idea is that for individuals without a university education, the remaining mechanisms are related to labor market factors. As stated above, labor market factors driving the effect could be initial uncertainty about match quality and a lower cost of switching occupations in larger labor markets, allowing workers to explore more occupations and learn more about what occupation is a good match. However, another possible labor market driver of the effect could be more learning possibilities on the job in larger cities (see Glaeser and Maré, 2001; De La Roca and Puga, 2016). According to this channel, experience in large cities gives valuable knowledge that might open up possibilities to enter other occupations.

Table 9, column 1 shows the results for all workers, column 2 shows the results for workers with high school or less, and column 3 shows the results for workers with university education. The regression includes experience in

⁸To be able to calculate the experience of the worker, I use data for where the worker lives and if they have earnings from 1990. To be able to observe the worker from the age 20 the sample is limited to workers born 1970-1976. Since occupation data is available from 1996-2013, years from 1990 are used to calculate experience, but only years after 1996 are used to calculate the outcomes. One year of experience in labor marker L is defined as a year of living in labor market L and having positive earnings. Table A4 in Appendix A.4 presents results when experience is instead defined as years of living in a labor market since age 20. The results remain similar whether years of experience are defined conditional on working or not.

large and medium-sized labor markets as well as overall experience. Thus, the term experience now captures having experience in a small labor market and experience in large and medium captures the extra benefit of having experience in a large and medium relative to a small labor market. When studying the result for all workers, it is clear that the static premium becomes insignificant. Experience in large and medium relative to a small labor market is positive and significant, indicating that learning is one important mechanism behind the higher match quality in large labor markets. The interaction term of having experience in a large labor market and living in a large labor market is negative, indicating that the experience from a large labor market. Thus, the knowledge the individuals now living in a small labor market. Thus, the knowledge the individual has obtained about what occupations are a good match seems to be portable.

When studying the result separately for individuals with high school education and individuals with university education, it is clear that for both groups experience in large and medium is more valuable for match quality than experience in small labor markets. The point estimates are larger for those with a university education, which could indicate that part of the learning effect found in column 1 can be learning through university education. However, experience in large labor markets is also more valuable than experience in small labor markets for individuals with a high school education, indicating that parts of the learning effect also come through the labor market. Thus, the result suggests that one explanation for higher match quality in larger labor markets is less frictions and more learning possibilities of what occupations to work in to have high match quality in larger labor markets.

Dependent variable	: Match quality	у	
	(1)	(2)	(3)
VARIABLES	All	High	University
		school or	
		less	
T 11 1.	0.001	0.040	0.074**
Large labor market	-0.001	0.049	-0.074**
	(0.026)	(0.037)	(0.035)
Medium labor market	-0.032	-0.001	-0.108***
	(0.023)	(0.032)	(0.033)
Experience large	0.071***	0.038***	0.060***
2	(0.006)	(0.010)	(0.008)
(Experience $large)^2$	-0.002***	-0.001*	-0.002***
	(0.000)	(0.001)	(0.001)
Experience medium	0.038***	0.020***	0.029***
	(0.005)	(0.007)	(0.008)
$(Experience medium)^2$	-0.002***	-0.001**	-0.001**
	(0.000)	(0.000)	(0.001)
Experience	0.022***	0.010***	0.049***
*	(0.002)	(0.002)	(0.004)
Experience ²	-0.000**	0.000	-0.001***
1	(0.000)	(0.000)	(0.000)
Large labor market#experience large	-0.041***	-0.014	-0.048***
6	(0.006)	(0.009)	(0.008)
Large labor market#(experience $large)^2$	0.002***	0.001	0.002***
Daige moor maneer (experience targe)	(0,000)	(0.001)	(0.001)
Medium labor market#experience large	-0.000	0.013	-0.011
inconstitution mainteen en perioritee mage	(0.006)	(0.010)	(0.008)
Medium labor market#(experience $large)^2$	-0.001	-0.001	0.000
fiedrani rabor markets (experience varge)	(0,000)	(0.001)	(0.000)
Large Jahor market#experience medium	-0.000	-0.001	-0.004
Large labor market/experience medium	(0.006)	(0.001)	(0.008)
Larga labor markat#(avpariance medium) ²	(0.000)	(0.000)	(0.008)
Large labor market#(experience meatum)	(0,000)	(0.000)	(0.001)
Madium labor montratteur prior as madium	(0.000)	(0.001)	(0.001)
Medium fabor market#experience medium	-0.029***	-0.017**	-0.038***
Madiene 1-1	(0.003)	(0.007)	(0.007)
Medium labor market#(experience meaium) ⁻	0.001****	0.001*	0.002^{****}
I	(0.000)	(0.000)	(0.001)
Large labor market#experience	0.01/***	0.004	0.033***
	(0.004)	(0.005)	(0.006)
Large labor market# <i>experience</i> ²	-0.001***	-0.000**	-0.001***
	(0.000)	(0.000)	(0.000)
Medium labor market#experience	0.006*	0.003	0.026***
2	(0.003)	(0.005)	(0.005)
Medium labor market# <i>expeirence</i> ²	0.000	-0.000	-0.001***
	(0.000)	(0.000)	(0.000)
Individual FE	Yes	Yes	Yes
Observations	1,351,753	679,803	671,950

Table 9. Learning

Notes: This table provides result estimating equation 2. The sample includes males born between 1970 and 1976, with outcomes for the years 1996-2013. Standard errors are clustered at the individual level. *** p<0.01, ** p<0.05, * p<0.1

8 Robustness

Earlier sections have shown that match quality is higher in larger labor markets. This section examines the robustness of the results. First, results are presented for individuals who move between labor markets of different sizes, making it possible to control for individual fixed effects. It should be noted that even if the movers' design allows for control for individual characteristics that are constant over time, the decision to move is a choice, and individuals who move have overcome the mobility frictions and might move optimally. Thus, the result from this group might differ from the general population. Nevertheless, studying movers is a good complement to the main specification. Finally, the section also presents alternative specifications of the match quality measure.

8.1 Comparing movers and stayers

This section presents results comparing individuals who move from the small local labor market they were born in, to a large labor market compared to individuals who always stay in the small labor, by estimating:

$$y_{it} = \alpha_i + \lambda_t + \sum_l \mu_l D_{it}^l + X_{it} + \rho_{it} \quad (3)$$

where α_i is individual fixed effects, λ_i is year-fixed effects, and X_{it} is age dummies, and D_{it}^l is time since the move. Including both year and age-fixed effects together with the individual fixed effect would give perfect multicollinearity, so to overcome this problem the age dummies are normalized to be constant between 45 and 54. To avoid the problem with two-way fixed effects models and heterogeneous treatment effects, I use the procedure from Sun and Abraham (2021), and use never treated as the control, and estimate a weighted average of the cohort-time-specific treatment effects, where weights are set to the estimated cohort shares. The result is estimated for the first move the individual makes. First moves are defined as the first move in the data, observed between the years 1996 and 2013 and that the move is from the local labor market the individual is born in. The control group is individuals who always stay in the small local labor market they were born in.

Figure 4 shows the results for moving to a large labor market where year zero is the year of the move. Figure 4a shows the results for wages, and Figure 4b shows the results for match quality. From Figure 4 it is clear that both wages and match quality on average increase after an individual moves to a large local labor market. However, the figure also shows a pre-trend before moving, indicating that individuals who are on a positive trend in match quality are more likely to move to a large local labor market. Thus, it seems like match quality increases for individuals who choose to move from a small to a large labor market. However, if individuals dislike moving, they may only choose to move if match quality increases. Thus, if individuals only move





Notes: The figures plot event graphs for individuals who move from a small labor market to a large labor market estimated with Sun and Abraham (2020) estimator, using never mover who always stays in the small local labor market they where born in as the control. The x-axis plot normalized time since the move, where period zero is the time of the move. Figure (a) plots the result for log wages and figure (b) plots the result for match quality. The sample includes males born between 1951 and 1976, with outcomes for the years 1996-2013.

to increase match quality the higher match quality might come from moving itself, and not from the larger labor market. To take this into account, Section 8.2 instead compares movers to labor markets of different sizes.

8.2 Comparing movers

Moving could in itself have a direct effect on match quality, and individuals may only choose to move if they currently have a bad match. Odio Zúñiga and Yuen (2020) have for example shown that individuals who have bad match quality are more likely to move. To take away the direct effect of the move, this section therefore estimates results using only movers and comparing individuals who move to local labor markets of different sizes. This is done by estimating the regression:

$$y_{it} = \sum_{l=i}^{l=k} \beta_l N_l + \beta_T treated_i * \sum_{l=i}^{l=k} N_l + treated_i + \lambda_t + X_{it} + e_{it}$$
(3)

where N is normalized time since the move, and treated are separate dummy variables for moving to a large or medium labor market with the omitted category being moving to a small local labor market. X_{it} includes age controls, and second-order polynomials in each of the eight skills, controls for the local labor market the individual is born in, and a fixed effect for the occupation the individual works in the year before the move. The parameter of interest β_T gives the difference between moving to a large and medium labor market compared to moving to a small local labor market.



Notes: The figure plots the difference between moving to a large or medium-sized labor market compared to the omitted category moving to a small labor market. The x-axis plot normalized time since the move, where period zero is the time of the move. Figure (a) plots the result for log wages and figure (b) plots the result for match quality. The sample includes males born between 1951 and 1976, with outcomes for the years 1996-2013.

Figure 5 shows what happens to wages and match quality for individuals who move to a medium-sized and large labor market compared to the omitted category moving to a small labor market. Figure 5a shows that after the move, wages increase over time for individuals who move to a large compared to a small labor market. The estimates also suggest that over time, wages increase somewhat for individuals who move to a medium compared to a small labor market, even though the estimates are not statistically different from zero. Figure 5b shows a similar pattern for match quality, where match quality increases for individuals who move to large compared to small labor markets. The point estimates suggest that match quality increases somewhat in the first two years after the move and then is relatively constant over time. Overall, the results from the movers support the earlier result of higher match quality in large compared to small labor markets.

8.3 Robustness of the match quality measure

This section provides robustness tests to the match quality measure by presenting results with alternative match quality constructions.

One potential worry with the match quality measure is that the return in small occupations might be estimated with more uncertainty. Figure A3 in Appendix A.1 plots match quality against occupation size, and the figure indicates that there might be a relationship between match quality and occupation size. Thus, to study the robustness of the results Table A3 presents results following Guvenen et al. (2020), using the external O*Net data to proxy for

skill requirements in each occupation. ⁹ Reassuringly the conclusion of higher match quality in larger labor markets remains when using O*Net data to measure skill requirements in each occupation.

Column 2 estimates the return to skills linearly instead of using a random forest, again the conclusions remain with this specification. One other potential concern with the construction of the match quality measure is that since more individuals live in large labor markets, the training data have more individuals from large labor markets. If returns to skills in occupations are different in large and small labor markets, the estimated returns might be more similar to the returns in large labor markets. To study if this drives the results, Table 10 presents results when the returns to skills in occupations are allowed to vary with labor market size by estimating separate random forest regression for small, medium, and large labor markets. Also, with this specification, the conclusions of higher match quality in larger labor markets remain.

Another potential concern with the match quality measure is that some occupations might be over-represented in large or small labor markets and that the return might be higher in large cities for other urban wage reasons, such as a lower cost of living in smaller labor markets. To take this into account column 5 presents result when only training the random forest in large labor markets, reassuringly the results remain very similar indicating that the result of higher match quality in larger labor markets are not driven by different return in labor market of different sizes.

Finally, one potential worry with the match quality measure is that individuals might be well-matched even if they are not correctly priced for their skills. To take this into account column 3 uses the random forest to predict the probability that a worker works in an occupation instead of estimating return to skills, again the conclusions remain similar. Thus, from Table 10 it is clear the conclusion of higher match quality in larger compared to smaller labor markets also holds with the alternative definitions of match quality.

⁹To construct the match quality measure I use five different occupation skill requirements from O*Net chosen to match the skills for the individuals from the military enlistment test; inductive skills, verbal skills, spatial skills, technical skills, and social skills. The mismatch measure is constructed as the sum of the absolute difference between the rank of the standardized skill from the military enlistment test and the rank of the occupation requirement for that skill for each of the five skills. Match quality is then defined as the negative of the mismatch measure.

	(1)	(2)	(3)	(4)	(5)
VARIABLES	O*Net	Linear	Random	Return	Random
		return	forest	vary with	forest
			probabil-	labor	trained on
			ity	market	large labor
				size	market
Log population size	0.005*** (0.001)	0.015*** (0.000)	0.019*** (0.001)	0.168*** (0.001)	0.075*** (0.001)
Year Fe	Yes	Yes	Yes	Yes	Yes
Age FE	Yes	Yes	Yes	Yes	Yes
Control skills	Yes	Yes	Yes	Yes	Yes
Observations	5,820,642	6,005,629	6,005,629	6,005,629	6,005,629

Table 10. Alternative match quality measures

Notes: This table provides results estimating equation 1, estimating the difference in match quality for individuals who live in labor markets of different sizes, using alternative measures of the match quality measure. The sample includes males born between 1951 and 1976, with outcomes for the years 1996-2013. Standard errors are clustered at the individual level. *** p < 0.01, ** p < 0.05, * p < 0.1

9 Conclusion

While earlier research found that wages are higher in larger than smaller labor markets, the mechanism behind this has been less studied. One theoretical mechanism for the city-size wage premium is that workers in larger cities are better matched to their occupations. However, even though this mechanism has been discussed theoretically, the empirical evidence for how match quality differs in small and large labor markets has been limited. Building on the recent literature on multidimensional skill mismatch, this paper contributes to the literature by providing direct evidence on how occupational skill match quality differs depending on the labor market size in Sweden.

The key result from the paper is that match quality indeed is higher in larger compared to smaller labor markets. This finding holds conditional on skills, when including individual fixed effects for individuals who move from a small to a large labor market, and when using movers comparing individuals who move to a large compared to a small labor market. In the preferred specification, match quality explains around 30 percent of the city-size wage gap. The difference in match quality between individuals in large and small labor markets is especially large for individuals with high skills, measured as cognitive skills, non-cognitive ability, or education, in line with the fact that the city-size wage premium is higher for highly skilled individuals (see Bacolod et al., 2009). The large differences in match quality are consistent with high mobility frictions, in line with research showing that people are unlikely to move in response to economic shocks (see Yagan, 2019).

I examine two different mechanisms behind the higher match quality in larger labor markets: a more diversified occupation structure in larger labor markets, increasing the likelihood that individuals find an occupation that matches their skills and more learning possibilities about match quality in larger cities. When exploring the empirical support for these mechanisms, suggestive evidence emerges that higher match quality in larger labor markets comes from both a more diversified occupation structure and higher learning possibilities.

It should be noted that the finding that match quality is higher in larger labor markets does not necessarily mean individuals would be better off by moving to large labor markets since the utility of the individual is also affected by the cost of living and their preferences about where to live. Thus, while the result here highlights the limited ability to find a good match in smaller labor markets, research is needed on how to best solve this problem. Possible solutions could be increasing mobility or increasing the size of small labor markets. Policies for increasing the size of the local labor market could be lowering the commuting cost by investing in infrastructure or increasing the possibility of working from home.

Moreover, since this study has shown that match quality is an important factor behind the city-size wage premium, policies directly focused on increasing match quality in small labor markets could be useful in decreasing the city-size wage gap. Such policies could be increasing occupation diversity in small cities, for example by moving government agencies to smaller cities or increasing incentives for entrepreneurship in different occupations. Another policy to increase match quality in small labor markets is to decrease the cost of exploring different occupations in small labor markets by lowering search costs. All of these possible solutions probably have different pros and cons, and more research is needed to decide on the best solutions.

References

- Abel, Jaison R. and Richard Deitz (2015). "Agglomeration and job matching among college graduates". *Regional Science and Urban Economics* 51, 14–24.
- Abowd, John M., Francis Kramarz, and David N. Margolis (1999). "High Wage Workers and High Wage Firms". *Econometrica* 67.2, 251–333. eprint: https://onlinelibrary.wiley.com/doi/pdf/10.1111/1468-0262.00020.
- Almgren, Mattias, John Vincent Kramer, and Jósef Sigurdsson (2022). "It Runs in the Family: Occupational Choice and the Allocation of Talent". *Essays on Macroeconomics, Monetary Policy and Mobility*. Monograph series / Institute for International Economic Studies, University of Stockholm 117. Department of Economics, Stockholm University, 276.
- Andersson, Fredrik, Simon Burgess, and Julia I. Lane (2007). "Cities, matching and the productivity gains of agglomeration". *Journal of Urban Economics* 61.1, 112–128.
- Andersson, Martin, Johan Klaesson, and Johan P Larsson (2014). "The sources of the urban wage premium by worker skills: Spatial sorting or agglomeration economies?" *Papers in Regional Science* 93.4, 727–747. eprint: https://rsaiconnect.onlinelibrary.wiley.com/doi/pdf/10.1111/pirs.12025.
- Andini, Monica, Guido de Blasio, Gilles Duranton, and William C. Strange (2013). "Marshallian labour market pooling: Evidence from Italy". *Regional Science and Urban Economics* 43.6, 1008–1022.
- Bacolod, Marigee, Bernardo S. Blum, and William C. Strange (Jan. 2009). "Urban interactions: soft skills versus specialization". *Journal of Economic Geography* 9.2, 227–262. eprint: https://academic.oup.com/joeg/article-pdf/9/2/227/18185959/lbn057.pdf.
- Bacolod, Marigee, Jorge De la Roca, and María Marta Ferreyra (2021). "In search of better opportunities: Sorting and agglomeration effects among young college graduates in Colombia". *Regional Science and Urban Economics* 87, 103656.
- Benmelech, Efraim, Nittai K. Bergman, and Hyunseob Kim (2022). "Strong Employers and Weak Employees". *Journal of Human Resources* 57.S, S200– S250. eprint: https://jhr.uwpress.org/content/57/S/S200.full.pdf.
- Berlingieri, Francesco (Sept. 2018). "Local labor market size and qualification mismatch". *Journal of Economic Geography* 19.6, 1261–1286. eprint: https://academic.oup.com/joeg/article-pdf/19/6/1261/31235503/lby045. pdf.
- Bleakley, Hoyt and Jeffrey Lin (2012). "Thick-market effects and churning in the labor market: Evidence from US cities". *Journal of Urban Economics* 72.2, 87–103.

- Boualam, Brahim (2014). "Getting a First Job: Quality of the Labor Matching in French Cities".
- Card, David, Jesse Rothstein, and Moises Yi (2021). "Location, Location, Location". Working Papers. U.S. Census Bureau, Center for Economic Studies.
- Carlsen, Fredrik, Jørn Rattsø, and Hildegunn E. Stokke (2016). "Education, experience, and urban wage premium". *Regional Science and Urban Economics* 60, 39–49.
- D'Costa, Sabine and Henry G. Overman (2014). "The urban wage growth premium: Sorting or learning?" *Regional Science and Urban Economics* 48, 168–179.
- Dauth, Wolfgang, Sebastian Findeisen, Enrico Moretti, and Jens Suedekum (Feb. 2022). "Matching in Cities". *Journal of the European Economic Association* 20.4, 1478–1521. eprint: https://academic.oup.com/jeea/article-pdf/20/4/1478/45349346/jvac004_dauth_etal_onlineappendix.pdf.
- De La Roca, Jorge and Diego Puga (July 2016). "Learning by Working in Big Cities". *The Review of Economic Studies* 84.1, 106–142. eprint: https://academic.oup.com/restud/article-pdf/84/1/106/9113030/rdw031.pdf.
- Duranton, Gilles and Hubert Jayet (2011). "Is the division of labour limited by the extent of the market? Evidence from French cities". *Journal of Urban Economics* 69.1, 56–71.
- Eckert, Fabian, Mads Hejlesen, and Conor Walsh (2022). "The return to bigcity experience: Evidence from refugees in Denmark". *Journal of Urban Economics* 130, 103454.
- Eliasson, Kent and Olle Westerlund (Oct. 2022). "The urban wage premium and spatial sorting on observed and unobserved ability". *Journal of Economic Geography* 23.3, 601–627. eprint: https://academic.oup.com/joeg/article-pdf/23/3/601/50408541/lbac029.pdf.
- Figueiredo, Octávio, Paulo Guimarães, and Douglas Woodward (July 2013). "Firm–worker matching in industrial clusters". *Journal of Economic Geography* 14.1, 1–19. eprint: https://academic.oup.com/joeg/article-pdf/14/1/ 1/11815429/lbt009.pdf.
- Fredriksson, Peter, Lena Hensvik, and Oskar Nordström Skans (Nov. 2018). "Mismatch of Talent: Evidence on Match Quality, Entry Wages, and Job Mobility". *American Economic Review* 108.11, 3303–38.
- Frenette, Marc (2006). "Too Far to Go On? Distance to School and University Participation". *Education Economics* 14.1, 31–58. eprint: https://doi.org/ 10.1080/09645290500481865.
- Glaeser, Edward L. and David C. Maré (2001). "Cities and Skills". *Journal of Labor Economics* 19.2, 316–342. eprint: https://doi.org/10.1086/319563.

- Guvenen, Fatih, Burhan Kuruscu, Satoshi Tanaka, and David Wiczer (Jan. 2020). "Multidimensional Skill Mismatch". *American Economic Journal: Macroeconomics* 12.1, 210–44.
- Harmon, Nikolaj Arpe (2013). "Are Workers Better Matched in Large Labor Markets?" Tech. rep.
- Korpi, Martin (Oct. 2007). "Does size of local labour markets affect wage inequality? A rank-size rule of income distribution". *Journal of Economic Geography* 8.2, 211–237. eprint: https://academic.oup.com/joeg/articlepdf/8/2/211/2639039/lbm034.pdf.
- Korpi, Martin and William A.V. Clark (2019). "Migration and occupational careers: The static and dynamic urban wage premium by education and city size". *Papers in Regional Science* 98.1, 555–574. eprint: https://rsaiconnect.onlinelibrary.wiley.com/doi/pdf/10.1111/pirs.12328.
- Koster, Hans R.A. and Ceren Ozgen (2021). "Cities and tasks". *Journal of Urban Economics* 126, 103386.
- Leknes, Stefan, Jørn Rattsø, and Hildegunn E. Stokke (2022). "Assortative labor matching, city size, and the education level of workers". *Regional Science and Urban Economics* 96, 103806.
- Lindqvist, Erik and Roine Vestman (Jan. 2011). "The Labor Market Returns to Cognitive and Noncognitive Ability: Evidence from the Swedish Enlistment". *American Economic Journal: Applied Economics* 3.1, 101–28.
- Mion, Giordano and Paolo Naticchioni (2009). "The spatial sorting and matching of skills and firms". *Canadian Journal of Economics/Revue canadienne* d'économique 42.1, 28–55. eprint: https://onlinelibrary.wiley.com/doi/pdf/ 10.1111/j.1540-5982.2008.01498.x.
- Mood, Carina, Jan O. Jonsson, and Erik Bihagen. (2012). "Socioeconomic Gradients in Children's Outcomes". From Parents to Children: The Intergenerational Transmission of Advantage. Ed. by Markus Jäantti John Ermisch and Timothy M Smeeding. Russell Sage Foundation, 32–52.
- Moretti, Enrico (2011). "Chapter 14 Local Labor Markets". Ed. by David Card and Orley Ashenfelter. Vol. 4. Handbook of Labor Economics. Elsevier, 1237–1313.
- Moretti, Enrico and Moises Yi (Mar. 2024). "Size Matters: Matching Externalities and the Advantages of Large Labor Markets". Working Paper 32250. National Bureau of Economic Research.
- Neves Jr, Edivaldo C., Carlos R. Azzoni, and Andre Chagas (Oct. 2017). "Skill wage premium and city size". Working Papers, Department of Economics 2017₁9. University of São Paulo (FEA-USP).
- Odio Zúñiga, Mariana and C. Y. Kelvin Yuen (2020). "Moving for Better Skill Match". *SSRN*.

- Papageorgiou, Theodore (July 2022). "Occupational Matching and Cities". *American Economic Journal: Macroeconomics* 14.3, 82–132.
- Puga, Diego (2010). "THE MAGNITUDE AND CAUSES OF AGGLOM-ERATION ECONOMIES*". *Journal of Regional Science* 50.1, 203–219. eprint: https://onlinelibrary.wiley.com/doi/pdf/10.1111/j.1467-9787.2009. 00657.x.
- Rosenthal, Stuart S. and William C. Strange (2008). "The attenuation of human capital spillovers". *Journal of Urban Economics* 64.2, 373–389.
- Sun, Liyang and Sarah Abraham (2021). "Estimating dynamic treatment effects in event studies with heterogeneous treatment effects". *Journal of Econometrics* 225.2. Themed Issue: Treatment Effect 1, 175–199.
- Thoresson, Anna (Jan. 2024). "Employer Concentration and Wages for Specialized Workers". *American Economic Journal: Applied Economics* 16.1, 447–79.
- Wheeler, Christopher H. (2008). "Local market scale and the pattern of job changes among young men". *Regional Science and Urban Economics* 38.2, 101–118.
- Yagan, Danny (2019). "Employment Hysteresis from the Great Recession". *Journal of Political Economy* 127.5, 2505–2558. eprint: https://doi.org/10. 1086/701809.

Appendix A: Additional figures and tables

A.1 Match quality measure

Figure A1. Predicted wage with random forest vs actual wage



Notes: The figure plot predicted wages based on skills estimated with the random forest against actual wages. Wages are residualized on age and year-fixed effects. The sample includes males born between 1951 and 1976, with outcomes for the years 1996-2013. The individuals used for training the random forest are excluded from the sample.



Notes: The figure plot match quality against skills, figure a shows the result for cognitive skills and figure b shows the result for social skills. The match quality measure is standardized to have mean zero and standard deviation one. The sample includes males born between 1951 and 1976, with outcomes for the years 1996-2013.



Figure A3. Match quality and occupation size

Notes: The figure plot match quality against occupation size. The sample includes males born between 1951 and 1976, with outcomes for the years 1996-2013.

Dependent variable: Ln Wage						
	(1)	(2)	(3)	(4)	(5)	(6)
VARIABLES		O*Net	Linear return	Random forest proba- bility	Return vary with labor market	Random forest trained on large labor market
Log population size	0.039*** (0.000)	0.039*** (0.000)	0.036*** (0.000)	0.039*** (0.000)	0.010*** (0.000)	0.027***
Match quality		0.010*** (0.000)	0.207*** (0.001)	0.028*** (0.000)	0.176*** (0.000)	0.167*** (0.000)
Observations	6,005,629	5,820,642	6,005,629	6,005,629	6,005,629	6,005,629
Year Fe	Yes	Yes	Yes	Yes	Yes	Yes
Age FE	Yes	Yes	Yes	Yes	Yes	Yes
Control skills	Yes	Yes	Yes	Yes	Yes	Yes

Table A1. Different match quality measures

Notes: This table provides results estimating equation 1, with log wage as the dependent variable adding match quality as a control comparing the coefficient for local labor market size with and without controls for match quality. Column 1 shows result without controls for match quality, and column 2-6 adds different match quality measures as controls. The sample includes males born between 1951 and 1976, with outcomes for the years 1996-2013. Standard errors are adjusted for clusters at the individual level. *** p<0.01, ** p<0.05, * p<0.1

A.2 Labor market definitions

	(1)	(2)	(3)	(4)
VARIABLES	Match	Match	Match	Match
	quality	quality	quality	quality
Large labor markets	0.208***	0.226***	0.184***	0.014**
C C	(0.004)	(0.004)	(0.004)	(0.006)
Medium labor markets	0.048***	0.056***	0.036***	-0.019***
	(0.004)	(0.004)	(0.003)	(0.006)
Year Fe	Yes	Yes	Yes	Yes
Age FE	Yes	Yes	Yes	Yes
Control skills	No	Yes	Yes	Yes
Control education	No	No	Yes	No
Individual FE		No	No	Yes
Observations	6,005,629	6,005,629	6,002,946	6,005,629

Table A2. Labor market size as categorical variables: Match quality

Notes: This table provides result estimating equation 1, estimating the difference in match quality for individuals who live in a large and medium-sized labor market compared to the omitted category living in a small labor market. The sample includes males born between 1951 and 1976, with outcomes for the years 1996-2013. In column 4, where individual fixed effects are included, the age dummies are normalized to be constant between 45 and 54 to avoid multicollinearity between age and year-fixed effects. Standard errors are adjusted for 659,206 clusters at the individual level. *** p<0.01, ** p<0.05, * p<0.1

	(1)	(2)	(3)	(4)
VARIABLES	Ln Wage	Ln Wage	Ln Wage	Ln Wage
Large labor markets	0.174***	0.123***	0.110***	0.045***
	(0.001)	(0.001)	(0.001)	(0.002)
Medium labor markets	0.040***	0.019***	0.013***	-0.001
	(0.001)	(0.001)	(0.001)	(0.002)
Year Fe	Yes	Yes	Yes	Yes
Age FE	Yes	Yes	Yes	Yes
Control skills	No	Yes	Yes	Yes
Education control	No	No	Yes	No
Individual FE	No	No	No	Yes
Observations	6,005,629	6,005,629	6,002,946	6,005,629

Table A3. Labor market size as categorical variables: Wage

Notes: This table provides results estimating equation 1, with log wage as the dependent variable, estimating the difference in log wages for individuals who live in a large and medium-sized labor market compared to the omitted category living in a small labor market. The sample includes males born between 1951 and 1976, with outcomes for the years 1996-2013. In column 4, where individual fixed effects are included, the age dummies are normalized to be constant between 45 and 54 to avoid multicollinearity between age and year-fixed effects. Standard errors are adjusted for 659,206 clusters at the individual level. *** p < 0.01, ** p < 0.05, * p < 0.1



Notes: The figure shows match quality against population size in the labor market. The vertical line shows the limit for small, medium, and large labor markets. The sample includes males born between 1951 and 1976, with outcomes for the years 1996-2013

A.3 Life cycle pattern



Figure A5. Match quality over the life-cycle

(a) All individuals

(b) Stayers

Notes: The figures plot match quality against potential experience separately for individuals living in small, medium, and large labor markets. Potential experience is constructed as age-(years of education + 6). The sample includes males born between 1951 and 1976, with outcomes for the years 1996-2013. Figure (a) includes all individuals and figure (b) only includes individuals who always stay in the local labor market they were born in.

A.4 Learning

Table A4 shows estimates of regression 2, where experience is defined as years of living in a labor market since age 20.

Dependent va	riable: Match q	uality	
-	(1)	(2)	(3)
VARIABLES	All	High school	University
		or less	
Large regions	-0.010	0.056	-0.107***
	(0.028)	(0.040)	(0.038)
Medium-sized regions	-0.026	0.010	-0.113***
	(0.025)	(0.034)	(0.036)
Experience large	0.068***	0.039***	0.053***
	(0.006)	(0.009)	(0.008)
(Experience $large)^2$	-0.002***	-0.001*	-0.001**
	(0.000)	(0.001)	(0.001)
Experience medium	0.035***	0.022***	0.022***
	(0.005)	(0.007)	(0.007)
(Experience <i>medium</i>) ²	-0.001***	-0.001**	-0.001*
	(0.000)	(0.000)	(0.000
Experience	0.028***	0.015***	0.064***
	(0.002)	(0.002)	(0.004)
Experience ²	-0.000***	-0.000	-0.001***
	(0.000)	(0.000)	(0.000)
Large region#experience large	-0.037***	-0.013	-0.042***
	(0.006)	(0.009)	(0.008)
Large region#(experience $large$) ²	0.002***	0.001	0.002***
	(0.000)	(0.001)	(0.001)
Medium region#experience large	0.003	0.015	-0.006
	(0.006)	(0.009)	(0.008)
Medium region#(experience $large)^2$	-0.001	-0.001*	-0.000
	(0.000)	(0.001)	(0.001)
Large region#experience medium	0.003	-0.002	0.002
68	(0.005)	(0.008)	(0.007)
Large region#(experience <i>medium</i>) ²	-0.000	0.000	-0.001
(,F	(0.000)	(0.001)	(0.001
Medium region#experience medium	-0.025***	-0.019***	-0.029**
	(0.005)	(0.007)	(0.007)
Medium region#(experience <i>medium</i>) ²	0.001***	0.001**	0.001**
(experience meaning)	(0,000)	(0,000)	(0,000)
Large region#experience	0.017***	0.004	0.034***
Lageregioniexperience	(0.004)	(0.005)	(0.006)
Large region $\#experience^2$	-0.001***	-0.000***	-0.001***
Large regioninexperience	(0.001)	(0,000)	(0,000)
Medium region#experience	0.005	0.003	0.022***
Wediani regioni experience	(0.003)	(0.003)	(0.005
Medium region# $arnairanca^2$	0.000	(0.00+)	-0.000**
incommin regionmen perience	(0,000)	-0.000	-0.000
Individual FE	(0.000) Vec	Ves	(0.000) Ves
Observations	1 351 753	670 803	671.050
Cosci vations	1,551,755	019,005	071,950

Table A4. Learning

Notes: This table provides result estimating equation 2. The sample includes males born between 1970 and 1976, with outcomes for the years 1996-2013. Standard errors are clustered at the individual level. *** p<0.01, ** p<0.05, * p<0.1

A.5 Occupation diversity



Figure A6. Occupation diversity

Notes: The figures plot occupation diversity, where occupation diversity is defined as the share of all possible occupations at the three-digit level that exists in the labor market. An occupation is defined as existing if somebody in the occupation sample works in the occupation. Both males and females are used to construct the occupation diversity variable. To measure the outcomes, the sample includes males born between 1951 and 1976, with outcomes for the years 1996-2013. Figure (a) plots occupation diversity against labor market size, and figure (b) plots occupation diversity against match quality.





Notes: The figures plot occupation diversity, where occupation diversity is defined as the share of all possible occupations at the three-digit level that exists in the labor market. An occupation is defined as existing if somebody in the occupation sample works in the occupation. Both males and females are used to construct the occupation diversity variable. Occupation diversity is estimated by taking a random sample of 10,000 individuals from each labor market. To measure the outcomes, the sample includes males born between 1951 and 1976, with outcomes for the years 1996-2013. Figure (a) plots occupation diversity against labor market size, and figure (b) plots occupation diversity against match quality.

Essay 2: Labor-market Drivers of Intergenerational Earnings Persistence

Co-authored with Martin Nybom and Jan Stuhler

Acknowledgements: Financial support from Forte and the Ministerio de Ciencia e Innovación (Spain, RYC2019-027614-I) and Comunidad de Madrid (AMEPUC3M11 and H2019/HUM-5891) is gratefully acknowledged. We thank Olle Folke, Isaac Sorkin, Per Engzell, Guido Neidhöfer, Emran Shahe, Forhad Shilpi, Isabel Martínez, Rasmus Landersø, Arizo Karimi and seminar participants at IFAU, Uppsala University, and the Research Network for Intergenerational Mobility for helpful comments.

Introduction

Children from high-income families earn substantially higher incomes than children from low-income families (Jäntti and Jenkins 2014; Deutscher and Mazumder, 2023). A key question in the social sciences is why this is the case. Seminal contributions in the economic literature emphasize human-capital mechanisms and differences in parental investment (Becker and Tomes, 1979; Solon, 2004), skill development across childhood (Heckman and Mosso, 2014), and the role of credit constraints in these processes (Lochner and Monge-Naranjo, 2012; Lee and Seshadri, 2019). A related strand of the literature studies the roles of nature and nurture by decomposing intergenerational transmission into pre- and post-birth factors (e.g. Björklund et al., 2006; Black et al., 2005). As such, much of the existing intergenerational mobility literature focuses on how various differences in human capital accumulation in childhood translate to pay differences in adulthood. In contrast, very few studies focus on the role of labor-market mechanisms in shaping intergenerational persistence in incomes. An earlier exception is Corak and Piraino (2010), who document that high-SES children are more likely to work for the same employer as their father.

In this paper, we study the contribution of firms, sorting, and career advancement on the labor market to intergenerational persistence. Employing Swedish population-wide earnings data with both employer-employee and parentchild links, we first decompose earnings into individual and firm components using the two-way fixed effects framework of Abowd et al. (1999) (similar to e.g. Dobbin and Zohar, 2023). We show that labor-market factors, broadly speaking, and socioeconomic status (SES) gradients in sorting across employers more specifically, explain 31% of overall income persistence measured at mid-age. The contribution of firms rises to 38% if we incorporate differences in returns to experience across firm types in our analysis. The importance of such labor-market mechanisms grows over age, suggesting that differences in labor-market behavior can partly explain why SES gradients in earnings increase with age. The SES gradient in firm pay (or the "firm pay gradient") can partly be attributed to confounding sorting on skills, but at least half of the gradient remains when conditioning on a rich set of skill measures. Finally, we analyze whether there is a similar gradient in terms of the overall attractiveness of employers, including both their pay and non-pay characteristics. We find little evidence that the purely monetary focus overstates intergenerational persistence in overall welfare.

We first present a set of empirical facts on the SES gradient in firm pay premia in Sweden. We here follow a similar approach as an earlier study by Dobbin and Zohar (2023), who study the role of firms in intergenerational persistence in Israel. While the individual component is generally more important in explaining earnings inequality and persistence, the "static" firm component does account for nearly one third (31%) of the intergenerational earnings elasticity (IGE), a slightly larger estimate than Engzell and Wilmers (2024), who focus on percentile ranks, find in similar data. Interestingly, the estimated contribution of firms to intergenerational persistence is substantially higher in our data as compared to Dobbin and Zohar (2023). We provide evidence suggesting that the comparatively high quality of our data, spanning over a 35 year period, allows us to estimate firm pay premia, and their contribution to intergenerational persistence, more precisely.

The sorting of workers across firms therefore explains a substantial part of the overall transmission of income advantages from one generation to the next. The selection into higher-paying industries (such as financial sector support services, telecommunications manufacturing, or IT services) explains some of the firm pay gap, while differences in firm pay by region or firm size are less important. Jointly, region, industry, and firm size explain more than half of the SES gradient in firm pay. Where one works plays a particularly important role for intergenerational persistence at the top of the parental income distribution. While both the worker and firm components increase with parental income, the firm component increases more strongly in relative terms. The mean firm premium is twice as large in the top percentile as compared to in the 90th percentile of the parental income distribution. As such, this steep rise in firm premia in the right tail of the parent-income distribution is a potentially important explanation to the elevated persistence at the top of Sweden's income distribution, as documented by Björklund et al. (2012).

In the second part of our paper, we study how the firm pay gradient evolves over the lifecycle. Do children from high-income parents sort into higherpaying firms right at the beginning of their career, as one would expect if parents (Kramarz and Skans, 2014; Staiger, 2022) or peers (Cornelissen et al., 2017; Zimmerman, 2019) provide access to higher-paying firms? Or does the firm pay gap expand gradually over age, as one might expect if children from high-income families make faster career progress, perhaps by exploiting social and informational networks and by navigating the job search process more proficiently? We find that much of the disparities in firm pay open up right at the beginning of the career, consistent with the interpretation that children from richer parents find more favorable "entry points" to the labor market. Still, the SES gradient in firm pay widens further in the early part of the career, before stabilizing in the mid-30s. Differences in early-career progress therefore explain part of the firm pay gap, coinciding with the career stage when many workers have high earnings growth and when differences in earnings magnify relatively fast.

Indeed, high-SES children not only more frequently switch firms in the early career, but also make larger gains in firm pay conditional on switching. These results can be interpreted using the family of models that illustrate how search frictions can lead to variation in firm pay and "job ladders" (Burdett and Mortensen, 1998; Manning, 2013), suggesting that high-SES children reach higher rungs of the firm job ladder faster. All these patterns hold within

education groups, but are especially strong among college graduates. These differences in the frequency and quality of job switching also hold when conditioning on voluntary firm switches, which we define as firm-to-firm transitions without any intermediate unemployment spell. Moreover, firm pay gradients exist not only in static comparisons ("firm fixed effects"), but also in a dynamic sense: children from more privileged backgrounds sort into expanding firms that are characterized by higher wage *growth*.

To study the implication of such dynamic firm advantages, we follow Arellano-Bover and Saltiel (2021) and use a *k*-means clustering algorithm to estimate an extended two-way fixed effects model that allows for firm-class specific returns to experience. We show that children from affluent families sort into firms with substantially higher wage growth; by age 40, these dynamic firm advantages contribute 20% to their overall pay advantage. Incorporating these dynamic advantages, the total contribution of firms increases to 35% of the IGE (or 38% when netting out the AKM residuals and covariates). We therefore find that labor market sorting explains a large share of the intergenerational persistence in income.

In the third part we investigate whether high-SES children are overrepresented in better-paying firms simply because they are more skilled. It is well known that firm and individual components from the AKM framework correlate positively due to assortative matching, i.e. that more productive workers tend to sort into higher-paying firms (Card et al., 2013). Because high-SES children have on average higher individual earnings components, the SES gradient in firm pay could potentially reflect such skill sorting. In this scenario, the firm component would merely be a "mediator" of the effect of skill differences and inequities in childhood.

We first show that the SES gradient in firm pay weakens when conditioning on the worker fixed effects from the AKM model: a unit increase in log parental income increases the firm pay premium at age 40 by 3.7 rather than 5.4 log points in the unconditional case. Thus, about 30% of the SES gradient in firm pay can be attributed to skill sorting as captured by the worker fixed effects – a similar share as Dobbin and Zohar (2023) find for Israel, using the same approach. However, our main contribution here is that we can test for skill sorting more directly, as we have access to late-adolescence skill scores from the mandatory military enlistment tests. Conditioning on cognitive and non-cognitive skills measured at age 18, the residual firm pay gradient decreases to about 50% of the unconditional SES gradient, i.e. *half* of the SES gradient in firm pay at age 40 is due to skill sorting. The cognitive skill measure provides the most additional explanatory power to understand the pattern of sorting.

While this analysis confirms that an important part of the SES gradient in firm pay is due to skills, we find that family background plays an important role for labor-market outcomes even conditional on own skill. Family background effects unrelated to skill play a particularly important role in the early career: at age 25, nearly 70% of the firm pay gradient is due to such direct family effects, falling to just 50% at age 40. Parental background and networks thus provide particularly crucial advantages at labor market entry; this finding is in line with recent evidence by Kramarz and Skans (2014), San (2022), and Staiger (2022), who document the importance of parents' co-worker networks for their children's job finding. These studies focus on isolating the effects of parental networks on job finding and earnings among young workers, but do not quantify the overall contribution of parental co-worker networks to the long-run firm pay gradient. In comparison, we can quantify the overall contribution of family background over the career, but cannot isolate the contribution of parental networks from other family background effects.

In the final part we ask whether the remaining SES gradient in firm pay can be traced to differences in preferences, compensating differentials, and/or other non-wage attributes of firms. If intergenerational income transmission partly stems from inherited preferences – e.g. that some families value income and consumption compared to non-monetary attributes more than other families – then measures of income persistence would overstate the extent to which differences in welfare persist across generations. On the other hand, if children from more affluent families sort into jobs with more favorable non-monetary attributes, in addition to higher firm pay premia, then intergenerational mobility in underlying welfare would be even lower than income-based estimates suggest. To our knowledge, there exists very little evidence on this potential mechanism behind intergenerational income associations.

We test for the role of non-monetary attributes of firms, using different approaches. We first explore how firm premia and parental income relate to auxiliary measures of the attractiveness of firms, including quit rates and hiring rates from other firms (poaching). The results confirm that higher-paying firms are more desirable employers (in the sense of being able to poach workers from other firms), and that children from more affluent families sort into more desirable firms. To deepen this analysis, we employ a revealed-preferences based approach similar to Sorkin (2018). The approach exploits two-sided firm-to-firm transitions of workers over time, and infers the non-monetary values of firms from whether a firm on net gains or loses workers from firms of different values. Overall, we find little evidence on compensating differentials being systematically related to the family background of workers. Skills and other drivers of labor-market advantages seem thus more important for intergenerational transmission than correlated preferences for non-wage job attributes within families.

Our work contributes to a nascent literature on the importance of labor market factors to intergenerational persistence. An important motivation for our study is Dobbin and Zohar (2023), whose methodological approach we adopt in Section 2. Compared to their paper, we find a greater contribution of firms to intergenerational persistence; while the firm pay gradient contributes 22% to the IGE in their Israeli data, we find that firm advantages contribute 31% in our static and 38% in our dynamic model (at age 40, and net of AKM residuals and covariates). Differences in data quality and, in the dynamic model, the incorporation of firm-specific returns to experience explain most of this contrast. We further find that skill sorting explains half of the contribution of firm pay to the IGE. This estimate is in line with Dobbin and Zohar, even though we use a different approach to study skill sorting. We use our richer data to extend their analysis in various ways. We study how the SES gradient in firm pay develops over the lifecycle, study the dynamic implications of labor market sorting, and examine the role of non-wage attributes of firms.

Parts of out analysis overlaps with Engzell and Wilmers (2024), who also study the intergenerational transmission of labor market advantages in Sweden. In comparison, they bring a sociological perspective based on stratification theory and place more attention to the source of earnings in the parent generation.

A novel part of our paper is that we highlight the *dynamics* of firm pay gaps, uncovering and understanding their development over the lifecycle. Our finding of a large SES gradient in firm pay opening upright at career start is consistent with recent studies tracing the effect of social connections on worker-firm sorting in early career. Corak and Piraino (2010) document that the intergenerational transmission of employers in Canada is stronger in highincome families; Bingley et al. (2012) and Stinson and Wignall (2018) confirm these findings for Denmark and the U.S. Further, Kramarz and Skans (2014) and Staiger (2022) quantify the role of parental and family connections in the early career. While their studies provide evidence on a particular mechanism (social connections), we quantify the contribution of firms to intergenerational persistence overall, which depends also on other mechanisms, such as skill sorting. We further show that rather than skills, the firm pay gradient at career start is primarily explained by other family-related advantages (including social connections). However, skill-based sorting explains most of the increase in the firm pay gradient over age.

Our work further relates to recent studies on skill sorting on the labor market (Eeckhout 2018, Card et al. 2018). Consistent with Dobbin and Zohar (2023), we find that there is far more sorting by parental background than one might expect from skill sorting alone. We document firm sorting with respect to cognitive and non-cognitive skill measures (similar to Nybom, 2017, for college education), over and above the extent of sorting captured by worker effects from worker-firm two-way fixed effects models. Our results also relate to recent evidence showing that differences in firm pay premia contribute to earnings gaps along other dimensions, such as gender (Card et al., 2016) and race (Gerard et al., 2021). For example, Gerard et al. (2021) find that skillbased sorting contributes to about 55-65% of the firm pay gap between whites and non-whites in Brazil. While the approaches differ, our estimates of the role of skill sorting for differences in firm pay by family background in Sweden are thus rather similar. Several recent studies also highlight the the extent
to which firm pay differences contribute to cross-sectional inequality overall (e.g. around 20% in Card et al., 2013 or 11-12% in our data). Our findings suggest that differences in firm pay contributes a considerably higher share to *intergenerational* persistence in earnings.

1 Data and specifications

1.1 Data

Our empirical analyses require data on earnings, employers, education, and other background characteristics of parents and their children covering multiple decades. For this purpose, we combine a set of linkable administrative registers made available by Statistics Sweden. We use tax registers for earnings records, full-count employer-employee data to identify the firms and establishments of workers, the education register for highest level and years of education, and the multigenerational register to link children to parents.

The earnings data cover the full working population for the years 1968-2018. Our main analyses use gross annual earnings from work including self-employment, bonuses and fringe benefits, and short-term (employer provided) sickness benefits.¹ Firm and establishments including their location and industry affiliation are available for the years 1985-2018. Official birth and family registers allow us to match nearly all Swedish-born children born 1932 or later, and foreign-born children born 1961 or later, to both their parents.

Main intergenerational sample Our main analysis is based on men and women born 1967-1977, who can be observed on the Swedish labor market between age 25-41, and have at least one observed firm fixed effects at those ages. We focus on these cohorts such that we can observe a long and important part of their own labor-market career, while still being able to observe the prime-age incomes of their fathers. We measure the father's long-run income between ages 45 and 55, and consider as long-run income measures either the mean log earnings or mean earnings rank across the 45-55 age range. Fathers annual earnings observations are net of year dummies and quadratic age effects and ranks are computed relative to fathers in the main sample. We drop individuals for whom we cannot observe their father's income, which only applies to a small fraction of individuals (typically for migration-related reasons). In parts of our analysis we focus on the child generation's peak career outcomes, measuring earnings and employers at age 39-41. About 10% of the main sample drops out due to missing firm or earnings at these ages.

AKM sample. To estimate firm premia using an AKM model we use a second matched employer-employee sample, covering the entire Swedish labor

¹In all analyses we exclude very low annual earnings observations. For each year, we compute the median earnings of men aged 45, and set annual earnings observations corresponding to less than 20% of this median to missing. This ensures that our estimates are not overly influenced by variation in labor supply.

force age 20-64 in the period 1985-2018. For this sample we use information on annual earnings, firm, age, gender, and education. The estimated firm and individual fixed effects are then used as inputs in our analyses of the main intergenerational sample.

Descriptive statistics. Table 1 provides descriptive statistics of the AKM sample and our main intergenerational samples. Differences between these samples are due to differences in age composition and due to the AKM samples containing observations from earlier years. As individuals in the AKM sample are on average born 12 years earlier, the fraction with a college degree and their log earnings are slightly lower compared to the main samples. The average log firm size is 6.35, but our data also allows us to identify the worker's specific establishment within large firms.² Our full-population data contain 1,301,551 observations with positive earnings at age 39-41. Imposing non-missing father links (e.g. due to migration) and earnings of the father, this number drops to 1,076,969 and 1,059,546, respectively. The sample size further drops to 910,665 when excluding very low earnings observations (of the child or father). Lastly, requiring a valid firm connection with identified firm fixed effect (and demographic characteristics such as education) leaves us with an age 39-41 sample of 857,064 observations.

	AKM sample	Main sample (born 1967-77)		
Age	20-64	25-41	39-41	
Earnings years	1985-2018	1992-2018	2006-2018	
Log earnings	12.46	12.54	12.79	
Share women	0.49	0.50	0.49	
Share with college degree	0.38	0.47	0.49	
Year of birth	1960	1972	1972	
Log firm size	6.35	5.97	6.05	
Log establishment size	4.21	4.16	4.18	
Number individuals	7,668,377	967,417	857,064	
Number firms	341,798	228,285	118,258	

Table 1. Descriptive statistics

Notes: Descriptive statistics for different samples. Column 1 shows statistics for the AKM sample, covering individuals aged 20 to 64 born between 1922 and 1997. Columns 2 and 3 show statistics for the intergenerational sample born between 1967 and 1977, separately for the 25-41 and 39-41 age ranges.

²Average firm size is relatively large for a few reasons. First, Sweden has a large manufacturing sector with many big firms. Second, public sector workers are included in our samples, and all working for a certain municipality count as working for the same "firm". Third, small firms without sufficient firm-to-firm mobility are excluded from all samples (see below).

1.2 Estimation of Worker and Firm Fixed Effects

We use the widely applied two-way fixed effects framework of Abowd et al. (1999) to decompose earnings into firm and individual components, conditional on a set of time-varying controls. Our specification for the log earnings, y_{ijt} , of individual *i* in year *t* while employed in firm j = J(i,t) is:

$$y_{ijt} = \alpha_i + \psi_j + X_{it}\delta + \varepsilon_{ijt} \tag{1}$$

where α_i is a worker fixed effect, ψ_j a firm fixed effect ("firm pay premium"), X_{it} a vector of time-varying controls with coefficient vector δ , and ε_{ijt} an error term. The time-varying covariates in X_{it} include year dummies and a restricted set of education-gender-specific age dummies. Due to the well-known collinearity between age, cohort, and time, unrestricted age dummies are not identified. Rather than imposing a particular functional form for lifecy-cle earnings profiles, we follow Engbom et al. (2023) and assume that the age effects are constant for ages when the earnings profile is roughly flat, where we impose the age effects to be constant between ages 45 to 54. For studying lifecycle firm sorting by parental income (Section 3) it is important to account for education-specific variation in earnings over age, given that own education and parental income tend to be positively correlated. We therefore allow for earnings profiles to vary freely by education groups.

The firm pay premia ψ_i in equation (1) are identified from conditional changes in earnings as workers switch firms. In particular, they are only identified relative to some baseline firm within a set of firms connected through such firm-to-firm transitions ("movers"). Firm fixed effects identified within non-overlapping connected sets of firms are not directly comparable to each other. We therefore follow standard procedures to compute the largest connected set for our time period and drop worker-year observations for which workers are employed in firms outside of this set (approximately 0.7% of all worker-year observations). Firm fixed effects tend to be noisily estimated for firms connected to other firms by only a small number of movers (e.g. Andrews et al., 2008; Bonhomme et al., 2023; Kline et al., 2020), which leads to an inflated variance of the firm component in a variance decomposition based on equation (1). We thus drop firms (and associated worker-year observations) that are connected to other firms through fewer than five movers (an additional 6% of worker-year observations). ³ This adjustment together with the fact that we use population-wide data for a long time period should diminish concerns about such limited-mobility bias.

We estimate equation (1) using our employer-employee sample for the years 1985-2018. We focus on full-time workers, approximated by excluding workeryear observations with annual earnings lower than 20% of the yearly median earnings of 45-year old males. We also provide robustness tests using data on

³However, table A4 in appendix A.2 shows that the results remain similar when we do not exclude firms with fewer than 5 movers.

actual wage rates instead of annual earnings for a large subsample covering the same time period (see Appendix A.2). In the subsequent sections, we use our estimates of α_i and ψ_i as inputs in our intergenerational analysis.

Using our estimates from equation (1), we can decompose the variance in earnings into its different components; Appendix A.1 reports such decompositions for both the AKM and the intergenerational samples. We find that worker effects explain 29-38% while firm effects explain 7-11% of the variance in log earnings. The covariance between worker and firm effects contributes another 7-11%, reflecting the sorting of more productive workers to better-paying firms (assortative matching). Overall, our decomposition results are similar to Engbom et al. (2023), who use similar data and specifications, and also largely in line with evidence from the US (e.g. Song et al., 2019).

2 The contribution of firms to intergenerational persistence

We start by providing some basic facts on how children from different socioeconomic backgrounds sort into firms with different pay premia, as well as how such firm sorting contributes to intergenerational income persistence. To this end, we estimate variations of the linear regression

$$y_{ijt} = \alpha + \beta y_{f(i)} + u_{ijt}, \qquad (2)$$

where y_{ijt} is child log earnings, $y_{f(i)}$ the log earnings of the father of child *i*, and β is the intergenerational earnings elasticity (IGE). We measure the father's long-run income as the mean residual of log income for the ages 45 and 55, where log income has been residualized of year dummies and quadratic age effects. The childrens income are measured as mean of log income for the ages 39-41. As y_{ijt} can be decomposed according to equation (1), the slope coefficients from separate regressions of each of its components on $y_{f(i)}$ will together sum up to β . Our primary focus is on the slope coefficient from a regression of the child's estimated firm pay premium $\hat{\psi}_j$ on $y_{f(i)}$, which we denote β_{firm} and refer to as the *socioeconomic status gradient in firm pay* or, shorter, *SES gradient in firm pay*.

Table 2 shows estimates of the IGE and its components. The estimates are based on our main intergenerational sample, with earnings and firm premia measured as averages over age 39-41. As shown in column (1), the IGE is roughly 0.20 in our sample, which aligns with prior Swedish estimates of the IGE in *labor* income in pooled samples that include both sons and daughters (e.g. Brandén and Nybom, 2019; Engzell and Mood, 2023).⁴ Columns (2) and

⁴The IGE tends to be slightly higher for total income, especially for sons (e.g. Nybom and Stuhler, 2016). In our sample, the IGEs estimated separately for sons and daughters are 0.23 and 0.17, respectively.

(3) of Table 2 decompose the IGE into individual and firm components. About 60% of the IGE is attributed to the individual effects, which capture all permanent determinants of earnings (e.g. time-constant abilities and skills). The firm effects contribute 27% of the IGE, rising to 31% if we first net out the AKM covariates and residuals from y_{ijt} .⁵ The firm pay gradient explains therefore a sizable share of income persistence from one generation to the next.⁶ Its magnitude for Sweden contrasts with earlier work by Dobbin and Zohar (2023), who find that firms contribute about 22% to the IGE (after having netted out the AKM residuals) in Israel – about two thirds of the share that we find in our setting.⁷

Why do we find stronger sorting of workers across firms by family background than previous work? One possibility is that firm sorting is really stronger in Sweden than in Israel. But another possible explanation is that differences in the quality of the underlying income data explain part of this gap. In particular, while Dobbin and Zohar observe all workers in the Israeli labor market in a six-year period (2010-2015), we observe the entire Swedish labor force over a 34-year period (1985-2018), allowing us to pin down firm pay premia more precisely. When we replicate their data scenario, with an earnings panel over the years 2010-2015, we find a substantially diminished SES gradient in firm pay (see Appendix Table A2), explaining only 16% of the IGE (18% when netting out the AKM covariates and residuals). A classical measurement noise should not effect the estimate since firm pay is the dependent variable. However, figure A1 shows that the noise it not classical, instead the firm premia estimated for the short period seems to systematically overestimate the firm premia at low values of the firm premia and underestimate the value of the firm premia for high values of the firm premia. Differences in the quality of the AKM estimates therefore seems to affect the estimated firm con-

⁵The correlation between the log of father's income and $X_{it}\beta$ and ε_{ijt} explains 14% of the IGE (see column 4). Abstracting from this component, firm effects explain 27%/ (59%+27%)=31% of the IGE.

 $^{^{6}}$ As shown in Appendix Table A6, the firm pay gradient is slightly stronger for women than men.

⁷The magnitude we find is also slightly larger than in Engzell and Wilmers (2024), who use similar Swedish data but focus on percentile ranks of earnings and firm pay premia

tribution to the IGE.⁸⁹ Section 2.3 provides more evidence on the sensitivity of our estimates to sampling and specification choices.

	Dependent variable						
	<i>Yijt</i>	$\hat{\psi}_{j=J(i,t)}$					
	(1)	(2)	(3)	(4)	(5)		
$\mathcal{Y}_{f(i)}$	0.201*** (0.001)	0.118*** (0.001)	0.054*** (0.000)	0.029*** (0.001)	0.037*** (0.000)		
\hat{lpha}_i					0.151*** (0.001)		
Share of IGE Worker obs.	100% 857,064	59% 857,064	27% 857,064	14% 857,064	18% 857,064		

Table 2. Decomposition of the intergenerational earnings elasticity

Notes: Column (1) reports the estimated slope coefficient from regression (2) of mean of log child earnings at ages 39-41 on log father's earnings. Columns (2)-(4) report the slope coefficients from the corresponding regressions when decomposing child log earnings y_{ijt} according to equation (1) into individual fixed effects α_i , mean firm fixed effects ψ_j for the same ages earnings are estimated, and time-varying controls. Robust standard errors in parentheses.

Skill Sorting. While worker-firm sorting thus explains a substantial share of the IGE, such sorting might be only indirectly related to family background. Prior research has documented assortative matching between workers and firms, such that firm and individual fixed effects tend to correlate positively. This sorting of more productive workers to higher-paying firms contributes to income inequality in the cross-section, as shown in eq. (A1) and Appendix Table A1 (see Card et al., 2013; Song et al., 2019). In addition, worker-firm sorting increases intergenerational persistence, since individual fixed effects correlate positively with parental income (see column (2) of Table 2). The SES gradient in firm pay could therefore be a "mechanical" conse-

⁸While the results from the measurement error correlations are in line with the importance of high quality data to estimate the AKM, differences in firm pay for the specific periods could also potentially explain the results.

⁹The quality of the data differs also in other aspects. First, we measure parental earnings over a longer age span, between age 45-55. Second, we can retain 93% of all Swedish children born in 1967-1977 in our analysis, while Dobbin and Zohar (2023) drop about half of their sampled cohorts due to insufficient information on earnings or incomplete sets in the AKM estimation. We show in Appendix Table A3 that randomly dropping half of our sample has little effect on our estimates. Excess homogeneity in non-representative sample could bias estimates of the IGE downward (Solon, 1992), but it is less clear how it would impact the share of the IGE attributable to firms.

quence of the assortative matching between workers and firms as documented by prior research.

To abstract from the general degree of assortative matching between workers and firms, we follow Dobbin and Zohar (2023) and regress the firm fixed effect on the log of father's income *conditional* on the individual fixed effect (column (5) of Table 2). The coefficient on the individual fixed effect is large and statistically significant, reflecting assortative matching. The coefficient estimate for log of father's income now falls by about 30%, to 0.037. Part of the SES gradient in firm pay thus originates from gradients in (permanent) worker-level characteristics (e.g. skills), as captured by the individual fixed effects. Put differently, by amplifying the pay-off to skills, worker-firm sorting magnifies the well-known SES gradient in skills and human capital. According to the estimates here, this mechanism explains about a third of the firm contribution to intergenerational persistence; even after accounting for assortative matching, systematic differences in firm premia still explain roughly 18% of the IGE. The sorting of workers across firms by family background appears therefore substantially stronger than one would expect based on the observed degree of skill sorting in a worker-firm fixed effects model. However, skill sorting is difficult to measure, and we return to this question in Section 4.

2.1 Non-linear firm pay gradients

Does the sorting across firms matter more among low- or high-income families? Figure 1 shows how the expected child earnings (subfigure A) and firm premia (subfigure B) at age 40 vary across the distribution of parental income, with incomes now expressed in ranks (see the corresponding Figure A2 in the Appendix for log income). The two figures reproduce the positive relationships from Table 2, but with a gradient that is strongly increasing starting from about the 75th percentile of the parental-income distribution. The SES gradient in firm pay is qualitatively similar when conditioning on the individual fixed effect (red triangles in subfigure B), but smaller in magnitude. Moreover, the difference between the unconditional (blue circles) and conditional (red triangles) gradients increases along the distribution, implying that skill-based sorting is particularly important among richer families. Overall, the results show that sorting into high-paying firms is an especially important driver of intergenerational transmission among high-income families.

2.2 The roles of region, industry, and firm size

We next explore some potential determinants of the SES gradient in firm pay. Pay premia differ not only across firms, but also across regions and industries. For example, the industries with the highest mean firm pay premia are oil and

Figure 1. Child income and firm premia by father's income (ranks) (*a*) Child vs father income rank (*b*) Firm premium vs father's income rank



Notes: Figure (a) shows binned scatter plots of child's income rank at age 40 by father's income rank. Figure (b) shows firm fixed effects ψ_j at age 40 estimated by equation 1 and firm premia residualized on individual fixed effect, by father income rank.

natural gas extraction, financial sector support services, telecommunications manufacturing, chemical manufacturing, IT services, and research and development. Moreover, large firms tend to pay higher premia than smaller firms. The positive relationship in Table 2 could therefore, among other things, be due to children entering firms in similar industries as their fathers, or located on the same local labor markets.

Table 3 extends our prior analysis by controlling for various sets of fixed effects. Controlling for region (21 counties, column 2) diminishes the SES gradient in firm pay by roughly 20%. Redoing the main analysis within 2-digit industries (59 in total, column 3) instead diminishes the SES gradient by about 36%. Controlling for the size of the workplace has a smaller, yet non-trivial effect on the estimated SES gradient (column 4). Controlling for region, industry and size jointly has a noticeable impact, with only about 42% of the unconditional SES gradient in firm pay being left unexplained. Thus, an important share of the SES gradient can be traced to observable firm characteristics, though a similarly important part of the gradient remains unaccounted for.

2.3 Sensitivity to specification and sampling choices

We perform a number of tests to probe the sensitivity of our estimates to specification. First, one might worry that the time period we use to estimate the AKM components is *too* long; while we capture a lot of worker flows between firms, the assumption that firm-specific pay is fixed over time is less likely to hold over long time spans. Thus, we consider a modified AKM specification in which firm pay is allowed to vary over time (e.g. Lachowska et al., 2020; Engbom et al., 2023). To estimate the time-varying AKM we divide the period

Dependent variable: Estimated firm pay premium $\hat{\psi}_{i=J(i,t)}$						
	(1)	(2)	(3)	(4)	(5)	
$\mathcal{Y}_{f(i)}$	0.053***	0.041***	0.034***	0.045***	0.022***	
	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	
Region FEs		Х			Х	
Industry FEs			Х		Х	
Establishment size (log)				Х	Х	
Share of β_{firm}	100%	77%	64%	85%	42%	
Worker obs.	812,697	812,697	809,758	780,025	779,596	

Table 3. Decomposing the relationship between firm premia and parental income

Notes: Column (1) reports estimates of the slope coefficient from regressing $\hat{\psi}_j$, as estimated from equation (1), on log fathers earnings. Columns (2)-(5) report coefficient estimates from the same regression with different sets of control variables. Column (2) controls for region fixed effects (21 counties), column (3) for industry fixed effects (2-digit level, 59 industries), column (4) for log establishment size, and column (5) controls for region, industry and log establishment size simultaneously. Robust standard errors in parentheses.

1985-2018 into 4 periods, and allow the firm fixed effects to vary between the periods. As shown in Appendix Table A4, Panel A, this modified specification only renders a very marginal increase in the role of firms; we therefore continue to use the traditional AKM specification as our baseline. Second, we consider sorting across establishments rather than firms. As shown in Panel B of Table A4, establishment fixed effects explain 35% of the IGE (39% if we first net out the AKM residuals), a substantially greater share than firm fixed effects (cf. Table 2). While there are arguments in favor of using establishment codes - for example, public sector employers constitute large and somewhat artificial "firms" - we align with the bulk of the related literature and stick to using firms in our main analyses. Third, we explore whether the results are sensitive to using wages rather than annual labor earnings as outcome. Wages are observed for a subsample of workers in an employer survey. While the role of firms is considerably diminished (see Appendix Table A5), part of this decrease appears to be due to the wage sample being smaller and more selective (the wage survey oversamples workers in large firms and the public sector). We thus proceed with using earnings, as much of the prior literature, but recognize that variation in labor supply might contribute to the estimated AKM components.

We further test how our estimates vary with sampling choices. One question is whether the firm pay gradient is driven by children choosing to work in the same firm as their parents. However, while the share of "firm followers" is indeed non-negligible in our sample (in line with evidence by Corak and Piraino 2010, Stinson and Wignall 2018), the estimated firm pay gradient remains nearly unchanged when we drop them from the estimation sample (Appendix Table A6, Panel C). In contrast, dropping public sector workers raises the firm contribution by 10% (Table A6, Panel D). This might reflect that wages are more compressed in the public sector, but also that some public employers are very large, such that their firm fixed effects may not well reflect pay in different groups within the "firm". Finally, restricting the sample to firms for which we observe at least 10 or 50 "movers" in the AKM estimation decreases the firm pay gradient (Table A6, Panels E and F). On the one hand, dropping small firms with few movers decreases the limited mobility bias in AKM estimates (Bonhomme et al., 2023). On the other hand, firm fixed effects might be less informative about worker-specific pay in large than small firms. We therefore stick with our baseline specification, which retains firms with at least five movers.

3 Firm pay gradients over the lifecycle

The previous section showed that children from high-income families end up working at higher-paying firms at age 40, that this pattern holds conditional on their own permanent skills as captured by the individual fixed effects, and that this firm-sorting explains a sizable part of the persistence of income inequality across generations. This section explores the career dynamics of firm pay, with the aim to further understand how and why children from more privileged family backgrounds end up at higher-paying firms.

3.1 Firm pay premia over the lifecycle

Figure 2 plots age profiles of the mean firm pay premium by quartile of parental income. Subfigure (a) shows a clear SES gradient in firm pay already at age 25, with children from higher parental-income quartiles working at firms with higher pay premia. Further, children originating from the top parent quartile (yellow squares) improve their firm premia at a substantially faster rate than other children in the early part of their careers: while the bottom three quartiles increase their firm premium by about 1 percentage point up until the early 30s, the increase among children from the top quartile is about 3 percentage points. The gap in firm pay stabilizes in the mid-30s. The standard deviation of the firm premium is 0.14-0.15, depending on the sample (see Table A1), which implies that children in the top quartile have a firm pay advantage compared to the next quartile corresponding to about 29% of a standard deviation.

However, some of these dynamics might be driven by education-specific differences in career profiles and systematically higher education levels and delayed labor-market entry among children from well-off backgrounds. Sub-figures 2b and 2c thus reproduce the same plot separately for children with at most high school and some college or more, respectively. In both groups, a

similar type of SES gradient in firm pay is apparent. However, among noncollege children, firm premia generally grow less over age. Among college children, all quartiles grow their firm premia over age, but the top-quartile children do so at a faster rate than others. While the levels and early-career growth in firm premia are higher among college graduates, the qualitative patterns are surprisingly similar whether we consider all children or only children with similar levels of education.¹⁰

Given the stability of the patterns when comparing the full sample and education-based subgroups, we proceed to focus on the former. However, results by education are reported in Appendix A.5, and we discuss noteworthy education-related differences in the main text.¹¹ Finally, Figure A4 in the Appendix plots the difference in firm pay premia over the lifecycle conditional on individual fixed effects, to account for the general degree of assortative matching between workers and firms that is due to skill sorting (i.e., the lifecycle version of column (5) of Table 2). The gaps in firm pay premia as well as the large early-career increase in firm pay among children with high-income fathers remain largely similar.

In sum, gaps in firm pay open up already at the beginning of the career, with high-SES children being substantially more likely to work in firms with more generous pay policies. This gap increases further in the early career stages, and then stabilizes by the mid-30s. In Section 4 we show that the initial gap *cannot* be explained by differences in worker skills, while the subsequent widening of the gap is mainly explained by skill differences that correlate with parental background.

3.2 Climbing the job ladder

We showed that high-SES children not only start their career at higher-paying firms, but also that a large part of the long-term SES gradients in firm pay build up over the first 10-15 years on the labor market. This divergence can be due to either *more frequent* firm switches among high-income children (i.e., climbing the job ladder faster), or that they do *better* switches (i.e., the rungs of their ladder are further apart), or a combination of both.

Figure 3a shows the annual firm switch rate, i.e. the probability of being observed in a new firm at age *a* compared to at age a - 1, by age and parent-income quartile. As expected, all children are more likely to switch firms at early age, with roughly 20-25% of young workers annually switching firm

¹⁰This similarity reflects that time-constant differences in worker pay are captured by the individual fixed effects, while differences in lifecycle growth within firm are captured by the education-age interactions in equation (1).

¹¹To further probe that our results on career dynamics are not driven by differences in age at labor-market entry we reproduce the analysis by potential experince, finding very similar results (see Appendix A.5).



Figure 2. Firm earnings premium over the lifecycle (*a*) Full sample

Notes: The figures plots the mean estimated firm premium $\hat{\psi}_j$ over the life cycle, by quartile of father's income. Sub-figure (a) shows the result for the full sample, sub-figure (b) shows the result for children without collage education and sub-figure (c) shows the result for children with college education. Father's income quartiles are defined in the full sample.

through ages 25-30, while only 11-12% switch at age 41. Up until the early 30s, there is also a small but clear SES gradient in the likelihood to switch firm, with high-SES children being more avid switchers. These differences in the probability of switching diminish over age, and become negligible after age 35. The patterns hold within education groups (Appendix Figure A6), though overall higher education is associated with slightly more firm switching and the elevated switching of high-SES children is more pronounced in the college group.

Simply moving to a new firm is not enough to improve your firm pay, unless such moves entail improvements in firm pay premia.¹² Figure 3b plots the proportion of firm switches that are indeed premium-improving, defined

¹²While there are other potential reasons why worker mobility may benefit human capital and wage growth, we focus here on the benefits in terms of firm pay premia.

as moving to a firm at age a with an estimated firm premium that is strictly larger than the one in the previous firm at age a - 1, again computed separately by age and parent-income quartile. Children tend to switch to better firms at early age (i.e., they "climb the firm ladder"), as the chance of a "good" (premium-improving) firm switch is greater than 50% for all SES groups. But again there are distinct differences with respect to SES, with high-SES children being more likely to experience such premium-improving switches. The likelihood of good and bad switches tend to even out over age, as well as the SES gradients. Moreover, while low-SES children are more likely to switch firms than mid-SES (subfigure a), they are not more likely to gain in firm pay conditional on switching. This observation indicates that high- and low-SES children switch for different reasons, e.g. the latter might to a larger extent be involuntary switches or moves between various forms of temporary employment.

Figure 3c shows the average change in the firm premium in a similar fashion. The figure shows a clear gradient indicating that with every firm switch, high-SES children raise their earnings by about one additional percent compared to others through improved firm pay. Put differently, children from the top quartile of parental income enjoy gains in firm pay that are more than twice as large as the gains achieved by children in the lower quartiles. This difference shrinks over age, and after age 35 the effect of switches on firm pay premia is close to zero, independent of parental income. However, this closing of the gaps at later ages may be a mechanical consequence of gaps at earlier age: rapid improvements at early age should mechanically make it harder to improve your firm premium at later age, while those who remain working at low-pay firms at early age retain more scope for improvements in firm pay later on. In the extreme scenario of fully randomized switches, low-SES children with lower initial firm premia would experience much larger gains from switching.

To account for this mechanical relation between the change and level of firm pay premia, Figure 3d reports the change in firm premia while conditioning on the level of the firm fixed effect in the previous year. Thus, instead of relating the new firm premium to your own premium in the year prior, we implicitly relate them to the mean premium of the cohort in the year prior. We find that over the entire age range, high-SES children switch to better firms than children in the lower quartiles. Thus, the estimates in Figure 3c are partly compressed by the fact that premium-improving switches are increasingly hard to find for those who have already climbed high up on the firm ladder. Accounting for this mechanical relation, the SES gradient is now essentially constant over the lifecycle.

Appendix Figures A7 and A8 show the proportion of "good" firm switches and the mean change in firm pay conditional on switching separately by education group. College-educated individuals are generally more likely to experience premium-improving switches and larger boosts in firm pay through-



Figure 3. Firm switching patterns over the lifecycle (*a*) Probability to switch firms (*b*) Premium-improving firm switches

Notes: The figures show pattern for switching firms over the life-cyle, by quartile of father's income. Figure (a) shows the probability of working in a new firm compared to the year before. Figure (b) shows the probability of switching to a firm with a higher firm premium than the one before conditional on switching. Figure (c) shows the difference between the new firm premia and the firm premia before for individuals who switch firms. Figure (d) shows the same but adds a control for the firm premia in the firm before the switch.

out the career, but the SES gradients are otherwise similar across education groups. Moreover, we show that the larger gains in firm pay among high-SES children does not appear to be driven by differences in unemployment – even considering only voluntary switches without any intermediate unemployment there is an SES gradient in the quality of switches.¹³

3.3 Commuting up the job ladder?

One reason for why children from high-income families gain higher firm pay premia over the life-cycle is that they might be more likely to commute longer for work. In particular, they might have access to wider social networks and/or

¹³We can use data on UI benefit receipts to distinguish voluntary switches (without any UI receipt in between employment spells) and involuntary switches via spells of unemployment (as measured by UI benefit receipt). As shown in Figure A9, high-SES children are more likely to experience improvement in the firm premium following both voluntary and involuntary switches.

have access to support structures that make commuting easier, such as child care or different modes of transportation. By expanding the choice set of feasible jobs, commuting may increase wages and firm pay premia (Le Barbanchon et al., 2020; Agrawal et al., 2024). On the other hand, commuting is costly, both financially and in terms of time. Hence, if differences in commuting patterns can explain a sizable fraction of the SES gradient in firm pay, then gaps in firm pay premia might overestimate the implied gaps in overall welfare.

Figure 4a shows that children from high-income families are indeed more likely to commute (i.e., working and residing in different municipalities). At age 25, the share of commuters is 8 pp. (nearly 30%) higher among children in the top compared to bottom quartile of parental income at age 25, increasing to 10 pp. by age 40. Figure A11 in the Appendix shows that these findings also hold within education groups; while more educated workers are more likely to commute, differences in education do not explain the strong SES gradient in the commuter share.

Figure 4b shows that commuters earn higher firm premia, in particular at later ages. At age 40, the gap in firm premia between commuters and noncommuters reaches 6 pp., nearly half of a standard deviation. However, this gap also reflects selection effects, as workers who commute might differ in other important ways from those who do not (e.g., live in a bigger city, where it is more common to work and reside in different municipalities). To abstract from selection, Table A9 in the Appendix reports event-study type regression estimates conditional on individual fixed effects, showing how firm pay changes for individuals who begin to commute. We find on average commuting raises firm premia by about 1 pp., increasing to 1.7 pp. at age 40. Differences in commuting behavior therefore contribute to the SES gradient in firm pay, and the observed increase of that gradient over age (cf. Figure 2).



Notes: Figure (a) shows the proportion of individuals who commute (i.e., work in another municipality than they live in) over the life-cycle, by quartile of father's income. Figure (b) plots the mean firm premia by age and commuting status.

3.4 Beyond firm pay premia

Firms differ in other important dimensions apart from their pay premia. In this section, we study whether the type of firms in which children from high-income families work offer other advantages apart from higher pay. We begin by considering "static" firm characteristics, such as the composition of the workforce. Figure 5a shows that children from high-income families tend to have more productive co-workers, as captured by their estimated AKM worker fixed effects $\hat{\alpha}_i$. The mean co-worker effect is about 6 pp. higher for children from the top compared to bottom quartile of parental income. They are also exposed to a higher share of co-workers from high-income families, as shown in Figure 5b. This type of firm-level segregation is relatively stable over age – if anything, workers are more segregated by own productivity and SES at early than at later age. If working with more productive co-workers increases one's own productivity (as in Kremer, 1993), such segregation might also contribute to the firm pay advantage that we documented in section 3.1.



Notes: The figures shows firm characteristics by age, by quartile of father's income. Subfigure (a) shows the mean individual fixed effect of the co-workers. Sub-figure (b) shows the share of co-workers with fathers in the top income quartile.

The fact that high-SES children end up in firms with more productive workforces (as measured by co-worker individual fixed effects) might have interesting implications regarding their career trajectories, as there might be positive spillovers on their own productivity due to learning or fostering of valuable social networks. Moreover, firms might offer different opportunities for career development, partly because different firms might themselves grow at different rates. As a consequence, firms may have different *dynamic* implications, apart from the static difference in firm pay as captured by the AKM approach.

To illustrate that firms also differ in a dynamic sense, Figure 6 shows the mean employment and earnings growth of co-workers over the following five years (i.e., between age a and age a + 5). Figure 6a shows that children from

high-income families are more likely to work in growing firms. Indeed, children from the top quartile of father's income work in firms that grow 50-100% faster than children in the lower quartiles.

High-SES children are also more likely to work in firms with high earnings growth, as shown in the next two sub-figures. Figure 6b plots the mean earnings growth between age a and age a + 5 for co-workers who stay in the firm, while Figure 6c tracks the five-year earnings growth for co-workers irrespectively of whether they stay or leave the firm. Interestingly, the SES gradients are much more pronounced in Figure 6c, suggesting that much of the difference in co-worker earnings growth comes from mobility to new firms, rather than just differences in earnings growth for incumbent workers within firms. This finding is in line with the results shown above that children from high-SES families tend to work together, are more likely to switch firms, and make switches that render higher firm pay premia.¹⁴

3.5 Heterogeneity in firm-specific returns to experience

Individuals from high-income families are more likely to sort into high-paying firms, but firms differ in other dimensions, such as the average earnings growth of workers in those firms. Such differences in earnings growth could arise because of different learning or promotion opportunities; and while some firms might not pay that well, it could still be optimal for individuals to work in such firms if they offer higher returns to experience, increasing income later in the career.

To more explicitly estimate the role of firm-specific returns to experience, we follow Arellano-Bover and Saltiel (2021) and split our sample of workers into two random samples. Using one of the random samples we divide firms into ten classes using the distribution of stayers' yearly unexplained earnings growth using a k-means clustering algorithm. We then use the other random sample to estimate the firm-class specific returns to experience following the methodology in Arellano-Bover and Saltiel (2021). In particular, we estimate an extended two-way fixed effects framework according to:

$$y_{ijt} = \alpha_i + \psi_j + \sum_{m=1}^{K} \gamma_m Exp(m)_{it} + X_{it}\beta + \varepsilon_{ijt}, \qquad (3)$$

where $Exp(m)_{it}$ is years of experience in firm class *m* up until year (or age) *t*. As above, we include individual and firm fixed effects, such that γ_m is identified from workers who are employed in the same firm, but have earlier experience from different firm classes. $X\beta$ controls for age and year fixed

¹⁴Interestingly, the gaps between the top quartile of father's income and the rest in co-worker earnings growth and firm employment are relatively stable across age, while differences between those from the first, second, and third quartiles are generally much smaller.



Figure 6. Firm-level employment and earnings growth



Notes: The figures show the difference in firm characteristics between year *t* and year t+5 for the individuals who work in the firm at that age by quartile of father's income. Figure (a) plots the mean employment growth in the firm, by quartile of father's income. Figure (b) plots the earnings growth for coworkers who stay in the firm between year *t* and t+5. Figure (c) shows the earnings growth for all coworkers, including co-workers who stay in the firm and co-workers who switch to another firm.

effects.¹⁵ To be able to estimate equation (3), we limit the sample to workers who we can observe for their whole career, and thus focus on workers born between 1967 and 1977 observed over ages 20 to 41.

Figure 7 summarizes our results. In sub-figure (a) we show the estimated returns per year of experience by firm class m, where class 1 is the firm class consisting of firms with the highest returns to experience and class 10 consists of the firms with the lowest returns to experience. Largely similar to Arellano-Bover and Saltiel (2021), we find important differences in the returns to working in firms belonging to different classes as defined by their un-

¹⁵Since we only include observations up to age 41 we cannot use the assumption that ages is constant between ages 45-54 as we do in our main AKM specification to be able to include both age and year fixed effects. Instead we normalize age relative 40 and include second and third order polynomials of age interacted with education and gender.

explained earnings growth. However, the differences in firm-specific returns between the top and bottom classes are considerably smaller in our Swedish data compared to their data from Italy and (especially) Brazil: while workers in the top firms (firm class 1) experience an annual firm-specific boost to their earnings growth of 2.5 percentage points, the expected firm-related earnings growth in the bottom firms (type 10) is slightly negative. Note that these growth components are net of the general education-by-gender specific earnings growth component as captured by $X_{ii}\beta$.

In Figures 7b and 7c we show how the probability to work in high-return firms (classes 1 or 2) and low-return firms (classes 9 or 10) differs by SES and over age. Children from high-income families are more likely to work in firms with very high returns to experience (belonging to the top-two classes) and less likely to work in firms with the lowest returns to experience (the bottom-two classes). The gap in the probability to work in firms with high returns is evident at all ages, and peaks around age 27-28, when about 16% of top-quartile children and about 10-11% of the non-top-quartile children work in high-return firms. Differences between those from quartiles 1-3 are generally smaller; if anything, it is those from the second quartile that have the lowest (highest) chance to work in firms with high (low) returns to experience.

Figure 7d shows the estimated average annual firm-specific return at the individual's current firm at different ages, separately by father's income quartile. We can see that top-quartile children are more likely to be employed in firms with higher returns to experience throughout the observed age range. There is also a clear tendency of an increasing SES gap in returns up until about age 28-29, while the gap shrinks after that age. In terms of magnitudes, those with parents in the top quartile enjoy firm-specific returns that are on average 20% higher between age 20 and age 30 than those with parents in the bottom quartile.

Table 3 quantifies how much of the intergernational earnings elasticity at age 40 (column 1) can be attributed to the different components in equation (3). In particular, we decompose the contribution to the firm pay premium at age 40 into the early-career firm premium at age 25 (column 3) and the *change* in firm effects between ages 25 and 40 (column 4). Almost half of the SES gap in firm pay at age 40 – and 12% of the IGE – can be attributed to changes in firm pay over age, i.e. SES gaps in firm-ladder climbing (see Section 3.2). Moreover, working in firms with higher *returns to experience* explains another 7% of the intergenerational elasticity (column 5). This estimate captures the cumulative impact of the SES gaps in returns shown in Figure 7d.

Since the identification of the return to the firm-specific experience comes from workers who currently are employed in the same firm but have earlier experience from different firm classes, the return component does not capture returns that may crystalize only from moving to other, better-paying firms. Thus, the contribution of returns to firm classes in column 5 of Table 3 can be seen as a lower bound of the contribution of firm-specific returns. Indeed, Fig-



Figure 7. Work in high- vs- low-return firms (*a*) Returns in different firm classes (*b*) Probability to work in high-return firms

Notes: Sub-figure (a) shows estimates of the coefficients on firm-class experience from equation (3). Sub-figures (b) and (c) show the probability to work in the firm class with the highest returns (firm types 1 or 2) and lowest returns (firm types 9 or 10), by quartile of father's income. Sub-figure (d) shows the average firm-specific returns to experience, by quartile of father's income.

ure A12b in Appendix A.8 shows that there is indeed a positive relationship between the returns to firm experience γ_m (as captured by column 5) and the change in firm premia (column 4), indicating that firm-specific returns to experience could also explain why children from high-income parents climb the firm ladder faster. Taken together, this dynamic view of firm pay suggests that firms play an even larger role for prime-age intergenerational earnings persistence than the static decomposition in Table 2. Adding columns 3-5 suggests that firms can explain at least 35% of the IGE (or 38% net of column 6).

One worry is that some of these dynamics are driven by the fact that children from high-income families always have higher returns to experience and that they are sorted into particular types of firms. The differences in returns across firm classes as shown in Figure 7a might therefore reflect heterogeneity across individuals rather than firms. Arellano-Bover and Saltiel (2021) test for this concern by interacting firm-specific experience with worker-fixed effects. We

	Dependent variable					
	<i>Yijt</i>	\hat{lpha}_i	$\hat{\psi}_{j=J(i,t)}$ at age	$\Delta \hat{\psi}_{j=J(i,t)}$ age	Returns to firm	$X_{it}\hat{eta} + \hat{arepsilon}_{ijt}$
	(1)	(2)	25 (3)	25-30 (4)	experi- ence (5)	(6)
$\mathcal{Y}_{f(i)}$	0.178*** (0.002)	0.099*** (0.001)	0.029*** (0.001)	0.021*** (0.001)	0.012*** (0.000)	0.015*** (0.001)
Share of IGE Worker obs.	100% 284,318	56% 284,318	16% 284,318	12% 284,318	7% 284,318	8% 284,318

 Table 4. Decomposition of the intergenerational earnings elasticity at age 40

Notes: Column (1) reports the estimated slope coefficient from a regression of log child earnings at ages 40 on log father's earnings. Columns (2)-(6) report the slope coefficients from the corresponding regressions when decomposing child log earnings y_{ijt} according to equation (3) into individual fixed effects α_i , firm fixed effects, where the firm fixed effect is divided into firm fixed effect at age 25 and the change in firm fixed effects between age 25 and 40, return to firm-specific experience and time-varying controls. The sample differs from the main sample since (3) is estimated using workers born between 1967-1977, to be able to observe the workers' whole career, and half of the sample is used to cluster firms into firm classes and half of the sample us used to estimate (3). The sample in the table is restricted to individuals who have an observed firm at age 25. Robust standard errors in parentheses.

follow a similar approach and allow the returns to experience to vary with the father's earnings quartile to estimate the regression:

$$y_{ijt} = \alpha_i + \psi_j + \sum_{m=1}^K \gamma_m Exp(m)_{it} + \sum_{m=1}^K \delta_m Exp(m)_{it} * \theta_i + X_{it}\beta + \varepsilon_{ijt}, \quad (4)$$

where θ_i is a vector of dummy variables indicating the quartile of the income of the father. Thus, we allow the returns to experience to vary depending on the father's income. Indeed, Figure 8 shows that children from high-income families have higher returns to experience *within* all firm types. However, all children independent of family background benefit from working in firms with higher returns, and the variation in returns across firm classes is nearly as large as in our estimates based on equation (3) that did not allow for returns to vary with the father's earnings quartile (cf. Figure 7a). In fact, for children from the top quartile of father's earnings, returns vary *more* when allowing for variation by family background.



Figure 8. Returns in different firm classes by father's income

Notes: The figure shows estimates of the coefficients on firm-class experience from equation 4, by quartile of father's income.

4 The role of sorting across firms by skill

We found that some of the SES gradient in firm pay is due to assortative matching between firms and workers. Children from high-income families tend to be more educated and productive (according to the permanent worker component), and prior research shows that workers with higher permanent productivity (or skills) tend to sort into firms that pay higher premia (e.g., Card et al., 2013). But as we also showed, two thirds of the SES gradient in firm pay remains when controlling for the individual fixed effects from the AKM regression (see Table 3, column 4).¹⁶ In the remainder, we refer to this specification as the "conditional firm pay gradient".

There are a few potential interpretations as to why much of the firm pay gradient remains, even conditional on this proxy for individual skills. First and foremost, parental income could indeed have a direct effect on firm sorting beyond what is mechanically driven by skill advantages among their children. Such direct effects could arise due to multiple sources, including informational advantages, social and co-worker networks, and credit or other constraints.¹⁷ They could also arise if preferences for non-monetary amenities differ across families of different income levels, such that part of the gradient in firm pay

¹⁶Note that we condition on the *estimated* individual fixed effect from the main AKM regression. Thus, we do not include a new set of individual fixed effects in the the regression of estimated firm pay premia on (log) parental income, which would obviously be collinear with parental income.

¹⁷For example, credit constraints in early age could force poorer children into safe but lowpaying jobs (see Staiger, 2022).

reflects compensating differentials or other non-pay attributes of firms (see Section 5).

However, a second possibility is that the estimated worker effects $\hat{\alpha}_i$ that we condition on are incomplete measures of skill, and therefore do not capture the full extent of skill sorting. First, the estimates $\hat{\alpha}_i$ are only a noisy measure of the α_i component in the AKM model, and such measurement error will bias the estimated contribution of assortative matching to the firm pay gradient (as shown formally by Dobbin and Zohar, 2023). Second, the individual fixed effects α_i capture *all* persistent within-firm differences in earnings, not just those that are due to differences in productivity. For example, persistent taste-based discrimination between ethnic groups would here be falsely interpreted as skill differentials, contributing to skill sorting.¹⁸ A third but related possibility relates to the multidimensionality of skills. One may surmise that worker-firm sorting is on specific dimensions of skill (e.g., cognitive skills) rather than the entire bundle of skills that contributes to the individual fixed effects in log earnings – and parental background might be more or less strongly associated with those specific sorting dimensions than with other dimensions of skill.

To analyze skill sorting more thoroughly, we consider data on cognitive and non-cognitive skills from military enlistment tests, which we use in addition to our fixed effects-based measures.¹⁹ Specifically, we use a decomposition similar to Gelbach (2016) and Hjorth-Trolle and Landersø (2023) to study sorting on (i) cognitive skills, (ii) social skills, (iii) education and (iv) the estimated individual fixed effect from the AKM model. As the enlistment test was only mandatory for males, we restrict the analyses in this section to males.²⁰ The

¹⁸Moreover, Dobbin and Zohar (2023) note that the AKM worker effects α_i may reflect "social capital", if parents help their children not only to secure a job in better-paying firms, but also to be promoted to better-paying jobs within those firms. While plausible, Stinson and Wignall (2018), San (2022) and Staiger (2022) find that most of the gains from parental networks come from working at a high-wage firm rather than from wage advantages within the firm. And in principle, parental connections might even have a negative effect on wages. For example, Bello and Morchio (2022) predict that "occupational followers" who choose their father's occupation earn lower wages, due to skill mismatch. To address these potential limitations in the interpretability of the AKM fixed effects α_i , we consider here more direct measures of skill.

¹⁹Dobbin and Zohar (2023) implement two alternative approaches to study the role of assortative matching. First, they develop an instrumental variable approach that uses the child's education as an instrument for their worker fixed effect α_i , which under plausible assumptions provides an upper bound for the contribution of assortative matching to the firm pay gradient. Second, they use education and demographic group as observable proxies for human and social capital, which under alternative assumptions also provides an upper bound for the contribution of assortative matching.

²⁰The military tests are taken at around age 18 and were compulsory for all men in the cohorts that we study. The overall cognitive skill score represents an aggregated score from four subtests that measure verbal, logical, spatial and technical skills. The non-cognitive/social test score is based on a half-hour semi-structured interview with a certified psychologist who grades the enlistee along dimensions such as leadership, social maturity, and emotional stability, but also in an overall sense (for further details, see e.g. Lindqvist and Vestman, 2011). We standardize

skill measures from this test are highly informative about labor productivity, as demonstrated by their strong associations with wages and other long-term labor-market outcomes (Lindqvist and Vestman, 2011; Nybom, 2017). This allows us to directly test the hypothesis that the AKM fixed effects that are typically conditioned on are incomplete measures of skill, such that sorting is underestimated. Moreover, we can compare the relevance of different dimensions of skills (cognitive, social, etc).

The decomposition, summarized by the regression equations (5a)-(7), parses out how much of the relationship between children's firm premia and parents' log income can be attributed to various factors influencing child log income. Having estimated the firm pay gradient β_{firm} in (5a), we then augment this regression with our four mediators of interest: cognitive skill, social skill, education, and the AKM individual fixed effect.

$$\hat{\psi}_{j(it)} = \mu_{\psi} + \beta_{firm} y_{f(i)} + \omega_i \tag{5a}$$

$$\hat{\psi}_{j(it)} = \mu_{\psi,res} + \beta_{firm,res} y_{f(i)} + \beta_{cog} cog_i + \beta_{soc} social_i + \beta_{edu} edu_i + \beta_{akm} \hat{\alpha}_i + \upsilon_i$$
(5b)

The coefficient $\beta_{firm,res}$ in this augmented regression (5b) captures the "direct" effect of family background not mediated by skills, while $\beta_{firm} - \beta_{firm,res}$ captures the part explained by the mediators. We then run a set of auxiliary regressions (6a)-(6d) to pin down how closely related each of the mediators are to parental income,

$$cog_i = \mu_{cog} + \phi_{cog} y_{f(i)} + \varepsilon_{1i}$$
 (6a)

$$social_i = \mu_{soc} + \phi_{soc} y_{f(i)} + \varepsilon_{2i}$$
 (6b)

$$edu_i = \mu_{edu} + \phi_{edu} y_{f(i)} + \varepsilon_{3i}$$
(6c)

$$\hat{\alpha}_i = \mu_{akm} + \phi_{akm} y_{f(i)} + \varepsilon_{4i} \tag{6d}$$

The contribution of the respective mediators to β_{firm} then together sum up to $\beta_{firm} - \beta_{res}$,

$$\beta_{firm} - \beta_{firm,res} = \beta_{cog}\phi_{cog} + \beta_{soc}\phi_{soc} + \beta_{edu}\phi_{edu} + \beta_{akm}\phi_{akm} \tag{7}$$

Table 5 presents the results from this decomposition. We again focus on firm premia observed at age 40. Column (1) replicates the baseline estimate of the relationship between firm pay and (log) parental income, β_{firm} , for the sample of males with observed skill measures. The estimate for this sample of enlisted males (0.056) is very similar to the ones for all males (0.058, see Table A6) and the full population (0.054, see Table 3). Column (2) shows that nearly half of the SES gradient in column (1) is due to sorting on our various skill measures. On the flip side, about 52% of the estimated β_{firm} cannot

both the overall cognitive and non-cognitive scores to mean zero and standard deviation one, separately for each birth year.

	Firm pay premia						
	Overall	Unexp.	Cognitive	Social	Education	Indiv. FE	
	eta_{firm}	β_{res}	$eta_{cog} \phi_{cog}$	$\beta_{soc}\phi_{soc}$	$eta_{edu} \phi_{edu}$	$\beta_{iFE}\phi_{iFE}$	
	0.056***	0.029***	0.011***	0.001***	0.004***	0.012***	
	(0.001)	(0.001)	(0.000)	(0.000)	(0.000)	(0.000)	
Share	100%	51.8%	19.6%	1.8%	7.1%	21.4%	
Obs.	371,411	371,411	371,411	371,411	371,411	371,411	

Table 5. Skill vs. SES gradient in firm pay

Notes: The table reports estimates from the decomposition outlined by equations (5a)-(7) using the main sample, but excluding women and males with missing enlistment scores. Standard errors obtained by 250 bootstraps.

be explained by skills, suggesting that parental background has a substantial direct effect on firm sorting beyond what is mechanically driven by gaps in the skills of children. Further, columns (3)-(6) show the contribution of each of the respective skill measures to overall sorting (i.e. the components $\beta\phi$ from equation (7)). The individual fixed effects and cognitive skills are both key in explaining why children from richer families work at better-paying firms, with each contributing about 20% to the firm pay gradient β_{firm} . While the education and social/non-cognitive skill components are substantially smaller, they are not inconsequential and together contribute nearly 10%.²¹

Table 5 confirms that the role of skill sorting is underestimated when approximating skill solely by the estimated fixed effects from an AKM model (as conjectured by Dobbin and Zohar, 2023). In Section 2, we reported a conditional firm pay gradient of roughly 0.037, corresponding to an unexplained part of 68.5% (0.037/0.054), increasing to 70.7% when considering males (0.041/0.058, see Appendix Table A6). But when adding our full set of skill measures, the unexplained part drops to 51.8% (0.029/0.056). Cognitive skills are particularly important, explaining 20% of the SES gradient in firm pay, even conditional on estimated individual fixed effects and other controls. However, our analysis also indicate that despite a rich set of skill measures, half of the parental-income gradient in firm pay cannot be explained by skills, thus suggesting that parental income plays a key role for firm sorting beyond what is "mechanically" driven by child skills.

²¹That education contributes much less to the firm pay gradient is interesting, given that it correlates *more* strongly with parental income than the other mediators (see Appendix Table A7). One possible interpretation is that the worker fixed effects from the AKM model capture differences in formal education better than differences in cognitive skills, but that the latter are an important determinant of worker-firm sorting.

Next, Figure 9 and 10 show the same type of decomposition but across the distribution of parental income and over child age, respectively. To analyze how skill sorting varies with parent income, we follow Hjorth-Trolle and Landersø (2023) and run local-linear regressions of equations (5a)-(7) at each decile of parental income.²² Figure 9 again demonstrate that the firm gradient generally increases in strength along the distribution, reaching almost 0.1 at high levels of parental income. Moreover, the unexplained part not attributable to sorting grows in absolute size along the distribution but decreases as a share of $\beta_{premium}$. The cognitive skill measure and the individual fixed effect grow steadily in importance, and higher up in the distribution they are important mediators for the raw relationship between firm premia and parental income.



Figure 9. Local linear decompositon of skills along the parent income distribution

Notes: The figure shows local linear regressions, from the decomposition outlined by equations (5a)-(7) using the main sample, but excluding women and males with missing enlistment scores. The local linear regression estimates a linear regression around the income mean at each decile of parental income, using a bandwidth of 50 000 Swedish kronor and an epan kernel.

Figure 10 shows the decomposition separately for each age over the lifecycle. The part of the firm pay gradient explained by skill sorting, $\beta_{firm} - \beta_{firm,res}$, increases substantially over age. In contrast, the direct effect of family background not mediated by skills, $\beta_{firm,res}$, is already large at age 25 and

²²The local linear regression estimates a linear regression around the income mean at each decile of parental income, using a bandwidth of 50 000 Swedish kronor and an epan kernel.

grows only slightly in size over age. Its relative contribution to the overall firm gradient β_{firm} decreases substantially: nearly 70% of the firm pay gradient at age 25 is due to direct family effects, falling to just 50% at age 40. The finding that family background effects unrelated to skill play a relatively more important role in the early career is intuitive, as children are then likely more closely linked to their parents and parental networks and contacts might be more useful. As children age, skills become more important for firm sorting, contributing further to the overall influence of family background on firm sorting.





Notes: The figure shows age specific estimates from the decomposition outlined by equations (5a)-(7) using the main sample, but excluding women and males with missing enlistment scores. The different skill part sum up to the SES gradient in firm premia.

5 Do firm pay reflect compensating differentials?

Pay is not the only firm attribute that matters to workers, and other aspects of the firm – such as its location, average workloads or fringe benefits – also vary across firms. It is therefore not necessarily the case that high-paying firms are more desirable firms. As these other aspects may also co-vary with parental background, the SES gradient in firm pay may under- or overstate the role of firms in the intergenerational transmission of advantages. To fix ideas, decompose the firm pay premium as

$$\psi_j = r_j - \kappa a_j \tag{8}$$

where the pay premium of firm *j* is the sum of a firm-specific worker rent r_j and an amenity component, κa_j . We think of r_j as arising when a firm has

some monetary rents to share with its workers, e.g. due to frictions and/or other forms of imperfect competition. The amenity a_j can be either positive, if the firm offers attractive non-pay attributes (e.g. temporal flexibility), or negative, if the firm offers bad non-pay attributes (e.g. poor work environment). Together with the non-negative amenity price $\kappa > 0$, which depends on the marginal worker's preferences, the amenity component, κa_j , thus constitutes a firm-specific compensating differential. We thus assume that a_j is perfectly priced and paid for on the market, and thus will be uncorrelated with worker utility. Consequently, if high-premium firms are overall more desirable it points to the importance of rents, while a weaker association between firm pay premia and the actual attractiveness of firms is evidence of compensating differentials.

Whether the SES gradient in firm pay is primarily due to rents or amenity compensation is crucial for its interpretation, and for the interpretation of intergenerational mobility estimates more generally. Perhaps the SES gradient arises because high-income families place a stronger value on consumption and/or are less averse to bad working conditions. In this scenario, children from high-income families end up in better-paying firms, but those firms are actually worse in other dimensions: measures of intergenerational income persistence would then overstate the extent to which levels of welfare persist across generations. Alternatively, high-income families are better equipped with contacts and networks, information, or other resources, and therefore end up in firms that pay their workers more, conditional on the level of amenities they offer. In this scenario, and if the SES gradient in amenity compensation is small, the gradient in firm pay would approximate the corresponding gradient in welfare. Or maybe firms with high pay tend to be *better* in other dimensions, too, and those non-monetary attributes of firms or jobs are generally better in high-income families. Indeed, given their more favorable financial position, children from high-income families might systematically select into firms that have worse pay but better non-pay attributes, all else equal. In this scenario, intergenerational mobility in underlying welfare would be even lower than income-based estimates suggest.

The decomposition above assumes that amenities are always fully priced into pay, which might not be an accurate description of the world. We can therefore consider an extended decomposition:

$$V_j = \psi_j + \kappa a_j + \kappa b_j = r_j + \kappa b_j \tag{9}$$

where we now focus on the overall value of a firm j, denoted V_j , which depends on the rent-part of the firm pay premium and a second component b_j capturing non-pay characteristics of the firm that are not priced into the worker's pay. Note that if a_j is correctly priced then this part of the pay premium has no influence on the value (or utility) of working for a firm. Thus,

the overall value of the firm depends potentially on rents and amenities that are "free of charge" (or at least imperfectly priced).

With some way of inferring the overall value or attractiveness of firms (the V_j), we can address a couple of key questions. First, we can explore whether high-paying firms in general also are more desirable firms. The extent to which higher-paying firms are more desirable firms can then be seen as evidence of rents, while the extent to which this is not the case is evidence of compensating differentials (Sorkin, 2018). In the extreme case, if variation in firm pay is solely due to compensating differentials, there is no relation between the value of firms and their pay premia. Second, we can study whether there is an SES gradient in firm values in a similar fashion as we did for firm pay. By further conditioning on α_i we can infer to what extent the SES gradients in firm value arise from skill sorting or not. Finally, under assumptions of $Corr(a_j, b_j)$, we can estimate SES gradients in firm value conditional on the firm pay premium and infer whether b_j is systematically related to SES.

To explore the different sources to the SES gradient in firm pay, the overall value of firms, and the role of compensating differentials, we use various alternative analyses that all involve using some proxy measure of V_j . Many of the analyses exploit worker transitions between firms, and it thus becomes crucial to distinguish voluntary from involuntary employer-to-employer transitions. Our main strategy here is to focus on voluntary moves, which we define as transitions without any intermediate period of unemployment.²³ First, we explore how parental income relates to alternative measures of a firm's attractiveness that also capture non-pay characteristics, such as the firm's "poaching" and quit rates. We then employ a revealed-preferences based approach similar to Sorkin (2018), which infers the overall values of firms (V_j) from worker transitions between firms.

5.1 Alternative measures of the attractiveness of firms

Are high-paying firms indeed more desirable firms? Figure 11 shows that new hires in high-paying firms mostly arrive from employment in other firms, i.e. the new employees have been "poached" from other firms. In contrast, low-paying firms often hire individuals from non-employment, who are less likely to have strong outside options at the time of their hire. This pattern is consistent with the "job ladder" from standard search models (Burdett and Mortensen, 1998), and indicates that high-paying firms are indeed more attractive from the perspective of workers.²⁴ Given our simple decomposition

²³We identify unemployment periods using data on UI benefit receipts, and include in our analyses only employer-to-employer transitions associated with zero received benefits.

²⁴However, while this relationship may hold on average, it does not necessarily follow that it also holds for the SES-gradient in firm pay (i.e., the way children from high-SES background select into firms may differ from the average relations observed in the labor market). In the next section, we quantify the attractiveness of each firm, to then study this question in more details.

above, the result would be inconsistent with that most or all of the firm pay premium reflects compensating differentials (priced non-pay characteristics).



Notes: Binned scatterplot of the share of hires from employment ("poaching rate") on the firm fixed effects estimated based on the AKM model in equation (1).

In Figure 12a, we study whether the firms' poaching rate varies systematically by parental background. Indeed, we see a similar pattern by parental income in the poaching rate as we do in firm pay premia (see Figure 2a): the gaps open up already at early age, and widen further up to the mid 30s. However, Figure 12b shows that *conditional* on firm pay, high-SES children end up in firms with slightly *lower* poaching rates. This might indicate that those firms are not as desirable as they seem in terms of firm pay, although the gaps in the conditional poaching rate in Figure 12b are much smaller than the unconditional gaps in Figure 12a. Overall, we find that high-SES children do sort into more attractive firms (as proxied by firm poaching rates).

In Figures 12c and 12d we show the corresponding pattern in the *quit rates* of firms. Again, the idea is that firms that for monetary or non-monetary reasons are attractive employers will see fewer quits among their existing workforce, as it is harder for other firms to poach workers from these firms. The evidence here is more mixed: early in their careers, children from the bottom but also from the top SES quartile work in firms that have slightly *higher* quit rates. The quit rates generally fall with age, and so do the gaps between SES groups. The patterns are also less sensitive to the inclusion of firm fixed effects.



Notes: The figure shows "poaching" and quit rates over the lifecycle by father's income quartile. Figure (a) shows the mean poaching rate, defined as the share of hires from employment, figure (b) shows the same poaching rate but controlling for the estimated firm premium. Figure (c) shows the quit rate at the firm and figure (d) shows quit rates controlling for the estimated firm premium.

5.2 Inferring SES gradients in firm values using revealed preferences

While the previous analyses of poaching and quit rates provided insights, a concern is that these measures are imperfect proxies of the overall values of firms. We therefore employ a more comprehensive way of inferring firm values, or V_j from above, using the revealed-preference measure of firm values based on worker flows across firms suggested by Sorkin (2018). One can think of Sorkin's approach as the Google PageRank algorithm, but for firms. In short, the idea is that if workers voluntarily move from one firm to another it must imply that the value of the destination firm is higher. We define voluntary employer-to-employer transitions as transitions where workers receive no unemployment benefits or have a year of zero earnings, in between adjacent employment spells associated with different firms. We thus retrieve a

value for each firm, which could consist of both rents and (non-priced) amenities. While showing that children from high-SES families work in firms with higher pay premia is not necessarily proof of them being better off in a welfare sense, studying the same gradients in terms of firm values allows us to draw inference about welfare differences. Table 6 shows results for how firm values relate to father's log income.

	Dependent variable						
		\hat{V}_j (2)	\hat{V}_j (3)	\hat{V}_j (4)	\hat{V}_j (5)		
$\overline{\mathcal{Y}_{f(i)}}$	0.063*** (0.001)	0.034*** (0.004)	0.016*** (0.004)	0.016*** (0.004)	0.006 (0.004)		
$\hat{\psi}_{j=J(i,t)}$	~ /			0.286*** (0.012)	0.233*** (0.013)		
\hat{lpha}_i			0.148*** (0.007)		0.110*** (0.007)		
Observations	356,774	356,774	356,774	356,774	356,774		

Table 6. Firm values and compensating differentials

Notes: Column (1) reports the slope coefficient from regressing $\hat{\psi}_j$ from equation (1) on father's log income for the subsample of firms that are included in the model to estimate the firm values. Column (2) reports the slope coefficient from regressing the estimated firm value, \hat{V}_j , following Sorkin (2018), on father log income. Columns (3)-(5) show slope coefficient estimates from regressing firm values on father log income including different controls. Column (3) controls for $\hat{\alpha}_i$, column (4) controls for $\hat{\psi}_j$, and column (5) controls for $\hat{\psi}_j$ and $\hat{\alpha}_i$ simultaneously. Robust standard errors in parentheses.

Because we cannot retrieve estimates of V_j for all firms – they need to be a part of a more restrictive *strongly* connected set in terms of voluntary firm-to-firm transitions (see Sorkin, 2018) – the sample size is reduced considerably. In column 1, we thus re-estimate the SES gradient in firm pay premia, which is slightly larger in this more restricted sample than in our baseline (.063 vs. .057). In column 2, we then document that the SES gradient extends to the estimated firm value, and thus the overall desirability of the firms. Thus, it seems like sorting across employers actually make children from high-SES families better off. When controlling for the individual fixed effect (column 3), the SES gradient is substantially weakened but remains positive and significant. Thus, one reason why high-SES children are able to enter more desirable firms is that they have higher skills, which enable them to sort into higher value firms. It is notable that skill sorting appears more important for firm values than for firm pay premia, which is intuitive if people use their skills to maximize their

overall welfare (rather than just their pay check).²⁵ However, even conditional on skills, high-SES children are considerably more likely to end up in higher value firms.

The firm value consists of both rents and non-priced amenities, and without further assumptions we cannot distinguish the relative roles of these components. The results are also silent about whether high-SES children experience positive or negative amenities — it only says that the value of the firm premium is larger than any potential negative amenities. Thus, and together with the fact that firm pay premia and firm values are positively correlated, there is evidence that not all of the variation in pay premia is due to compensating differentials – there is some room for rents.

In column 4, we show the SES gradient in firm value conditional on the firm premium, thus comparing children from different backgrounds that work in firms with similar pay premia. In doing so, the SES gradient in terms of firm value decreases in size but remains positive. Thus, when working in firms with similar pay premia, people from high-SES families are able to enter firms with relatively higher rents and non-priced amenities than priced (bad) amenities compared to low-SES children. However, without further assumptions we cannot know if this is since high-SES children earn higher rents or higher non-priced amenities given a certain firm premium. Obviously, we might assume away non-priced amenities, and define the firm value as a proportional function of firm-specific rents. In that case, the estimates in column 4 would suggest an SES gradient in rents, even conditional on firm pay premia. However, when we in addition control for the individual fixed effect (column 5), the SES gradient becomes insignificant, indicating that the main reason that high-SES children are able to enter high-value firms, given the firm premium, is skill sorting.

6 Conclusions

This paper examined the extent to which the sorting of workers across firms contributes to intergenerational earnings persistence. We build on the large literature on the drivers of intergenerational persistence. While the literature has traditionally focused on childhood development and inequalities in parental investments in their children's human capital, we add by providing a labor-market perspective. In particular, we use Swedish administrative data and decompose earnings into permanent individual components (approximating productivity) and firm-specific pay premia a la Abowd et al. (1999) and many others in their footsteps. We then add data enabling us to link parents to children, and provide a multitude of evidence on the SES gradient in the firm

 $^{^{25}}$ Controlling for the individual fixed effect diminishes the SES gradient in firm value by 53% but the firm pay premium "only" by 31% (see Table 2).

portion of pay, how it evolves over the lifecycle, its underlying drivers, and how the gradient ought to be interpreted.

Our findings indicate that disparities in firm pay premia can account for a significant portion of the intergenerational elasticity of income in Sweden. This suggests that the advantages or disadvantages associated with people's family backgrounds can have lasting impacts on their career trajectories and long-run outcomes in life. The emergence of SES gaps in firm pay already at the outset of one's career implies that individuals from more privileged backgrounds have access to more favorable entry points into the labor market. These advantages are compounded by the fact that they are able to climb the firm pay ladder faster, frequently switching employers and securing higher pay gains conditional on such changes. While skill sorting - the fact that high-SES children tend to have higher skills, and highly skilled people sort into better firms – accounts for a sizable portion of the widening of these pay gaps, a large share of the SES gradient in firm pay remains also conditional on a very detailed set of controls for skill. Furthermore, our results remain robust even after accounting for compensating differentials and alternative measures of firm quality. Thus, high-SES children sort into firms that not only deliver larger pay checks, but ultimately also higher overall welfare.

References

- Abowd, John M, Francis Kramarz, and David N Margolis (1999). "High wage workers and high wage firms". *Econometrica* 67.2, 251–333.
- Agrawal, David R, Elke Jutta Jahn, and Eckhard Janeba (2024). "Do Commuting Subsidies Drive Workers to Better Firms". Tech. rep. LASER Discussion Paper No. 149.
- Andrews, M. J., L. Gill, T. Schank, and R. Upward (2019/04/26 2008). "High Wage Workers and Low Wage Firms: Negative Assortative Matching or Limited Mobility Bias?" *Journal of the Royal Statistical Society: Series A* (*Statistics in Society*) 171.3, 673–697.
- Arellano-Bover, Jaime and Fernando Saltiel (2021). "Differences in On-the-Job Learning Across Firms".
- Becker, Gary S. and Nigel Tomes (1979). "An Equilibrium Theory of the Distribution of Income and Intergenerational Mobility". *Journal of Political Economy* 87.6, 1153–1189.
- Bello, Salvatore Lo and Iacopo Morchio (2022). "Like Father, Like Son: Occupational Choice, Intergenerational Persistence and Misallocation". *Quantitative Economics* 13.2, 629–679.
- Bingley, Paul, Miles Corak, and Niels Westergård-Nielsen (2012). "The Intergenerational Transmission of Employers in Canada and Denmark". From Parents to Children: The Intergenerational Transmission of Advantage. Ed. by John Ermisch, Markus Jantti, and Timothy M. Smeeding. New York: Russell Sage.
- Björklund, Anders, Mikael Lindahl, and Erik Plug (2006). "The Origins of Intergenerational Associations: Lessons from Swedish Adoption Data". *Quarterly Journal of Economics* 121.3, 999–1029.
- Björklund, Anders, Jesper Roine, and Daniel Waldenström (2012). "Intergenerational Top Income mobility in Sweden: Capitalist Dynasties in the Land of Equal Opportunity?" *Journal of Public Economics* 96.5, 474–484.
- Black, Sandra E., Paul J. Devereux, and Kjell G. Salvanes (Mar. 2005). "Why the Apple Doesn't Fall Far: Understanding Intergenerational Transmission of Human Capital". *American Economic Review* 95.1, 437–449.
- Bonhomme, Stéphane, Kerstin Holzheu, Thibaut Lamadon, Elena Manresa, Magne Mogstad, and Bradley Setzler (2023). "How Much Should we Trust Estimates of Firm Effects and Worker Sorting?" *Journal of Labor Economics* 41.2, 291–322.
- Brandén, Gunnar and Martin Nybom (2019). "Utvecklingen av intergenerationell rörlighet i Sverige". Tech. rep. Swedish Ministry of Finance.
- Burdett, Kenneth and Dale Mortensen (1998). "Wage Differentials, Employer Size, and Unemployment". *International Economic Review* 39.2, 257–73.

- Card, David, Ana Rute Cardoso, Joerg Heining, and Patrick Kline (2018). "Firms and Labor Market Inequality: Evidence and Some Theory". *Journal of Labor Economics* 36.S1, S13–S70.
- Card, David, Ana Rute Cardoso, and Patrick Kline (2016). "Bargaining, sorting, and the gender wage gap: Quantifying the impact of firms on the relative pay of women". *The Quarterly journal of economics* 131.2, 633–686.
- Card, David, Jörg Heining, and Patrick Kline (2013). "Workplace Heterogeneity and the Rise of West German Wage Inequality". *The Quarterly Journal of Economics* 128.3, 967–1015.
- Corak, Miles and Patrizio Piraino (2010). "The Intergenerational Transmission of Employers". *Journal of Labor Economics* 29.1, 37–68.
- Cornelissen, Thomas, Christian Dustmann, and Uta Schönberg (2017). "Peer effects in the workplace". *American Economic Review* 107.2, 425–56.
- Deutscher, Nathan and Bhashkar Mazumder (2023). "Measuring Intergenerational Income Mobility: A Synthesis of Approaches". *Journal of Economic Literature* 61.3, 988–1036.
- Dobbin, Caue and Tom Zohar (2023). "Quantifying the Role of Firms in Intergenerational Mobility". Unpublished manuscript.
- Eeckhout, Jan (2018). "Sorting in the Labor Market". *Annual Review of Economics* 10. Volume 10, 2018, 1–29.
- Engbom, Niklas, Christian Moser, and Jan Sauermann (2023). "Firm pay dynamics". *Journal of Econometrics* 233.2, 396–423.
- Engzell, Per and Carina Mood (2023). "Understanding Patterns and Trends in Income Mobility through Multiverse Analysis". *American Sociological Review*, forthcoming.
- Engzell, Per and Nathan Wilmers (2024). "Firms and the Intergenerational Transmission of Labor Market Advantage".
- Gelbach, Jonah B (2016). "When do covariates matter? And which ones, and how much?" *Journal of Labor Economics* 34.2, 509–543.
- Gerard, François, Lorenzo Lagos, Edson Severnini, and David Card (2021). "Assortative Matching or Exclusionary Hiring? The Impact of Employment and Pay Policies on Racial Wage Differences in Brazil". *American Economic Review* 111.10, 3418–57.
- Grönqvist, Erik, Björn Öckert, and Jonas Vlachos (2017). "The Intergenerational Transmission of Cognitive and Noncognitive Abilities". *Journal of Human Resources* 52.4, 887–918.
- Heckman, James J and Stefano Mosso (2014). "The Economics of Human Development and Social Mobility". *Annual Review of Economics* 6, 689–733.
- Hjorth-Trolle, Anders and Rasmus Landersø (2023). "The Different Sources of Intergenerational Income Mobility in High and Low Income Families". Tech. rep. Rockwool Foundation Research Report Study Paper 195.
- Jäntti, Markus and Stephen Jenkins (2014). "Income Mobility". Handbook of Income Distribution. Ed. by Anthony Atkinson and François Bourguignon. Vol. 2. Elsevier, 807–935.
- Kline, Patrick, Raffaele Saggio, and Mikkel Sølvsten (2020). "Leave-out estimation of variance components". *Econometrica* 88.5, 1859–1898.
- Kramarz, F. and O. N. Skans (Jan. 2014). "When Strong Ties are Strong: Networks and Youth Labour Market Entry". *The Review of Economic Studies* 81.3, 1164–1200.
- Kremer, Michael (1993). "The O-Ring Theory of Economic Development". *The Quarterly Journal of Economics* 108.3, 551–75.
- Lachowska, Marta, Alexandre Mas, and Stephen A Woodbury (2020). "Sources of displaced workers' long-term earnings losses". *American Economic Review* 110.10, 3231–66.
- Le Barbanchon, Thomas, Roland Rathelot, and Alexandra Roulet (Oct. 2020). "Gender Differences in Job Search: Trading off Commute against Wage*". *The Quarterly Journal of Economics* 136.1, 381–426. eprint: https://academic. oup.com/qje/article-pdf/136/1/381/35653390/qjaa033.pdf.
- Lee, Sang Yoon and Ananth Seshadri (2019). "On the intergenerational transmission of economic status". *Journal of Political Economy* 127.2, 855–921.
- Lindqvist, E. and R. Vestman (2011). "The Labor Market Returns to Cognitive and Noncognitive Ability: Evidence from the Swedish Enlistment". *American Economic Journal: Applied Economics* 3.1, 101–128.
- Lochner, Lance and Alexander Monge-Naranjo (2012). "Credit Constraints in Education". *Annual Review of Economics* 4.1, 225–256. eprint: https://doi.org/10.1146/annurev-economics-080511-110920.
- Manning, Alan (2013). "Monopsony in motion". *Monopsony in Motion*. Princeton University Press.
- Nybom, Martin (2017). "The distribution of lifetime earnings returns to college". *Journal of Labor Economics* 35.4, 903–952.
- Nybom, Martin and Jan Stuhler (2016). "Heterogeneous Income Profiles and Life-Cycle Bias in Intergenerational Mobility Estimation". *Journal of Human Resources* 51.1.
- San, Shmuel (2022). "Who Works Where and Why: The Role of Social Connections in the Labor Market". Tech. rep. Working Paper, Hebrew University of Jerusalem.
- Solon, Gary (June 1992). "Intergenerational Income Mobility in the United States". *American Economic Review* 82.3, 393–408.

- Solon, Gary (2004). "A Model of Intergenerational Mobility Variation over Time and Place". *Generational Income Mobility in North America and Europe*, 38–47.
- Song, Jae, David J Price, Fatih Guvenen, Nicholas Bloom, and Till Von Wachter (2019). "Firming up inequality". *The Quarterly Journal of Economics* 134.1, 1–50.
- Sorkin, Isaac (2018). "Ranking firms using revealed preference". *The Quarterly Journal of Economics* 133.3, 1331–1393.
- Staiger, Matthew (2022). "The Intergenerational Transmission of Employers and the Earnings of Young Workers". Tech. rep. Working Paper, Harvard University.
- Stinson, Martha and Christopher Wignall (Mar. 2018). "Fathers, Children, and the Intergenerational Transmission of Employers". Working Papers 18-12. Center for Economic Studies, U.S. Census Bureau.
- Zimmerman, Seth D (2019). "Elite colleges and upward mobility to top jobs and top incomes". *American Economic Review* 109.1, 1–47.

Appendix A: Additional figures and tables A.1 Variance decomposition of the AKM model

	AKM sample	Main sample (born 1967-77)
Age	20-64	25-41	39-41
	(1)	(2)	(3)
Variance of log earnings	0.273	0.248	0.208
Components:			
Individual FEs	0.103 (37.7%)	0.073 (29.4%)	0.074(35.6%)
Firm FEs	0.020 (7.3%)	0.023 (9.3%)	0.023(11.1%)
Covariance (sorting)	0.019 (7.0%)	0.021 (8.5%)	0.023 (11.1)
Covariates and residual	0.131 (48.0%)	0.129 (52.0%)	0.088(42.3%)
Worker obs.	7,668,377	967,417	857,064
Number of firms	341,798	228,285	118,258
Worker-year obs.	126,475,937	13,550,074	2,437,567

Table A1. Variance decomposition

Notes: The table shows a variance decomposition of log earnings into the of components of equation (1). Column (1) show the variance decomposition for the AKM sample, column (2) shows the result for the main lifecycle sample and column (3) shows the result for the main sample with mean earnings estimated for the ages 39-41.

Using our estimates from equation (1), we can decompose the variance in income as

$$Var(y_{ijt}) = Var(\alpha_i) + Var(\psi_j) + 2Cov(\alpha_i, \psi_j) + Var(X_{it}\delta) + 2Cov(X_{it}\delta, \alpha_i + \psi_j) + Var(\varepsilon_{ijt})$$
(A1)

We report the results in Appendix Table A1, separately for three samples: our AKM sample, our main intergenerational sample across the entire age span (ages 25-41), and our main sample at age 39-41. The first two terms on the right-hand side in equation (A1) describe what fraction of the overall earnings variance is due to individual and firm components, respectively. The third component measures the contribution of worker-firm sorting; if this covariance is positive, there is positive assortative matching in the sense that workers with high (permanent) unobserved productivity sort into firms with high pay premia. The last three terms capture earnings variation due to covariates and the error term.

As found by others, the most important component for explaining the variance of log earnings is the (variance of) worker effects, here at 29-38% across the three samples. On the other hand, firm fixed effects and the covariance

between firm and worker fixed effects together explain 14-22% of the total variance. When we compare the samples, we see that the variance decomposition is largely stable across samples. However, for the main sample observed over the lifecycle (column 2) we find a somewhat decreased importance of the individual component, compared to the full AKM sample (column 1). For the prime-age version of the main sample (column 3), which only includes incomes at ages 39-41, there is a slight uptick in the importance of firms and sorting (rows 2 and 3) compared to the baseline. Overall, the decomposition is very similar to Engbom et al. (2023) who use similar data and specifications, and also largely in line with evidence from the US (e.g. Song et al., 2019).

A.2 Decomposition of the IGE

Measurement error

The table shows the IGE decomposition when the AKM model have been estimated for shorter time period, for the years 2010-2015.

		Dependent variable				
	<i>Yijt</i> (1)	$\hat{\alpha}_i$ (2)	$\hat{\psi}_{j=J(i,t)}$ (3)			
Panel A: AKM estimated for the years 2010-2015						
$\mathcal{Y}_{f(i)}$	0.197***	0.165***	0.031***	0.021***		
0 ()	(0.001)	(0.001)	(0.000)	(0.000)		
\hat{lpha}_i				0.059***		
				(0.000)		
Share of IGE	1.0	0.84	0.16	0.11		
Worker obs.	784,259	784,259	784,259	784,259		
Pan	el B: AKM	estimated fo	or 1985-201	8		
(interg	enerational	sample same	e as in Panel	A)		
$\mathcal{Y}_{f(i)}$	0.197***	0.117***	0.051***	0.033***		
0 ()	(0.001)	(0.001)	(0.000)	(0.000)		
$\hat{\alpha}_i$				0.154***		
				(0.001)		
Share of IGE	1.0	0.59	0.26	0.17		
Worker obs.	784,259	784,259	784,259	784,259		

Table A2. Decomposition of the IGE, AKM estimated for the years 2010-2015

Notes: Column (1) reports the estimated slope coefficient from regression (2) of mean of log child earnings over the ages 39-41 on log father's earnings. Columns (2)-(4) report the slope coefficients from the corresponding regressions when decomposing child log earnings y_{ijt} according to equation (1) into individual fixed effects α_i and mean firm fixed effects ψ_j over the ages 39-41. The columns for time-varying control are not included since we have imputed firm values for the ages 39-41, even if these ages are outside of the 2010-2015 window if the firm existed for years 2010-2015, and thus we miss data on the time-varying controls for those observations. In Panel A, equation (1) is estimated for the years 2010-2015. In Panel B, equation (1) is estimated for the full time-period 1985-2018, but the observations in the intergenerational regression are limited to the same as in Panel A. Robust standard errors in parentheses.



Figure A1. Difference AKM 2010-2015 and long AKM

Notes: The figure shows that difference in the estimated firm premia between the AKM estimated for the long period 1985-2018, and the AKM estimated for the short period 2010-2015 against the firm premia from the AKM estimated for the long period.

		Dependent variable					
	<i>Yijt</i> (1)	$\hat{\alpha}_i$ (2)	$\hat{\psi}_{j=J(i,t)} \tag{3}$				
A: Dropping 50% of individuals before AKM estimation							
$\mathcal{Y}_{f(i)}$	0.203***	0.120***	0.053***	0.036***			
	(0.002)	(0.001)	(0.001)	(0.001)			
$\hat{\alpha}_i$				0.145***			
				(0.001)			
Share of IGE	1.0	0.59	0.26	0.18			
Worker obs.	418,980	418,980	418,980	418,980			
В	: AKM esti	mated for f	ull sample				
(interg	enerational	sample same	e as in panel	A)			
$\mathcal{Y}_{f(i)}$	0.203***	0.120***	0.053***	0.035***			
• ()	(0.002)	(0.001)	(0.001)	(0.000)			
\hat{lpha}_i				0.152***			
				(0.001)			
Share of IGE	1.0	0.59	0.26	0.17			
Worker obs	418,980	418,980	418,980	418,980			

 Table A3. Decomposition of IGE, random sub-sample

Notes: Column (1) reports the estimated slope coefficient from regression (2) of mean of log child earnings over the ages 39-41 on log father's earnings. Columns (2)-(4) report the slope coefficients from the corresponding regressions when decomposing child log earnings y_{ijt} according to equation (1) into individual fixed effects α_i , mean firm fixed effects ψ_j over the ages 39-41, and time-varying controls. In panel A equation (1) is estimated after dropping a random subsample of 50% of all individuals. In panel B equation (1) is estimated for the full sample, but the observations in the intergenerational regression are limited to the same as in panel A. Robust standard errors in parentheses.

Alternative estimation of the AKM regression

	Dependent variable				
	<i>Yijt</i> (1)	$\hat{\alpha}_i$ (2)	$\hat{\psi}_{j=J(i,t)}$ (3)	$\begin{array}{c} X_{it}\hat{\beta} + \hat{\varepsilon}_{ijt} \\ (4) \end{array}$	
	A. AKM es	timated with	n time-varyi	ng firm FE	
$y_{f(i)}$	0.200***	0.125***	0.057***	0.018***	0.040***
5 (-)	(0.001)	(0.001)	(0.000)	(0.001)	(0.000)
Share of IGE	1.0	0.63	0.29	0.09	0.20
Worker obs.	847,447	847,447	847,447	847,447	847,447
	B. AKM es	timated usin	ng establishn	nent codes	
$\mathcal{Y}_{f(i)}$	0.198***	0.105***	0.072***	0.022***	0.056***
5 (1)	(0.001)	(0.001)	(0.000)	(0.001)	(0.000)
Share of IGE	1.0	0.53	0.36	0.11	0.28
Worker obs.	831,927	831,927	831,927	831,927	831,927
C. AKM	estimated us	sing establis	hment codes	for large firm	ns $y_{f(i)}$
0.198***	0.107***	0.069***	0.023***	0.052***	5 (1)
	(0.001)	(0.001)	(0.000)	(0.001)	(0.000)
Share of IGE	1.0	0.54	0.35	0.12	0.26
Worker obs.	836,554	836,554	836,554	836,554	836,554
D. AKM estim	nated withou	t excluding	firms with fe	ew movers	
$y_{f(i)}$	0.199***	0.115***	0.058***	0.027***	0.043***
5 (1)	(0.001)	(0.001)	(0.000)	(0.001)	(0.000)
Share of IGE	1.0	0.58	0.29	0.14	0.22
Worker obs.	904,384	904,384	904,384	904,384	904,384

Table A4. Decomposition of the IGE: Alternative estimations of the AKM equation

Notes: Column (1) reports the estimated slope coefficient from regression (2) of mean of log child earnings at ages 39-41 on log father's earnings. Columns (2)-(4) report the slope coefficients from the corresponding regressions when decomposing child earnings y_{ijt} according to equation (1) into individual fixed effects α_i , mean of firm fixed effects ψ_j for ages 39-41, and time-varying controls. The different panel shows different variants of estimating equation (1). In panel A we estimate time-varying firm fixed effects by dividing the period 1985-2018 into 4 periods, and allow the firm-fixed effects to vary between the periods. In panel B we use establishment codes instead of firm codes to estimate equation (1). In Panel C establishment codes are used for large firms and firm codes for small firms with 1,000 or fewer unique workers during the analysis period. In panel D we estimate the AKM without excluding firms with few movers.

	Dependent variable					
	<i>Yijt</i> (1)	$\hat{\alpha}_i$ (2)		$\begin{array}{c} X_{it}\hat{\beta} + \hat{\varepsilon}_{ijt} \\ (4) \end{array}$	$\hat{\psi}_{j=J(i,t)}$ (5)	
	A: A	KM estimat	ed with wag	jes		
$y_{f(i)}$	0.175***	0.132***	0.022***	0.021***	0.008***	
	(0.001)	(0.001)	(0.000)	(0.000)	(0.000)	
\hat{lpha}_i					0.105***	
					(0.001)	
Share of IGE	1.0	0.75	0.13	0.12	0.05	
Worker obs.	565,231	565,231	565,231	565,231	565,231	
	B: AF	KM estimate	d with earni	ngs		
	(excluding	individuals	not in wage	sample)		
$\mathcal{Y}_{f(i)}$	0.194***	0.120***	0.044***	0.030***	0.025***	
	(0.001)	(0.001)	(0.000)	(0.001)	(0.000)	
\hat{lpha}_i					0.156***	
					(0.001)	
Share of IGE	1.0	0.62	0.23	0.15	0.13	
Worker obs.	565,231	565,231	565,231	565,231	565,231	

Table A5. Decomposition of the IGE: AKM estimated with wages

Notes: Column (1) reports the estimated slope coefficient from regression (2) of mean of log child wages at ages 39-41 on log father's earnings. Columns (2)-(4) report the slope coefficients from the corresponding regressions when decomposing child wage y_{ijt} according to equation (1) into individual fixed effects α_i , mean of firm fixed effects ψ_j for ages 39-41, and time-varying controls. In Panel A we estimate equations (1) and (2) using the wage structure sample, which covers roughly a third of private sector employees (with those in larger firms oversampled) and all public sector employees, in total corresponding to about 50% of the workforce. In Panel B we estimate equations (1) and (2) using earnings, but limiting the observations to the same observations as in the wage sample. Robust standard errors in parentheses.

Decomposition of the IGE: Heterogeneity

Table A6 shows the decomposition of the intergenerational earnings elasticity for different subsamples.

		Dependent variable				
	<i>Yijt</i> (1)	$\hat{\alpha}_i$ (2)	$\hat{\psi}_{j=J(i,t)}$ (3)	$\begin{array}{c} X_{it}\hat{\beta} + \hat{\varepsilon}_{ijt} \\ (4) \end{array}$		
		A. Sample	e: Men			
$y_{f(i)}$	0.230***	0.135***	0.058***	0.036***	0.041***	
-) ()	(0.002)	(0.001)	(0.001)	(0.001)	(0.001)	
Share of IGE	1.0	0.59	0.25	0.16	0.18	
Worker obs.	436,709	436,709	436,709	436,709	436,709	
		B. Sample:	Women			
$y_{f(i)}$	0.169***	0.097***	0.050***	0.022***	0.040***	
-) (-)	(0.002)	(0.001)	(0.000)	(0.001)	(0.000)	
Share of IGE	1.0	0.57	0.30	0.13	0.24	
Worker obs.	420,355	420,355	420,355	420,355	420,355	
C. Ex	xcluding wor	rkers who w	ork in same	e firm as fath	er	
$y_{f(i)}$	0.191***	0.112***	0.051***	0.028***	0.033***	
-) (-)	(0.001)	(0.001)	(0.000)	(0.001)	(0.000)	
Share of IGE	1.0	0.59	0.27	0.15	0.17	
Worker obs.	748,293	748,293	748,293	748,293	748,293	
	D. Ex	cluding publ	lic sector fir	rms		
$y_{f(i)}$	0.228***	0.124***	0.068***	0.035***	0.051***	
	(0.001)	(0.001)	(0.000)	(0.001)	(0.000)	
Share of IGE	1.0	0.54	0.30	0.15	0.22	
Worker obs.	549,635	549,635	549,635	549,635	549,635	
	E. Excludin	g firms with	less then 1	0 movers		
$y_{f(i)}$	0.200***	0.118***	0.052***	0.030***	0.034***	
-) ()	(0.001)	(0.001)	(0.000)	(0.001)	(0.000)	
Share of IGE	1.0	0.59	0.26	0.15	0.17	
Worker obs.	836,199	836,199	836,199	836,199	836,199	
	F. Excluding	g firms with	less then 5	0 movers		
$y_{f(i)}$	0.201***	0.121***	0.047***	0.032***	0.028***	
J (*/	(0.001)	(0.001)	(0.000)	(0.001)	(0.000)	
Share of IGE	1.0	0.60	0.23	0.16	0.14	
Worker obs.	745,473	745,473	745,473	745,473	745,473	

 Table A6. Decomposition of the IGE: Heterogeneity

Notes: Column (1) reports the estimated slope coefficient from regression (2) of mean of log child earnings at ages 39-41 on log father's earnings. Columns (2)-(4) report the slope coefficients from the corresponding regressions when decomposing child earnings y_{ijt} according to equation (1) into individual fixed effects α_i , mean of firm fixed effects ψ_j for ages 39-41, and time-varying controls. Panel A shows results for males, and panel B shows results for women. In Panel C, workers who work in the same firm as their fathers are excluded, where working in the same firms as the father is defined as having ever worked in the father's main firm (main firm is the firm the father works in for most years between 1985-2018). Panel D shows results where public sector firms are excluded, where public sector firms are defined as firms in industries with SNI92 codes 75, 80, 85, 90, 91, 92, 93 and 99. Panel E excludes firms with less than 10 movers and Panel F excludes firms with less than 50 movers (baseline: at least five movers).

A.3 Skill sorting

In this Appendix we provide additional evidence on the decomposition of the SES gradient in firm pay into skill-based sorting ("assortative matching") and residual sorting. Table A7 reports estimates from the auxiliary regressions (6a)-(6d) of each skill measure on parental income (i..e, estimates of $\phi_{cog}, \phi_{soc}, \phi_{edu}$ and ϕ_{akm}). Note that the regression coefficients are not directly comparable, as they also reflect differences in the scaling of each variable. We therefore focus on the correlation coefficient, which is equal to the square root of the R-squared reported in the table. We find that parental income correlates most strongly with child education, while the correlation with the child's social skills is lowest. Despite correlating strongly with parental income, child education contributes only a small share to the firm pay gradient (see Table (5)). The comparatively low correlation for our proxy of social skills is possibly explained by measurement error, as social skill measures tend to be more noisy than measures of cognitive skills (Grönqvist et al., 2017).

Table A8 reports estimates regression (7), showing that the coefficients change only marginally when excluding parental income from the regression.

	Dependent variable				
	Cognitive skills	Social skills	Education	\hat{lpha}_i	
$\mathcal{Y}_{f(i)}$	1.022***	0.714***	1.477***	0.129***	
	(0.007)	(0.006)	(0.009)	(0.001)	
Observations	371,411	371,411	371,411	371,411	
R-squared	0.060	0.038	0.070	0.052	

 Table A7. Skills and father's income

Notes: The figure shows result from the Gelbach decomposition, equations (6a)-(7), regressing each of the skills on father's income.

	Dependent variable		
	$\hat{\psi}_{j=J(i,t)}$	$\hat{\psi}_{j=J(i,t)}$	
$\mathcal{Y}_{f(i)}$	0.029***		
3 ()	(0.001)		
Cognitive skills	0.010***	0.011***	
	(0.000)	(0.000)	
Social skills	0.002***	0.002***	
	(0.000)	(0.000)	
Education	0.003***	0.004***	
	(0.000)	(0.000)	
\hat{lpha}_i	0.089***	0.096***	
	(0.001)	(0.001)	
Observations	371,411	371,411	
R-squared	0.086	0.080	

Table A8. Firm pay premia and skills

Notes: The figure shows result from the Gelbach decomposition, equation (5b), regressing the firm premia on father's income and each of the skills.

A.4 Non-linear firm pay gradients

Figure A2. Child income and firm premia by father's income (logs) (*a*) Child vs father log income (*b*) Firm premium vs father's log income



Notes: Figure (a) shows binned scatter plots of child's log income at age 40 by father's log income. Figure (b) shows firm fixed effects ψ_j at age 40 estimated by equation (1) and firm premia residualized on individual fixed effect by father log income.

A.5 Additional evidence on lifecycle dynamics



Figure A3. Firm earnings premium over the lifecycle using potential experience (*a*) Full sample

Notes: The figures plots the mean estimated firm premium $\hat{\psi}_j$ against potential experience over the life cycle, by quartile of father's income. Potential experience is defined as the year minus the the year the individual enters the labor market. Entering the labor market is defined as the first year, after age 20 of having higher then low earnings (where low earning is defined as 20% of the median earnings of men aged 45), and after the age for potential finishing school (defined as age- (years of education age - education +6)). Sub-figure (a) shows the result for the full sample, sub-figure (b) shows the result for children without collage education and sub-figure (c) shows the result for children with college education. Father's income quartiles are defined in the full sample.



Figure A4. Firm pay premium over the lifecycle conditional on individual fixed effects

Notes: The figures show the estimated firm premium $\hat{\psi}_i$ from equation (1), residualized on the individual fixed effects, over the life cycle and by quartile of father's income.



Figure A5. Firm premia over the life-cycle by gender

Notes: The figures show the estimated firm premium $\hat{\psi}_i$ from equation (1) over the life cycle, by quartile of father's income. Sub-figure (a) shows the results for men and (b) shows the result for women. Father's income quartiles are defined in the full sample

A.6 Firm switching



Notes: The figures show pattern for switching firms over the life-cyle by quartile of father's income. Sub-figure (a) shows the result for individuals who do not have a college education and sub-figure (b) shows the results for n individuals who have a college education.





Notes: The figure shows the probability of switching to a firm with a higher firm premium than the one before conditional on switching, by father's income quartile. Sub-figure (a) shows the result for individuals who do not have a college education and sub-figure (b) shows the results for individuals who have a college education.



Figure A8. Mean change in firm premium among switchers *(a)* Non-college *(b)* College

Notes: The figure shows the difference between the new firm premia and the firm premia before for individuals who switch firms. Sub-figure (a) shows the result for individuals who do not have a college education and sub-figure (b) shows the results for individuals who have a college education.

Figure A9. Change in firm FE for voluntary and involuntary switches *(a)* Voluntary switches only *(b)* Involuntary switches only



Notes: The figure shows the difference between the new firm premia and the firm premia before for individuals who switch firms, by father's income quartile. Sub-figure (a) shows the results for voluntary switches, defined as a switch without any unemployment insurance or without any year with zero income. Sub-figure (b) shows the result for involuntary switches, defined as a switch with either unemployment insurance or a year of zero income between working at the old firm and starting at the new firm.

A.7 Working patterns

In this section, we provide additional evidence on how working and commuting patterns vary by parental background. Figure A10 plots the share of individuals who work in the same firm as their father, separately for those





Notes: The figure shows the proportion of children who work in the father's main firm at different ages, separately by quartile of fathers' income. The father's main firm is defined as the firm the father works in for the most years between the years 1985-2018. Figure (a) include children without college education and figure (b) include children with college education.

with and without a college degree. Figure A11 plots the share of individuals who commute (i.e., work and reside in different municipalities), separately for those with and without a college degree. Children from high-income parents are more likely to commute, even within education group. Finally, Table A9 provides event-study type regression results conditional on individual fixed effects, showing how firm pay changes for individuals who begin to commute. Also when we consider such within-individual variation, the firm premium increases when individuals start to commute. As shown in column 1, commuting raises firm premia by about 1 pp. The pay benefit of commuting grows somewhat with age (column 2). Finally, column 3 shows that individuals from different parental backgrounds in terms of father's income benefit similarly from commuting. Thus, we conclude that firm pay is positively related to commuting and in a similar manner irrespective of parental background, but that high-SES children are more likely to commute.



Notes: The figure shows the proportion of individuals who commute (i.e., work in another municipality than they live in) over the life-cycle, by quartile of father's income. Sub-figure (a) includes children without college education and sub-figure (b) children with college education.

	Dependent variable: $\hat{\psi}_{i=J(i,t)}$		
	(1)	(2)	(3)
Commuting indicator	0.010***	0.017***	0.011***
	(0.000)	(0.000)	(0.000)
Commuting # 2nd father's income quartile			-0.001**
			(0.000)
Commuting # 3rd father's income quartile			0.000
			(0.000)
Commuting # 4th father's income quartile			-0.000
			(0.000)
Commuting # age normalized at 40		0.001***	
		(0.000)	
Age controls	Yes	Yes	Yes
Individual fixed effects	Yes	Yes	Yes

Table A9. Commuting

Notes: The table shows results with the firm premia as the dependent variable. The regressions include a dummy variable for working in a municipality other than living in and individual fixed effects. Columns 1 and 3 include flexible age controls (age, age squared, age interaction with father income quartile, and age squared interacted with father income quartile). Column (2) includes linear age dummies normalized at age 40 to ease interpretation and includes dummies for working in other municipalities than living in interacted with age. Column (3), includes dummies for working in other municipalities than living in interacted with the father income quartile.

A.8 Return to experience in firm classes



Notes: The figures show firm fixed effects and experience estimated with regression (3). Figure (a) shows the relationship between firm fixed effects and returns to experience in the firm class. Figure (b) shows the relationship between the change in firm fixed effects for individuals who change firms, and the return to experience in the firm class the individuals switch from.

Essay 3: Do sibling correlations in skills, schooling, and earnings vary by socioeconomic background? Insights from Sweden

Co-authored with Akib Khan and Olof Rosenqvist

Acknowledgements: We thank Adrian Adermon, Georg Graetz, Raoul van Maarseveen, Martin Nybom, and seminar participants at IFAU and Uppsala University for valuable comments.

1 Introduction

Family background can shape an individual's life trajectory. There are strong correlations in socioeconomic outcomes between parents and children, and between siblings (e.g., Solon et al., 1991; Solon, 1999; Björklund and Jäntti, 2009; Björklund and Jäntti, 2020; Chetty et al., 2014; Behrman and Taubman, 1989; Vosters and Nybom, 2017; Adermon et al., 2021). But is life more formed by family background in certain groups than in others? Cross-country heterogeneity in the importance of family background is well documented. For example, sibling correlations – a comprehensive measure of family influence – in education is higher in developing countries (e.g., Dahan and Gaviria, 2001; Ahsan et al., 2022).¹ But there is also variation among industrialized countries. Nordic countries, for instance, are often documented to have relatively low sibling correlations in earnings and educational attainment (e.g., Björklund et al., 2009; Björklund and Jäntti, 2020). However, little is known about heterogeneities across the socioeconomic spectrum within a country.

Two societies where the importance of family background as a whole is similar can be different in terms of the distribution of this importance across social groups within a society.² In one society, sibling differences in socioe-conomic outcomes may primarily arise in families with high socioeconomic status (SES), whereas the other society can display relatively constant sibling differences across groups with different SES. Although there might be disagreement over whether one society is preferred to the other, documenting the relationship between sibling similarity and parental SES arguably offers a richer picture of the equality of opportunity in a society.

An individual's (perceived) options in life, and thereby the possibilities of differentiating him/herself from their siblings, are potentially affected by the family's resources, preferences, and expectations, all of which may vary by SES. There is a theoretical discussion linking a potential socioeconomic gradient in sibling similarity to parental strategies for investing in their children. While several different mechanisms could give rise to differences in sibling similarity by parental SES,³ parents' decisions on whether to reinforce, or compensate for, initial ability differences between siblings have been empha-

¹Björklund et al. (2010) describe the sibling correlation as: "an omnibus measure of the importance of family background and community effects. It includes anything shared by siblings: parental income and parental influences such as aspirations and cultural inheritance, as well as things not directly experienced in the home, such as school, church and neighborhood effects" (p. 4).

²Just as the importance of family background can be different in two societies with the same cross-sectional inequality (see discussion in Solon, 1999)

³For instance, children of low SES parents might attend different schools than children of high SES parents and the schools might be different in terms of how they benefit children with different endowments. In addition, the degree of complementarity between child endowments and parental inputs could also matter for the relationship between parental SES and sibling similarity.

sized in the literature (Becker and Tomes, 1976; Griliches, 1979; Behrman et al., 1982). Griliches (1979), for instance, argues that parents want to compensate for initial ability differences and that high SES parents have more resources to make such investments.⁴ Smaller sibling differences in high SES families would result in higher sibling correlations in these groups. Given this long-standing theoretical interest, the question has received surprisingly little empirical attention.

A key empirical challenge in answering this question is the availability of data on both generations and on a scale that would allow constructing granular groups of differing socioeconomic affluence. There is a small but growing set of studies on this question, mainly based on data from the U.S., Germany, and Sweden. Collectively, the evidence these papers offer on social gradients in sibling similarity is inconclusive.⁵ Many of these papers use relatively small survey-based samples, while others construct broad SES groups. Both of these approaches, however, may limit the possibility to detect gradients and non-linearities in the relation between sibling similarity and parental SES.

In this paper, using population-wide Swedish register data spanning multiple generations, we provide one of the most comprehensive examinations yet on sibling similarity across the socioeconomic spectrum. In particular, we compare sibling correlations in skills, schooling, and earnings across finegrained groups defined by parental education and earnings.⁶ The register data contain information on mid-life earnings and years of schooling for the parent generation (born between 1940–1950) as well as the child generation (born between 1965–1982). For men in the child generation, the dataset also includes measures of cognitive and non-cognitive skills from the military conscription assessments made at age 18–19.⁷

In contrast to existing work, the register data allows us to construct finegrained groups by parental education and earnings. It is important to use finegrained groups rather than, e.g., divisions by the median, because the degree of homogeneity in the different groups matters for the comparison of the sibling correlations. For example, the variation in years of schooling and income is typically much higher above the median than below. Consequently, families in the group below the median will be more homogeneous than families above the median. Since the sibling correlation is the ratio of the between-family

⁴He writes, "Thus, I would expect that the within-family variance in socioeconomic achievement would decline at higher income levels" (Griliches, 1979, p. S62).

⁵Please see, for the US: Conley and Glauber (2007), Conley and Glauber (2008), and Conley et al. (2007); for Germany: Anger and Schnitzlein (2017), Grätz (2018), and Baier (2019); and for Sweden: Grätz et al. (2021), Grätz and Kolk (2022), and Hällsten and Thaning (2022).

⁶In the paper, we sometimes use the term income rather than earnings, but in both cases it is labor income we refer to.

⁷The ability measures from the military conscription have previously been used by, e.g., Lindqvist and Vestman (2011), Edin et al. (2022), and Grätz et al. (2021). Lindqvist and Vestman (2011) and Edin et al. (2022) have shown that these measures are strong predictors of labor market performance.

variation to the total variation, this will lead to a lower sibling correlation in the below-median group even if the within-family variation is the same across the two groups (Solon et al., 1991). Since we, in this paper, are not interested in the between-family variation per se, we want to avoid this. Lastly, a finer division is better equipped to detect gradients and non-linearities in the data.⁸

A more granular division tends to equalize the between-family variation in the different groups so that any differences in sibling correlations across groups are more likely to reflect differences in within-family variation, which arguably is of higher interest and more related to the parental investment theories discussed above. Hence, we also directly compare this within-family variation across groups as suggested by Breen and Ermisch (2021). An alternative approach would be to only focus on the within-family variation. However, the measure is then never put in relation to the overall variation, which could be misleading. Therefore, we generally show both the sibling correlation and the underlying variance components, paying particular attention to differences in sibling correlations across social groups that are driven by differences in the within-family variance.

Our results show a clear and consistent pattern. Sibling correlations generally decrease with parental SES. For years of schooling, mid-life earnings and cognitive skills, sibling correlations decline almost monotonically with both the education and earnings of the parents. In comparison, sibling correlations in non-cognitive skills only decrease with parental earnings.

The socioeconomic gradient we find is substantive. Moving from the 5th to the 15th ventile of parental income, for instance, sibling correlations in income and cognitive skills decline by over 20% and 7%, respectively. The corresponding relative decline for the sibling correlation in non-cognitive skills is 14%.

The decline in sibling correlations for education and income with parental SES is mainly driven by an increase in within-family variation, indicating that siblings are more similar to each other in families with low SES (i.e., contrary to the prediction by Griliches, 1979). In contrast, the decrease in sibling correlations in skills is driven by a relative decrease in between-family variation.

Our main results are based on male siblings as we lack data on skills for women. However, reassuringly, the pattern for sibling correlations in years of schooling and earnings remains very similar when women are included in the sample. Lastly, we show that these patterns are robust to measuring income at different ages, and are not driven by differences in family structure (number of siblings and age differences between them) across the socioeconomic spectrum.

⁸Figure A3–Figure A6 in the Appendix substantiate this point, presenting results in an ascending order of granularity. Furthermore, our replication of Hällsten and Thaning (2022) in Section 7.1 offers a concrete example.

We contribute to three strands of the literature. First, we add to a small set of papers documenting variations in sibling similarity across different demographic groups.⁹ As mentioned before, these papers often rely on small samples or create broad groups that might not be sufficient to detect meaningful inter-group differences in sibling correlations. Using population-wide administrative data covering multiple generations in Sweden, which enables us to create granular subsets of the population with varying SES, we provide one of the first pieces of evidence of a robust negative relationship between parental SES and sibling correlations in skills, earnings and education.¹⁰

It is particularly interesting to discuss the findings in Grätz et al. (2021), Grätz and Kolk (2022), and Hällsten and Thaning (2022) since they also use Swedish register data. Grätz et al. (2021) focus on cognitive ability, school grades, and educational attainment, finding higher sibling correlations among low SES families when defined by parental occupation, while results are less consistent when defined by parental education. Grätz and Kolk (2022) find similar sibling correlations in total earnings over the ages 18–60 across three SES groups based on parental occupation. Lastly, Hällsten and Thaning (2022) use data on education, occupation, income and wealth for both the child and parent generation and estimate sibling correlations for each outcome in groups defined by parental SES quintiles. They generally find higher correlations in high SES families, particularly when defined by wealth. We replicate some of Hällsten and Thaning's findings and find suggestive evidence that these patterns are sensitive to variations in the granularity with which we measure SES.

The evidence we generate also relates to the theoretical literature on the role of parental inputs in inter-sibling differences in human capital formation and earnings. Parents might be averse to inequality and attempt to compen-

⁹The results in these papers are mixed with both positive, negative and non-existent associations between sibling similarity and parental SES being reported. To some extent, this can perhaps be explained by the fact that different outcomes and SES definitions are used by different studies. But even within countries, and for similar outcomes, it is hard to discern consistent patterns. In Germany, e.g., Baier (2019) finds that the sibling correlation in cognitive ability is higher in low SES families than in high SES families while Grätz (2018), who studies cognitive ability and educational attainment, finds no differences by parental SES. In addition, Anger and Schnitzlein (2017) find that high SES families display a higher sibling correlation in non-cognitive ability. There are also indications that the relation between sibling similarity and parental SES can be non-linear. Karhula et al. (2019), studying education and labor market outcomes in Finland, find a U-shaped pattern between sibling similarity and parental SES.

¹⁰Since papers estimating intergenerational parent-child correlations in socioeconomic outcomes typically find stronger associations among high SES families (e.g., Acciari et al., 2022; Bratberg et al., 2017), our finding of an opposite-signed relationship with respect to sibling correlations might seem somewhat surprising. However, sibling correlations and parent-child associations have different interpretations, with the latter being a narrower measure of family influence. In fact, Solon (1999) derives: Sibling Correlation = $(Child - ParentCorrelation)^2$ + other shared factors that are orthogonal to the parental variable. Using this relationship, Björklund and Jäntti (2020), for instance, show that these "other shared factors" are substantially more important in their relative contribution to the sibling correlation.

sate for ability differences across their children. Efficiency considerations, on the other hand, might lead them to reinforce these ability differences. The magnitude of such reinforcing investments, however, would depend on the resources a family has to begin with. Our results are therefore consistent with a scenario where parents generally reinforce ability differences across their children, with higher SES parents having greater capacity to do so more effectively. However, we acknowledge that the observed socioeconomic gradient in sibling similarity could arise even under compensating parental investments if there are sufficiently strong complementarities between parental investments and child ability. Still, this scenario arguably relies on stronger assumptions than the scenario with reinforcing investments.

Finally, we add to a large body of literature that uses the sibling correlation as a measure of the importance of family background in shaping children's socioeconomic outcomes. This strand of literature often uses this measure to compare different countries or broad groups within a country (e.g., males vs females; Björklund and Jäntti, 2020). Our paper undertakes an in-depth study of how sibling correlations vary across the socioeconomic spectrum within a country, thereby offering a more nuanced portrayal of the country's opportunity landscape.

The paper is organized as follows: In section 2, we discuss theories and previous empirical evidence concerned with parental investment strategies and child-rearing principles and discuss what they imply in terms of sibling differences across social groups. Section 3 describes the data whereas Section 4 outlines the empirical approach. Results are presented in Sections 5 and 6. Section 7 provides a discussion of the results and section 8 concludes.

2 Conceptual framework

The theoretical literature on a potential socioeconomic gradient in sibling similarity goes back to the 1970s. In the model of Becker and Tomes (1976), parents have incentives to reinforce initial ability differences between siblings for efficiency reasons. Siblings with poorer endowments are then compensated later in life with monetary transfers. Since parents with high SES have more resources they can make larger reinforcing investments. Thus, according to this theory, sibling differences in human capital and earnings would tend to be larger in more well-off families (and sibling correlations thereby lower).

On the other hand, Griliches (1979) and Behrman et al. (1982) suggest that parents have equity concerns, not only regarding consumption levels of the siblings, but also regarding the human capital levels. A key conclusion is that parents may make compensating investments even if it's more efficient to reinforce. Therefore, sibling differences would instead be smaller, and sibling correlations higher, in families with high SES if the influence from the equity concern dominates.¹¹

Clearly, the direction of the models' predicted socioeconomic gradient in sibling similarity depends on whether parents make reinforcing or compensatory investments. This question has received substantial empirical attention (e.g., Fan and Porter, 2020; Savelyev et al., 2022; Yi et al., 2015; Frijters et al., 2013; Hsin, 2012; Restrepo, 2016; Grätz and Torche, 2016; Rosenzweig and Wolpin, 1988; Rosenzweig and Zhang, 2009; Behrman et al., 1994). The evidence from these studies is mixed, but reinforcing investments are slightly more common than compensating investments (see, e.g., the review and discussion in Almond and Mazumder, 2013).

Importantly, parental responses to endowment differences between siblings may vary by SES. Hsin (2012), Restrepo (2016), and Grätz and Torche (2016) study this question. Despite the fact that all studies analyze US data and use birth weight as an indicator of the child's endowment, they reach different conclusions. Hsin (2012) and Restrepo (2016) find that low-educated mothers spend more time with children with a higher birth weight (i.e., reinforce) whereas high-educated mothers spend more time with children with a lower birth weight (i.e., compensate).

In contrast, Grätz and Torche (2016) generally find small parental responses to differences in birth weight that do not vary by SES. In addition, Grätz and Torche (2016), using a measure of cognitive ability at age 4, find that parents with high SES provide more cognitive stimulation to higher-ability children, while parents with low SES do not react to ability differences. Thus, the socioeconomic gradient in parental responses to endowment differences between siblings appears to be complex with varying patterns across different types of child endowments. Consequently, this research gives little guidance on what to expect for the relationship between sibling differences in socioeconomic outcomes and family SES.

Baier (2019) introduces another perspective, suggesting that sibling outcome differences might be larger in high SES families due to child-rearing strategies. High SES parents are better equipped to supply more child-specific inputs, adjusting interactions based on individual talents and interests (e.g., play music with a child that displays musical talent/interest and do math with a child that displays mathematical talent/interest). In contrast, low SES par-

¹¹It should also be noted that the nature of marginal returns to investments in children can play a role for the predictions. Arguably, parents with higher SES generally invest more in the children and so potential differential investments in the children occur at a higher baseline level compared to parents with lower SES. If the return to investments is decreasing, which is often assumed, a given differential investment will matter less for outcomes if the baseline investment level is higher. With this perspective, it could be argued that it actually is harder for high SES parents to compensate or reinforce. However, Becker et al. (2018) challenge the assumption of decreasing marginal returns to investments in children and argue that the returns might actually be increasing instead.

ents might provide more generic inputs to all their children. This hypothesis suggests that sibling outcome differences in high SES families are expected to be larger than in low SES families.

A couple of recent papers highlight a weaker association between potential educational attainment (measured by a polygenic score) and actual educational attainment for individuals from low SES families.¹² This has been shown in the US Papageorge and Thom (2020) as well as in the welfare state of Denmark (Ronda et al., 2022). Besides suggesting that there is wasted potential in society, these findings indicate that sibling differences in educational attainment can be lower in low SES families as endowment differences are carried through to a lesser extent to differences in actual educational attainment. These results align with Becker and Tomes (1976), suggesting low SES parents may lack resources for reinforcing investments.

Other papers explore parental responses to variations in public investments in the children (Das et al., 2013; Pop-Eleches and Urquiola, 2013; Fredriksson et al., 2016). Fredriksson et al. (2016) find that high-income parents of children in larger classes help more with homework whereas no corresponding pattern is found for low-income parents. Such substitutability between public and private investments in children from high SES families suggests that sibling outcome similarity may increase with family SES.

In conclusion, the relationship between sibling similarity in socioeconomic outcomes and family socioeconomic status (SES) is multifaceted. Early models suggest that parental reinforcement of initial ability differences could lead to larger disparities in well-off families, while others propose that equity concerns may drive compensatory investments, potentially reducing differences in high SES families. Empirical evidence presents a mixed picture, with varied parental responses to endowment differences by SES. Moreover, recent findings suggest that educational attainment differences may be lower in low SES families due to constrained resources, echoing earlier insights. Overall, the intricate interplay between parental investment strategies, socioeconomic context, and public policies underscores the complexity of this relationship.

3 Data

We combine multiple Swedish registers covering the population. These registers have been compiled by Statistics Sweden and include pseudonymized personal identifiers, making it possible to collate information from different registers. A key data source is the multigenerational register which links children with parents for children born in the period 1932–2019. This information lets us identify siblings in the data. In our dataset, a row corresponds to a child,

¹²Some specific genetic variants are highly associated with educational attainment. The polygenic score for educational attainment is obtained by combining these known associations with individual level DNA information.

and to this unit of observation we add data on skills, schooling, and earnings pertaining to the child and their parents.

For skills, we observe results on cognitive and non-cognitive ability tests taken during the military conscription at age 18–19. The test of cognitive ability, which is a written test, is similar to a standard intelligence test (Carlstedt, 2000; Carlsson et al., 2015), whereas non-cognitive ability is assessed by a trained psychologist on the basis of an interview with the conscript (Mood et al., 2012). This information is only available for men and covers cohorts born between 1950–1980. The test results are standardized within test cohorts.

Data on years of schooling come from the LISA register, which includes all individuals aged 16–65 residing in Sweden in a given year. This register is available since 1985 and contains annually updated information on the highest level of education. Annual income data is also available from 1985 and we use this data to average income over key mid-life years for both children and parents.

For children, we approximate life-time earnings with the average annual earnings for the ages 35 to 37 years.¹³ We then perform a percentile rank of the income measure by birth year. Thus, the sibling correlations in income are measured as rank-rank correlations. Education is measured at age 30. If education data is missing at age 30, we use the next available observation after age 30.

To rank parents along the socioeconomic spectrum, we use two different measures of socioeconomic status: income and education. For parents, we define individual life-time income as the average income for the ages 45 to 50 years and calculate parental income as the total of maternal and paternal incomes. We then divide these parental incomes into ventiles by the average birth year of the two parents.¹⁴ Similarly, for education, we take the average of the two parents' years of education and divide them into ventiles by the average birth year of the parents.

The sample is limited to parents born between 1940 and 1950, and children born between 1965 and 1982. The appendix includes results when only doing the cohort restrictions for children, extending the parent cohorts by using income from 1968¹⁵ (Figure A12 and Figure A13).¹⁶ Skills data are only available for men. The results in the main body of the paper are therefore limited to only include male siblings to keep the sample harmonized across outcomes. However, the appendix includes results where both men and women

 $^{^{13}}$ Our conclusions remain similar if we instead use average earnings between ages 25–30 (see Figure A11 in the appendix).

¹⁴A ventile represents 5% of the underlying population.

¹⁵We observe earnings data for the years 1968, 1970, 1971, 1973, 1975, 1976, 1979, 1980, 1982 and annually for 1985–2019.

¹⁶The results are not sensitive to these parent cohort variations.

are included (Figure A1 and Figure A2). Table 1 below describes our analysis samples and shows summary statistics for the outcomes. ¹⁷

	1	Male sample	e	Male a	nd female sa	ample
Outcomes	Children	Families	Mean	Children	Families	Mean
	(N)	(N)		(N)	(N)	
Income	441,644	322,943	319,924	858,061	460,594	266,948
Education	410,782	304,204	12.7	798,383	438,422	12.9
Cognitive ability	357,435	273,334	0.061			
Noncognitive ability	357,435	273,334	0.068			

 Table 1. Summary Statistics of Sample

Notes: This table describes our main samples. Education represents years of schooling. Income represents average annual labor income during ages 35–37 in 2018 SEK. Ability measures are standardized by test year.

4 Estimation

We compare sibling differences in socioeconomic outcomes across groups by estimating sibling correlations separately by outcome and group. Following the literature (see, e.g., Björklund et al., 2010), we estimate mixed-effect models of the type specified in Equation (1) below.¹⁸ The subscripts *i* and *j* denote family and child respectively, and the control vector includes gender (where applicable) and birth-year dummies.

$$y_{ij} = x_{ij}\beta + a_i + b_{ij} \tag{2}$$

From this model, we obtain estimates of the within-family variation (σ_b^2) and the between-family variation (σ_a^2) . We can then estimate the sibling correlation as follows:

$$\rho = \frac{\sigma_a^2}{\sigma_a^2 + \sigma_b^2} \tag{3}$$

Thus, the sibling correlation amounts to the share of total variation that is due to between-family variation. Alternatively, the sibling correlation can be thought of as the correlation in the outcome between randomly drawn pairs of siblings. Note, however, that all families, no matter the number of siblings, are included in the analysis sample.¹⁹ Since the sibling correlation depends on both the between-family variation and the within-family variation in the

¹⁷Almost 50% of the children in the male sample are the only child in their family within the sample. The main results are virtually unchanged when we drop these singletons from the estimation sample.

¹⁸We use restricted maximum likelihood estimation (reml). The analyses are performed using the mixed command in Stata.

¹⁹All observations are given the same weight, i.e., we don't let weights vary by family size.

studied outcome, it is crucial to understand the relative importance of the two components. Throughout the paper, we therefore present estimates of the variance components alongside estimates of the sibling correlations.

To investigate how the sibling correlations in skills, earnings and education vary by parental SES, we do separate estimations for each ventile of parents' SES, where parents' SES is defined by either education or income.

5 Main results: sibling correlations by parental earnings

This section presents estimates of sibling correlations, stratified by ventiles of parents' earnings, for the following outcomes: income, education, cognitive skills and non-cognitive skills.

5.1 Sibling correlations in income and education by parental income



Figure 1. Sibling correlations in income and education by parental income ventiles

Notes: The figure shows sibling correlations in earnings and years of schooling for male siblings by parental income ventiles. The green line shows a second-order polynomial fitted line for the sibling correlations.

We start by documenting sibling correlations in income and years of schooling by ventiles of parental income (Figure 1). The left and right panels show sibling correlations in income and education, respectively. Both figures show a decline in sibling correlation with parents' income.²⁰ In other words, siblings in families with low parental income are more similar to each other in terms of both income and education compared to siblings in families with parents who earn more. The decline is particularly salient for income in both absolute and relative terms. For instance, moving from the 1st to the 20th ventile of parental income, the sibling correlation in income declines by more than 0.3 (or by over 70%; please see Table 2).²¹ The corresponding decline for education is 0.12 (30%). In both cases, the differences are statistically significant at the 1% level. Figure 1 includes male siblings only, but the pattern holds when we expand the sample to comprise both men and women (results are reported in Figure A1).

What drives this decline in sibling correlations with parents' socioeconomic status: within-family (σ_b^2) or between-family (σ_a^2) variation? Ceteris paribus, higher within-family variation leads to lower sibling correlation whereas the correlation increases monotonically with between-family variation. Both for earnings and education, Figure 1 shows an increase in within-family variation with parental income, especially on the right tail of the parental income distribution (i.e. contrary to the prediction by Griliches, 1979). This indicates that the decrease in sibling correlations by parental SES is mainly driven by an increase in within-family variation. The results look similar when the sample is limited to families with two children with an age gap of four years or less (see Figure A7), suggesting that the decline is not driven by differences in family structure by SES.

5.2 Sibling correlations in cognitive and non-cognitive skills by parental income

Figure 2 reports sibling correlations in cognitive and non-cognitive skills by parental income. These skills can complement years of schooling as a measure of human capital investment. In particular, while compulsory education policies might place a lower bound on completed years of schooling, this might not be the case for these measured skills. However, these are measured at an earlier age, namely when the individuals are 18–19 years old, whereas completed years of schooling is measured at age 30 or later.

²⁰Figure A3 and Figure A4 show that the negative relation between sibling correlations and parental income is more pronounced with a more granular division of parental quantiles. This is particularly true for education.

²¹The results for the first ventile should be interpreted with some caution as some parents with very low (or zero) labor income might in practice be well-off individuals who receive income via capital investments rather than labor. The first ventile could therefore potentially be more diverse than the other ventiles. We do also see slightly higher between-family variation in this group (see Figures 1c and 1d). Still, the very high sibling correlation in income in ventile 1 (Figure 1a) is primarily driven by a markedly lower within-family variation (Figure 1c).



Figure 2. Sibling correlations in cognitive and non-cognitive skills by parental income

Notes: The figure shows sibling correlations in cognitive and non-cognitive ability for male siblings by parental income ventiles. The green line shows a second-order polynomial fitted line for the sibling correlations.

Consistent with the results for income and education, sibling similarity in both cognitive and non-cognitive skills tends to decline with parental income.²² ²³ However, in contrast to the result for education and income, this decline in sibling correlations seems to be driven by a decline in between-family variation rather than an increase in within-family variation.

6 Robustness: sibling correlations by parental education

This section replicates the analyses above by using parental education - instead of income - to define SES. Note, however, that, compared to parental income, education is measured more coarsely in our dataset. In addition, as mentioned before, compulsory education policies constrain the variation in this measure, especially on the left tail. For these reasons, we view parental

²²Again, while the negative relation is always there, it is less clear when parents are grouped by the median income, or by quintiles or deciles (Figure A5 and Figure A6).

 $^{^{23}}$ The results look similar when the sample is limited to families with two children with an age gap of four years or less (see Figure A8 in the appendix).

Outcome	Income	Education	Cognitive skills	Non-cognitive skills			
Panel A. Comparison between parental income ventiles 5 and 15							
Ventile 5	0.151	0.353	0.433	0.316			
Ventile 15	0.119	0.348	0.399	0.272			
Difference	0.032*	0.005	0.034*	0.044**			
Panel B. Comparison between parental income ventiles 1 and 20							
Ventile 1	0.468	0.416	0.484	0.353			
Ventile 20	0.119	0.292	0.372	0.304			
Difference	0.349***	0.124***	0.112***	0.049***			

Table 2. Differences in sibling correlations between different parental income ventiles

Notes: This table presents sibling correlations estimated for parental income ventiles 1, 5, 15, and 20, and the difference between these correlations for ventiles 5 and 15 and 1 and 20, respectively. *** p<0.01, ** p<0.05, * p<0.1

income as our main measure of parental SES, while parental education constitutes an alternative measure.

6.1 Sibling correlations in income and education by parental education

Figure 3 shows how sibling correlations in income and education vary with parental education. Consistent with previous results, we see a decline in sibling correlation in education at higher ventiles, driven by rising within-family variations. However, the corresponding pattern for income looks weaker. Similar results are found when female siblings are included (Figure A2).

Although the pattern of decreasing sibling correlations by parents' SES holds whether this status is measured in terms of parental education or income, some differences exist. When parents' SES is defined by education, the decline in sibling correlations is steeper at the end of the distribution (particularly for education). In contrast, when parents' SES is measured as income, the decrease in sibling correlations is more pronounced on the left tail of the parental income distribution (particularly for income).



Figure 3. Sibling correlation in income and education by parental education

Notes: The figure shows sibling correlations in income and years of schooling for male siblings by parental education ventiles. The green line shows a second-order polynomial fitted line for the sibling correlations.

6.2 Sibling correlations in skills by parental education

Figure 4 replicates Figure 3 for sibling correlations in cognitive and noncognitive skills. The negative gradient we found earlier for cognitive skills in Figure 2 appears to hold. However, the decline is less salient for the correlations in non-cognitive skills. One reason is that the correlation goes up sharply at ventile 20. Also note that the negative relation between the sibling correlation in cognitive skills and parental education is driven by a decrease in between-family variation at the right tail of the SES distribution rather than an increase in within-family variation. For non-cognitive skills, on the other hand, we find a somewhat positive relation between within-family variation and parental schooling ventile.



Figure 4. Sibling correlations in skills by parental education

Notes: The figure shows sibling correlations in cognitive and non-cognitive ability for men by parental education ventiles. The green line shows a second-order polynomial fitted line for the sibling correlations.

7 Discussion

7.1 Comparison with previous estimates

How do our findings compare to those in the existing studies, particularly those using similar data from Sweden? The closest paper of interest is Hällsten and Thaning (2022), especially their results on sibling correlations in education by parental education, which we can attempt to replicate with similar data.²⁴ Note, however, that, in the absence of a replication package, we cannot ensure that our analysis is technically identical to theirs.²⁵

Figure A9 is an attempt to replicate their analysis of sibling similarities in education across parental education quintiles. We find a somewhat positive gradient similar to their estimates. As we increase the granularity of the parental education groups by creating twenty ventiles, this positive slope disappears (see Figure A10). Instead, consistent with our results (see panel b of

²⁴This is what we study in panel b of Figure A3

²⁵In similarity with Hällsten and Thaning (2022) we limit the sample to children born between the years 1945–1976 and parents born between the years 1930–1939, and measure children's education in ranks. To measure parents' socioeconomic status, we follow Hällsten and Thaning (2022) and define their status on the basis of their education, and divide education into five quintiles.
Figure 3), we see a marked decline in sibling correlations at the right tail of the parental education distribution. This exercise further strengthens the argument that creating granular SES groups is critical to discerning how sibling similarity might vary with family background.

7.2 Implications for theories about parental investments

Our results relate to the theoretical literature on the role of parental inputs in inter-sibling differences in human capital formation and earnings (Becker and Tomes, 1976; Griliches, 1979; Behrman et al., 1982). Parents might be averse to inequality and attempt to compensate for ability differences across their children. Efficiency considerations, on the other hand, might lead them to reinforce these ability differences. High SES parents have more resources to make compensating or reinforcing investments, and thus we expect smaller sibling differences (and larger sibling correlations) in high SES families if the inequality concern dominates and larger sibling differences (and smaller sibling correlations) in high SES families if the efficiency concern dominates.

Since we generally find smaller sibling correlations in high SES families, the results are consistent with the theory that parents reinforce endowment differences between siblings. But this is of course very indirect evidence and does not prove that parents generally reinforce. The pattern we observe can arise for other reasons. In fact, there could be smaller sibling correlations in high SES families even if parents generally make compensating investments if there are strong complementarities between parental inputs and child endowments.

Still, on balance, we believe that our results are more consistent with reinforcing parental inputs than compensating. Interestingly, that interpretation aligns well with the conclusion in Almond and Mazumder (2013) that reinforcing investments are slightly more common than compensating investments. It has proven difficult to directly document reinforcing or compensating behavior in a compelling way and the results in this literature are mixed. The type of indirect evidence that we contribute with here can therefore complement earlier studies.

8 Conclusion

A person's (perceived) possibilities in life, and consequently, the opportunities to distinguish themselves from siblings, can be influenced by the family's resources, preferences, and expectations. These aspects may vary depending on the family's socioeconomic status (SES). Hence, to understand the opportunity landscape in a society, it is critical to assess, for instance, if life is more formed by individual endowments and considerations in families with better access to resources. In other words, are outcome differences between siblings greater in high SES families than in low SES families?

In this paper, we have provided one of the most comprehensive examinations yet on sibling similarity in skills, schooling and earnings across granular groups defined by parental SES. Measuring sibling similarity in terms of sibling correlations, the results show a remarkably consistent pattern. Sibling correlations generally decrease in the SES of the parents. For years of schooling, mid-life earnings and cognitive ability, we see that sibling correlations decrease in both the education and income of the parents, while sibling correlations in non-cognitive ability only decrease in the income of parents.

The main results are based on male siblings since we lack data on skills for women, but the pattern for sibling correlations in years of schooling and earnings remains very similar when also women are included in the sample. Our results are to some extent different compared to earlier studies from Sweden, particularly Hällsten and Thaning (2022). We believe that differences in cohorts, exact outcome measures and SES granularity can explain at least parts of these disparities. Importantly, we are able to replicate their results on sibling correlations in education by education of the parents. We further show that the positive relation between the sibling correlation and parental SES documented in Hällsten and Thaning (2022) becomes more similar to our results (i.e. more negative) when the same granularity of SES groups as in our paper is used.

Since the sibling correlation depends on both the between-family variation and the within-family variation in the studied outcome, it is crucial to understand the relative importance of these two components for the observed social gradient in sibling outcome similarity. Ceteris paribus, higher within-family variation leads to lower sibling correlation whereas the correlation increases monotonically with between-family variation. We show that the decline in sibling correlations of income and education is driven by an increase in withinfamily variation by parental SES. In contrast, the decline in sibling correlations of skills is driven by a decline in between-family variation. Thus, the interpretation of the skills results is not straightforward as they do not consistently point toward lower sibling similarity in the upper part of the socioeconomic spectrum. It is interesting to note that within-family variation in education and earnings is increasing in parental SES while the relation is relatively flat for skills. This means that it is not larger within-family differences in skills that give rise to the larger differences in education and earnings among the high SES families. A potential explanation is that high-ability children from low SES families cannot reach their full potential in terms of earnings and educational attainment, but instead end up closer to their lower-ability siblings (as also suggested by Papageorge and Thom, 2020, and Ronda et al., 2022).

Finally, while more research is required to uncover the mechanisms behind the observed social gradient in sibling similarity, the results suggest that life is more formed by individual endowments and considerations for individuals from high SES backgrounds as compared to individuals from low SES backgrounds. In other words, children from low SES homes not only, have worse average socioeconomic outcomes than children from high SES homes, but they also appear to have worse possibilities to develop individually. We argue that this is an important aspect of inequality in a society that largely has been overlooked in the literature. Lastly, we document a robust negative relationship between parental SES and sibling correlations in Sweden, a welfare state in which sibling correlations for the country as a whole generally tend to be lower than in most other industrialized countries. Future studies from other countries, using similar levels of SES granularity as in the current study, is therefore of potential interest.

References

- Acciari, P., A. Polo, and G. L. Violante (2022). "And yet it moves: Intergenerational mobility in Italy". *American Economic Journal: Applied Economics* 14.3, 118–163.
- Adermon, A., M. Lindahl, and M. Palme (2021). "Dynastic human capital, inequality, and intergenerational mobility". *American Economic Review* 111.5, 1523–1548.
- Ahsan, M. N., M. S. Emran, H. Jiang, Q. Han, and F. Shilpi (2022). "Growing Up Together: Sibling Correlation, Parental Influence, and Intergenerational Educational Mobility in Developing Countries".
- Almond, D. and B. Mazumder (2013). "Fetal origins and parental responses". *Annual Review of Economics* 5.1, 37–56.
- Anger, S. and D. D. Schnitzlein (2017). "Cognitive skills, non-cognitive skills, and family background: evidence from sibling correlations". *Journal of Population Economics* 30.2, 591–620.
- Baier, T. (2019). "Does sibling and twin similarity in cognitive ability differ by parents' education?" *Zeitschrift für Familienforschung* 31.1, 58–82.
- Becker, G. S., S. D. Kominers, K. M. Murphy, and J. L. Spenkuch (2018). "A theory of intergenerational mobility". *Journal of Political Economy* 126, S7–S25.
- Becker, G. S. and N. Tomes (1976). "Child endowments and the quantity and quality of children". *Journal of Political Economy* 84.4, S143–S162.
- Behrman, J. R., R. A. Pollak, and P. Taubman (1982). "Parental preferences and provision for progeny". *Journal of Political Economy* 90.1, 52–73.
- Behrman, J. R., M. R. Rosenzweig, and P. Taubman (1994). "Endowments and the allocation of schooling in the family and in the marriage market: the twins experiment". *Journal of Political Economy* 102.6, 1131–1174.
- Behrman, J. R. and P. Taubman (1989). "Is schooling "mostly in the genes"? Nature-nurture decomposition using data on relatives". *Journal of Political Economy* 97.6, 1425–1446.
- Björklund, A. and M. Jäntti (2009). "Intergenerational income mobility and the role of family background". *Oxford Handbook of Economic Inequality*. Ed. by W. Salverda, B. Nolan, and T. M. Smeeding. Oxford: Oxford University Press, 491–521.
- Björklund, A. and M. Jäntti (2020). "Intergenerational mobility, intergenerational effects, sibling correlations, and equality of opportunity: A comparison of four approaches". *Research in Social Stratification and Mobility* 70, 100455.
- Björklund, A., M. Jäntti, and M. J. Lindquist (2009). "Family background and income during the rise of the welfare state: brother correlations in income

for Swedish men born 1932–1968". *Journal of Public Economics* 93.5–6, 671–680.

- Björklund, A., L. Lindahl, and M. J. Lindquist (2010). "What more than parental income, education and occupation? An exploration of what Swedish siblings get from their parents". *The BE Journal of Economic Analysis & Policy* 10.1.
- Bratberg, E., J. Davis, B. Mazumder, M. Nybom, D. D. Schnitzlein, and K. Vaage (2017). "A comparison of intergenerational mobility curves in Germany, Norway, Sweden, and the US". *The Scandinavian Journal of Economics* 119.1, 72–101.
- Breen, R. and J. Ermisch (2021). "Sibling models, categorical outcomes, and the intra-class correlation". *European Sociological Review* 37.3, 497–504.
- Carlsson, M., G. B. Dahl, B. Öckert, and D. O. Rooth (2015). "The effect of schooling on cognitive skills". *Review of Economics and Statistics* 97.3, 533–547.
- Carlstedt, B. (2000). *Cognitive abilities aspects of structure, process, and measurement*. Vol. 148. Göteborg Studies in Educational Sciences. Göteborg: Acta Universitatis Gothoburgensis.
- Chetty, R., N. Hendren, P. Kline, and E. Saez (2014). "Where is the land of opportunity? The geography of intergenerational mobility in the United States". *The Quarterly Journal of Economics* 129.4, 1553–1623.
- Conley, D. and R. Glauber (2007). "Family background, race, and labor market inequality". *The Annals of the American Academy of Political and Social Science* 609.1, 134–152.
- Conley, D. and R. Glauber (2008). "All in the family?: Family composition, resources, and sibling similarity in socioeconomic status". *Research in Social Stratification and Mobility* 26.4, 297–306.
- Conley, D., K. M. Pfeiffer, and M. Velez (2007). "Explaining sibling differences in achievement and behavioral outcomes: The importance of withinand between-family factors". *Social Science Research* 36.3, 1087–1104.
- Dahan, M. and A. Gaviria (2001). "Sibling correlations and intergenerational mobility in Latin America". *Economic Development and Cultural Change* 49.3, 537–554.
- Das, J., S. Dercon, J. Habyarimana, P. Krishnan, K. Muralidharan, and V. Sundararaman (2013). "School inputs, household substitution, and test scores". *American Economic Journal: Applied Economics* 5.2, 29–57.
- Edin, P. A., P. Fredriksson, M. Nybom, and B. Öckert (2022). "The rising return to noncognitive skill". *American Economic Journal: Applied Economics* 14.2, 78–100.

- Fan, W. and C. Porter (2020). "Reinforcement or compensation? Parental responses to children's revealed human capital levels". *Journal of Population Economics* 33.1, 233–270.
- Fredriksson, P., B. Öckert, and H. Oosterbeek (2016). "Parental responses to public investments in children: Evidence from a maximum class size rule". *Journal of Human Resources* 51.4, 832–868.
- Frijters, P., D. W. Johnston, M. Shah, and M. A. Shields (2013). "Intrahousehold resource allocation: do parents reduce or reinforce child ability gaps?" *Demography* 50.6, 2187–2208.
- Grätz, M. (2018). "Competition in the family: inequality between siblings and the intergenerational transmission of educational advantage". *Sociological Science* 5, 246–269.
- Grätz, M., K. J. Barclay, Ø. N. Wiborg, T. H. Lyngstad, A. Karhula, J. Erola, Patrick Präg, Thomas Laidley, and D. Conley (2021). "Sibling similarity in education across and within societies". *Demography* 58.3, 1011–1037.
- Grätz, M. and M. Kolk (2022). "Sibling similarity in income: A life course perspective". *Research in Social Stratification and Mobility* 78, 100688.
- Grätz, M. and F. Torche (2016). "Compensation or reinforcement? The stratification of parental responses to children's early ability". *Demography* 53.6, 1883–1904.
- Griliches, Z. (1979). "Sibling models and data in economics: Beginnings of a survey". *Journal of Political Economy* 87.5, S37–S64.
- Hällsten, M. and M. Thaning (2022). "Wealth as one of the "Big Four" SES dimensions in intergenerational transmissions". *Social Forces* 100.4, 1533– 1560.
- Hsin, A. (2012). "Is biology destiny? Birth weight and differential parental treatment". *Demography* 49.4, 1385–1405.
- Karhula, A., J. Erola, M. Raab, and A. Fasang (2019). "Destination as a process: Sibling similarity in early socioeconomic trajectories". *Advances in Life Course Research* 40, 85–98.
- Lindqvist, E. and R. Vestman (2011). "The labor market returns to cognitive and noncognitive ability: Evidence from the Swedish enlistment". *American Economic Journal: Applied Economics* 3.1, 101–128.
- Mood, Carina, Jan O. Jonsson, and Erik Bihagen (2012). "Socioeconomic Persistence across Generations: Cognitive and Non-cognitive Processes". From Parents to Children: The Intergenerational Transmission of Advantage. Ed. by John Ermisch, Markus Jäntti, and Timothy M. Smeeding. New York: Russell Sage, 53–83.

- Papageorge, N. W. and K. Thom (2020). "Genes, education, and labor market outcomes: evidence from the health and retirement study". *Journal of the European Economic Association* 18.3, 1351–1399.
- Pop-Eleches, C. and M. Urquiola (2013). "Going to a better school: Effects and behavioral responses". *American Economic Review* 103.4, 1289–1324.
- Restrepo, B. J. (2016). "Parental investment responses to a low birth weight outcome: who compensates and who reinforces?" *Journal of Population Economics* 29.4, 969–989.
- Ronda, V., E. Agerbo, D. Bleses, P. B. Mortensen, A. Børglum, O. Mors, et al. (2022). "Family disadvantage, gender, and the returns to genetic human capital". *The Scandinavian Journal of Economics* 124.2, 550–578.
- Rosenzweig, M. R. and K. I. Wolpin (1988). "Heterogeneity, intrafamily distribution, and child health". *Journal of Human Resources*, 437–461.
- Rosenzweig, M. R. and J. Zhang (2009). "Do population control policies induce more human capital investment? Twins, birth weight and China's "onechild" policy". *The Review of Economic Studies* 76.3, 1149–1174.
- Savelyev, P. A., B. C. Ward, R. F. Krueger, and M. McGue (2022). "Health endowments, schooling allocation in the family, and longevity: Evidence from US twins". *Journal of Health Economics* 81, 102554.
- Solon, G. (1999). "Intergenerational mobility in the labor market". *Handbook of labor economics*. Vol. 3. Elsevier, 1761–1800.
- Solon, G., M. Corcoran, R. Gordon, and D. Laren (1991). "A longitudinal analysis of sibling correlations in economic status". *Journal of Human Resources* 26.3, 509–534.
- Vosters, K. and M. Nybom (2017). "Intergenerational persistence in latent socioeconomic status: evidence from Sweden and the United States". *Journal* of Labor Economics 35.3, 869–901.
- Yi, J., J. J. Heckman, J. Zhang, and G. Conti (2015). "Early health shocks, intra-household resource allocation and child outcomes". *The Economic Journal* 125.588, F347–F371.

Appendix A: Additional figures and tables



Figure A1. Sibling correlations in income and education by ventiles of parents' income (men and women)

Notes: The figure shows sibling correlations in income and years of schooling by parental income ventiles in a sample in which both men and women are included. The line shows a second-order polynomial fitted line for the sibling correlations.



Figure A2. Sibling correlations in income and education by ventiles of parents' education (men and women)

Notes: The figure shows sibling correlations in income and years of schooling by parental education ventiles in a sample in which both men and women are included. The line shows a second-order polynomial fitted line for the sibling correlations.



Figure A3. Sibling correlations in income by different quantiles of parents' income

Notes: The figure shows sibling correlations in income for men by different divisions of parental income quantiles, where parents' income has been divided into 2, 5, 10, 20 or 40 quantiles.



Figure A4. Sibling correlations in education by different quantiles of parents' income

Notes: The figure shows sibling correlations in education for men by different divisions of parental income quantiles, where parents' income has been divided into 2, 5, 10, 20 or 40 quantiles.



Figure A5. Sibling correlations in cognitive skills by different quantiles of parents' income

Notes: The figure shows sibling correlations in cognitive ability for men by different divisions of parental income quantiles, where parents' income has been divided into 2, 5, 10, 20 or 40 quantiles.

Figure A6. Sibling correlations in non-cognitive skills by different quantiles of parents' income



Notes: The figure shows sibling correlations in non-cognitive ability for men by different divisions of parental income quantiles, where parents' income has been divided into 2, 5, 10, 20 or 40 quantiles.

Figure A7. Sibling correlations in earnings and education by parental income ventiles (family structure held constant across parent income ventiles)



Notes: The figure shows sibling correlations in earnings and years of schooling for male siblings by parental income ventiles. The line shows a second-order polynomial fitted line for the sibling correlations. The sample is limited to families with two children with an age gap of four years or less.



Figure A8. Sibling correlations in cognitive and non-cognitive skills by parental income ventiles (family structure held constant across parent income ventiles)

Notes: The figure shows sibling correlations in cognitive and non-cognitive ability for male siblings by parental income ventiles. The line shows a second-order polynomial fitted line for the sibling correlations. The sample is limited to families with two children with an age gap of four years or less.



Figure A9. Sibling correlations in education by parental education quintiles

Notes: The figure shows sibling correlations in schooling measured in ranks by parental education, where education is divided into 5 quintiles. The line shows a second-order polynomial fitted line for the sibling correlations. The sample is limited to children born between the years 1945–1976 and parents born between 1930–1939.



Figure A10. Sibling correlations in education by parental education ventiles

Notes: The figure shows sibling correlations in schooling measured in ranks by parental education quartile, where education is divided into 20 ventiles. The line shows a second-order polynomial fitted line for the sibling correlations. The sample is limited to children born between the years 1945–1976, and parents born between 1930–1939.

Figure A11. Sibling correlations in income by ventiles of parents' income (income measured at ages 25–30)



Notes: The figure shows sibling correlations in income ranks for men by parental income ventiles. The line shows a second-order polynomial fitted line for the sibling correlations. Income is measured as an average for the ages 25–30.

Figure A12. Sibling correlations in income and education by ventiles of parents' income (Extended parent sample)



Notes: The figure shows sibling correlations in income ranks for men by parental income ventiles. The line shows a second-order polynomial fitted line for the sibling correlations. The sample consist of males born between 1965 and 1982 and their parents, using income data 1968–2019.

Figure A13. Sibling correlations in skills by ventiles of parents' income (Extended parent sample)



Notes: The figure shows sibling correlations in cognitive and non-cognitive skills by parental income ventiles. The line shows a second-order polynomial fitted line for the sibling correlations. The sample consist of males born between 1965 and 1982 and their parents, using income data 1968–2019.

Essay 4: Making Background Salient: The Effects of Open Access Criminal Databases on Offender Behavior

Co-authored with Hans Grönqvist, Susan Niknami and Mårten Palme

Acknowledgements: This paper has benefited from discussions with Will Dobbie, Rasmus Landersø, Arizo Karimi, Jens Ludwig, Enrico Moretti, Peter Nilsson, Sam Norris, Aurelie Ouss, Crystal Yang and seminar participants at Uppsala Labor Group, IFAU and Linnaeus University. Adam Birgersson and Gabriel Nielsen provided excellent research assistance. We thank Marcos Demetry for generously sharing his data. We are also grateful to the company behind the database for providing some of the data used in the analysis and for graciously taking the time answering our questions. Funding from Handelsbankens forskningsstiftelser and the Swedish Research Council is gratefully acknowledged.

1 Introduction

Criminal background information is arguably more readily available to the general public today than ever before. Large online databases containing information on criminal offenders, accessible through state agencies or private background check firms, have substantially reduced the cost for citizens. employers, and landlords to access criminal records.¹ The effect of these databases on the outcomes of the offenders is heavily debated. On the one hand, it is possible that being included in criminal databases could improve the outcomes of offenders through specific deterrence (Anker et al., 2021; Chalfin and McCrary, 2017). On the other hand, the information could send a negative signal that harm rehabilitation efforts. The potential for this is supported by results in quasi-experimental studies showing that arrests and incarceration reduce formal sector employment (Mueller-Smith, 2015; Dobbie et al., 2018; Dobbie et al., 2018; Grenet et al., 2024). Because of this concern, many states in the US have implemented reforms to conceal criminal background information, e.g. through expungement (Prescott and Starr, 2020; Agan et al., 2024; Selbin et al., 2018), sealing (Mueller-Smith and Schnepel, 2021), or "Banning-the-Box" (Doleac and Hansen, 2020; Rose, 2021). While the literature has mostly focused on how employers respond to this type of information,² the behavioral adjustments of the offenders actually included in these databases are much less clear. Most empirical studies also struggle with the fact that having a criminal record is not random and therefore likely correlates with unobserved risk factors.

We address these issues by studying the effects of accessing criminal records within the context of the unexpected launch of Sweden's first online criminal database. The platform, introduced in January 2014, transformed the way criminal background checks were conducted by enabling the general public to anonymously search by name for individuals charged with a crime. The criminal background information is sourced from records mandated by law to be provided by courts. Unlike most criminal databases, which offer limited background information on specific crime types (e.g., sex crimes) only for some groups of offenders and target certain users (e.g., employers or landlords), this database covers all criminal charges in Swedish courts and is accessible to everyone without the need for a user profile or any costs. Search results map name to current age and place of residence and reveal whether an individual has faced a criminal charge. For a nominal fee (approximately USD 5), users can obtain the complete verdict, detailing the crime type, conviction year, sentence, and any additional circumstances of the crime. The platform rapidly

¹This development has been particularly stark in the United States where it is estimated that the three largest background screening companies alone (out of approximately 2,000) in 2019 conducted 56 million searches (NCLC, 2019).

²See e.g. Pager (2003), Holzer et al. (2006), Agan and Starr (2018), Cullen et al. (2024), Agan et al. (2024), Doleac and Hansen (2020), Rose (2021), and Finlay (2008).

garnered widespread use. By 2019, when competing platforms were finally launched, searches for criminal records on the site had neared 100 million in total, averaging around 10 searches per person in Sweden.

To identify the causal effects of revealing criminal background information, we exploit a legislative rule which stipulates that courts should not disclose individual information on criminal charges dating back more than five years since the inquiry. Consequently, this legislation created a time window beyond which the company behind the platform was unable to access criminal records information. By analyzing data provided by the company we were able to determine the specific cutoff month for each court, which depend on the date when the company applied for and was later granted access to the records. We then utilize administrative data containing information on the complete criminal histories of individuals. These data enable us to distinguish between individuals charged after the specified cutoff date, resulting in their criminal records being exposed online, and non-exposed defendants who were charged prior to the cutoff date. The data also contains a wide array of labor market, educational, and demographic characteristics, which allows us to investigate behavioral responses of the defendants.

We start by verifying empirically that individuals on both sides of the cutoff date are similar in their baseline characteristics and that the pre-reform trends for exposed and non-exposed offenders are parallel. Moreover, we show that there is no evidence of bunching of cases or changes in offender characteristics near the cutoff. These findings confirm that neither courts nor offenders were able to foresee and respond to the future arrival of the platform.

We then proceed to the results from our difference-in-differences model which accounts for correlated unobservables by contrasting the change in outcomes for exposed versus non-exposed defendants. Since all individuals in our sample were charged with a crime and exposure is defined before the platform was introduced our empirical design also addresses potential concerns of general deterrence effects. The results suggest that exposure significantly decrease log earnings by 9.3 percent. This effect size corresponds to about one quarter of that from event study estimates of the effect of a criminal charge on earnings in the period before the platform was introduced. We find no significant effect on labor supply at the extensive margin or criminal recidivism. There are, however, significantly stronger detrimental effects on both labor market outcomes and recidivism in defendant subgroups: those with at least a high school degree, acquitted individuals, native born, and those either charged with a misdemeanor or living in places with a relatively low concentration of ex-criminals. These are groups of the population with arguably a relatively higher opportunity cost of engaging in crime. In many of these groups we also find adverse effects on educational performance among the children of the defendants.

When trying to identify the sources of the earnings losses we find small and mostly insignificant exposure effects on behaviors such as enrolling in labor market training programs, residential mobility, family formation, and sorting across industries based on the share of ex-offenders working in a given industry. However, we find a significant differential effect in defendant sorting with respect to the firm earnings premium estimated using the methods suggested by Abowd et al. (1999). Our results suggest that workers in the subgroups most strongly affected by the mark of a criminal record are more likely to work in firms that pay a lower earnings premia. This observation aligns with job ladder models, which suggest that workers continuously search for better job opportunities throughout their careers, gradually transitioning to higher-paying employers. In these models, pay disparities among firms reflect differences in how the surplus is distributed between workers and employers (Burdett and Mortensen, 1998; Haltiwanger et al., 2018; Lachowska et al., 2020). Within this framework, affected workers suffer earnings losses due to the impact of moving down the job ladder to employers that offer lower wages.

Our results are robust to a battery of specification checks, including changing the bandwidth for inclusion in our sample, and we also show that there is no significant effect when assigning court-specific placebo cutoff dates one year before and one year after the actual cutoff date.

Our study provides some of the very first pieces of evidence regarding the causal effect of open access to online criminal background information on the behavior of offenders. One of few exceptions is the study by Finlay (2008) who uses cross-state cross-time variation to study the labor market effects of employer access to state-provided online information on incarceration history for individuals with moderate to long prison sentences. The results show adverse employment consequences in states with open records policies.³ Our study differs in that the database includes the universe of individuals charged with a crime regardless their sentence and type of crime committed. This is important since, for instance, the ciminal records of felony conviction may have different consequences for defendants, relative to criminal records of misdemeanor charges. We are also able to document the effects for a wide range of outcomes as well as probe underlying behavioral mechanisms.

Our results also add to the literature on the impact of criminal background checks from the employers' perspective.⁴ This body of literature either studies the callback rate to fictitious job applications signaling criminal background or examines policies intended to influence the hiring practices of employers. A common finding in the former line of research is that employers are averse to calling back or hiring job applicants with criminal records (Pager, 2003;

³Finlay (2008) studies a sample of individuals included in the National Longitudinal Survey of Youth (NLSY), in which 369 offenders had received prison sentences for a minimum of one year.

⁴These studies are typically motivated by the fact that individuals with a criminal record are more likely to be unemployed. For instance, Bushway et al. (2022) show that 46 % of unemployed men in the United States have been convicted. Other studies include e.g. Western (2002) and Freeman (1992).

Holzer et al., 2006; Agan and Starr, 2018; Uggen et al., 2014; Cullen et al., 2024).⁵ The latter line of research primarily focuses on Ban-the-Box (BTB) legislation, which restricts inquiries about criminal histories on job applications until later in the hiring process. It often relies on state-level variations in these laws to address correlated unobservables. Doleac and Hansen (2020) demonstrate that these policies decrease the probability of employment by 3.4 percentage points for young, low-skilled black men, suggesting that in the absence of information about an applicant's criminal history, employers statistically discriminate against demographic groups with higher proportions of ex-offenders. Rose (2021) finds that a 2013 Seattle BTB law had minimal effects on ex-offenders' labor market outcomes, indicating that employers may delay background checks until the final stages of the hiring process or that exoffenders may gravitate towards jobs where clean records are not obligatory.⁶ While the responses of employers to criminal background information represents a potentially important mechanism, the setting we consider allows us to investigate the effects of criminal background information that is much more widely available also to the general public.

A few recent studies examine policies that not necessarily only target employers. Agan et al. (2024) study the labor market impacts of retroactively reducing felonies to misdemeanors in San Joaquin County, CA where criminal justice agencies reclassified certain theft and drug offenses from felonies to misdemeanors, without requiring input or action from affected individuals. The results reveal no discernible benefits among individuals whose records were reduced proactively.⁷ Mueller-Smith and Schnepel (2021) study diver-

⁵For instance, Pager (2003) demonstrates that disclosing a criminal record to employers who do not explicitly request such information has minimal impact on hiring decisions. Agan and Starr (2018) reveal a significant six-fold increase in callback rates for white-to-black applicants following the implementation of Ban-the-Box legislation in New Jersey and New York. Cullen et al. (2024) conduct experimental research to assess strategies for increasing the demand for workers with criminal records. Their study involves presenting hiring managers with hypothetical scenarios that influence whether workers with criminal records can accept job offers in the future. Results indicate that initially, 39% of employers are open to hiring individuals with criminal records. However, this figure rises to over 50% when managers are offered incentives such as crime insurance, a single performance review, or a limited background check covering only the past year.

⁶Furthermore, Shoagii and Veugeriii (2016) estimate the effects of BTB by considering impacts on residents of high-crime versus low-crime neighborhoods. The results show positive effects on employment in high-crime neighborhoods, which the authors rationalize by minority men may benefit from the law overall, despite negative impacts on some sub groups as suggested by Doleac and Hansen (2020). Jackson and Zhao (2017) use unemployment insurance records to study a 2010 BTB reform in Massachusetts in a difference-in-differences framework. Their results suggest that BTB lowered ex-offender's employment by 2.4 percentage points and quarterly earnings by USD 300.

⁷Agan et al. (2024) also conduct a field experiment notifying a subset of individuals about their proactive reduction also showing null results, implying that lack of awareness is unlikely to explain the findings.

sion programs in Harris County, Texas, providing first-time felony defendants with an opportunity to circumvent a criminal record either by reclassifying drug and property offenses or by diverting low-risk defendants. Their findings suggest that these programs significantly reduce recidivism rates and improve quarterly earnings by about 50 percent. We add to this literature by studying an intervention that not only affected certain groups of felony offenders but the universe of individuals charged with a crime. We also contribute by using data that allows us to shed light on the sources underlying the labor market penalty of having a criminal record.

Importantly, our results are connected to an extensive literature that examines the effects of criminal justice sanction strictness on specific deterrence. often finding mixed evidence.⁸ However, while the formal sanctions studied in this literature (e.g. incarceration or community service) are the criminal justice system's primary tool to discourage deviant behavior, many scholars also argue that informal social sanctions attached to the stigma of being marked as a criminal may potentially also act as a crime deterrent (Chalfin and Mc-Crary, 2017; Zimring et al., 1973; Nagin and Pogarsky, 2003; Weibull and Villa, 2005; Funk, 2004).⁹ In this case, conveying socially valuable information on criminal background may actually be an efficient low-cost strategy for society to combat crime.¹⁰ On the other hand, the criminal labeling theory, one of the most fundamental models in criminology, predicts that exposing criminal background will amplify offending behavior. According to this view, interactions with the criminal justice system labels the offender a criminal, which leads him to internalize stigmatizing attitudes and conform to a deviant identity (e.g. Becker, 1968; Lemert, 1967).¹¹ However, since a conviction almost always involves at least one sanction, it has proven almost impossible to separate informal sanction costs from the imposed formal ones (Chalfin and

⁸Specific studies include but are not limited to: Agan et al. (2023), Abrams (2012), Aizer and Doyle Jr (2015), Barbarino and Mastrobuoni (2014), Bhuller et al. (2020), Buonanno and Raphael (2013), Dobbie et al. (2018), Dobbie et al. (2018), Drago et al. (2009), Garin et al. (2023), Helland and Tabarrok (2007), Hjalmarsson and Lindquist (2022), Kessler and Levitt (1999), Kling (2006), Kuziemko (2013), Lee and McCrary (2017), Mueller-Smith (2015), Owens (2009), Grönqvist et al. (2022), Ouss et al. (2023), Rose and Yotam (2021), and Raphael (2014).

⁹In fact, in the Becker (1968) economic model of crime, informal sanction costs alone can fully predict criminal behavior if individuals are more sensitive to changes in the probability of apprehension, thus acquiring a criminal record, than to changes in the sanction regime (Chalfin and McCrary, 2017; Nagin, 2013).

¹⁰See e.g. Mungan (2016), Nagin (2013), Polinsky and Shavell (2007), Kahan and Posner (1999), and Rasmusen (1996).

¹¹The concept of labeling is also evident in the economic model of crime. Once an individual is labeled a criminal, the opportunity cost of refraining from crime may decrease, as potential employers may limit their access to job opportunities (Agan and Starr, 2018; Becker, 1968; Grogger, 1995; Holzer et al., 2006).

McCrary, 2017).¹² This is important as it is fully possible that these mechanisms operate in different direction.

A few recent studies provide quasi-experimental evidence from contexts that arguably approaches distinguishing between these effects. Agan et al. (2023) estimate the causal effects of misdemeanor prosecution on defendants' subsequent criminal justice involvement. The analysis exploits the quasirandom assignment of nonviolent misdemeanor cases to District Attorneys in Suffolk County, MA, who vary in their leniency of prosecuting decisions. The results show that nonprosecution of a nonviolent misdemeanor offense leads to a 53 percent reduction in the likelihood of a new criminal complaint. These effects are largest for defendants without prior criminal records, suggesting that averting criminal record acquisition is an important mechanism. Kamat et al. (2024) use a random-judge design and show that misdemeanor convictions cause an increase in the number of new offenses committed over the following five years while incarceration on more serious felony charges has no effect after release. The study by Mueller-Smith and Schnepel (2021) on diversion programs arguably also approaches distinguishing between these effects. While these studies compellingly empirically rule out many alternative explanations the authors are also clear that there could also be other mechanisms aside from stigma that the data do not allow them to eliminate.¹³ To our knowledge, no prior study has employed a quasi-experimental research strategy where the empirical approach by design is capable of identifying the impact of informal sanctions net of that of formal sanctions.

The rest of this paper is structured as follows. Section 2 describes the institutional background. Section 3 discusses the data and sample construction. We outline the empirical strategy in Section 4. The results are provided in Section 5 followed by conclusions in Section 6.

2 Background

In most countries, criminal background checks serve as a key tool utilized by various actors such as employers, landlords, and higher education institutions to address concerns surrounding public safety and the prevailing notion that individuals with criminal histories may be less inclined to fulfill their responsibilities. Additionally, these checks are often employed to mitigate potential liability, particularly in the realms of negligent hiring or negligent renting doc-

¹²Most empirical studies compare offenders either convicted or incarcerated with those not formally sanctioned e.g. Bernburg and Krohn (2003), Farrington et al. (1978), and Murray et al. (2017).

¹³For instance, selective out-migration, re-sentence deterrence connected to the original offense and changes in monitoring by law enforcement in response to defendants' observable criminal histories could in some cases not fully be discarded.

trines.¹⁴ The criminal background information tend to be accessible through various channels, including direct provision by states and an increasing number of private companies, especially in the United States, offering background search services.

The databases used for the records checks compile data from diverse origins, often procured in bulk from public sources such as law enforcement agencies, state courts, or through web scraping technology from public websites (NLCS, 2019). The checks tend to reveal current criminal charges or past arrests, but may also encompass non-conviction data, including records of police interactions, mere allegations, withdrawn or stayed charges, as well as acquittals.¹⁵ A background check typically contains limited details about listed offenses but generally includes the individual's name associated with the record, the jurisdiction of origin, the record's creation date, and a case number or law enforcement identifier (NLCS, 2019). While a brief description of the offense, such as "possession of marijuana", may be provided, details about the offense or related circumstances are usually omitted. Some public sources only display criminal convictions, whereas private firms may report arrests for misdemeanors that later result in dismissal (Bushway and Sweeten, 2007).

2.1 The Swedish setting

The regulations governing the request of criminal records in Sweden have traditionally been stringent, primarily for employment-related purposes (Backman, 2012). Employers are strictly limited in their ability to request a candidate's criminal record, typically permitted only at the concluding stages of the hiring process. Subsequently, the job applicant is required to formally request his criminal record from the police, who dispatches it in a sealed envelope. Upon receipt of the record, the applicant must then forward the envelope directly to the employer. This protocol often results in a significant time delay, with the entire process spanning several weeks. It is therefore not surprising that survey data from a large sample of employers collected in 2011 indicates a relatively low utilization of criminal background checks by employers in Sweden, with only approximately 14 percent of the 1,200 employers reported to have engaged in such practices (SOU, 2011). This conclusion is also supported by Figure A2 showing that only about two percent of the job ads posted on the Public Employment Office's in 2011 mentioned criminal background checks. Consistent with this, the same year the ratio of the number of official

¹⁴About 94 percent of employers in the United States conduct some form of criminal history check, and about 90 percent of landlords run background checks on prospective tenants (NLCS, 2019). In 2004, a survey from the United States showed that about 50 percent of the employers check the criminal background of job applicants (Holzer et al., 2006)

¹⁵Several US states have laws prohibiting the use of arrests that did not result in convictions in employment screenings, e.g. https://ccresourcecenter.org/state-restoration-profiles/50-state-comparisoncomparison-of-criminal-records-in-licensing-and-employment/

background checks to the number of firms was 17 percent. Since then, there has been a slowly growing trend towards increased use of official background checks.

The information provided in the official criminal background checks is expunged after 3-10 years. The exact amount of time depends on factors such as the age of the offender when committing the crime and the sentence.

2.2 The database

In January 2014, the online platform Lexbase completely transformed the landscape of criminal background checks in Sweden. Instead of targeting solely employers, the platform offered unrestricted access to criminal records for the general public. The database allowed anyone to anonymously search for charged individuals by name, leveraging data sourced from court records mandated by law to be provided upon request. The check maps the name of the individual to current age and place of residence and reveals whether that specific person has been charged with a crime. This means that even for common names it was typically possible to uniquely identify an individual.

Unlike many other online criminal databases, which often specialize in specific crime types and cater exclusively to certain actors like employers or landlords, the platform presents a comprehensive repository of every criminal charge in Swedish courts, without any fees or user profile. For a nominal fee, at the time approximately USD 5, registered users could also obtain full court proceedings, gaining access to information about the number of charges, conviction and crime date, type of crime, circumstances surrounding the offense, the sentence, and the address and personal id number (cf. social security number) of the individual. However, acquiring the complete court proceedings seems to have been a rare occurrence. Data from the database's company indicate that 0.16 % of all background searches actually resulted in a purchased court proceeding in the first year (see Appendix Figure 6). When it was introduced, there was no information on the website that indicated that the database also included acquittals, only that it contained the records of individuals convicted in court.¹⁶

¹⁶The following text is a translation of the official press release from the company sent on the day of the launch: "All criminal convictions from the last five years are now easily searchable - New unique site is launched today. Today, a beta version of a new site, Lexbase.se, is being launched, which makes all criminal convictions, five years or younger, easily searchable. Lexbase will be Sweden's largest database with legal information and a new ground-breaking service on the Swedish market." (https://www.lexbase.se/site/press/21?s=285b36b14745ae4aedb1d218ea9989c4). Reports in the media also highlighted that only criminal convictions were included. For instance, the following text is a translation of parts of a report in one of Sweden's largest newspapers called Expressen. "At Lexbase you could find out everything about the convictions of your friends, neighbors and acquaintances. Is the neighbor convicted of rape? Or maybe just for speeding?

Appendix Figure A4 show the homepage of the platform and Appendix Figure A5 shows the different stages in the search process.

The platform immediately received large media attention and gained immense popularity. Data provided by the company behind the platform show that 32 million background checks were conducted in just the first year. Appendix Figure 6 shows that by 2019 when the first competing platform was introduced, the cumulative number of searches had soared to 103 million. Appendix Figure A6 shows the sharp increase in the number of media articles written in printed pres mentioning the database. Appendix Figure A7 shows that from the week after its inception and up until 2019, Lexbase consistently outnumbered other popular Google search terms in Sweden, surpassing common queries such as "new car" or "best restaurants".

Since the platform did not have any competitors until 2019, the massive use of the database cannot be explained by a shift in the use from other similar platforms. It is also seems unlikely that employers substituted away from official background checks by the police to instead using the platform. First, the introduction of the platform did not coincide with any meaningful shift in the long-run trend towards increased use of official criminal background checks (see Appendix Figure A2). Secondly, the demand for regular background checks was relatively low among employers. As discussed earlier, a large survey of 1,200 firms just before the introduction of the platform revealed that 14 percent of employers reported that they check criminal records (SOU, 2011) and only two percent of the job ads posted on the Public Employment Office's mentioned criminal background checks. Third, the number of official background checks, 171,229 in 2011, is completely dwarfed by the about 20 million of searches conducted every year on the platform. Taken together, the large use of the platform is most likely explained by a high demand for the service in the general public.

The platform was closed down by the Internet service provider just two days after its launch, after the site came under heavy criticism from Swedish media and the general public for making criminal records too easily accessible. However, since April 2014 the platform has been fully operational. Several attempts by the Swedish government to shut town the platform have failed because the information is protected by the Swedish constitution.

The only thing you needed to find the convictions was their name or social security number. The convictions were exposed as red dots on a map and could then be downloaded for a small fee. The new site was launched at a high-profile press conference on Monday. Already on the same day, lots of curious people started logging in, while harsh criticism was immediately directed at Lexbase. And immediately the first reports from desperate people came in to the Chancellor of Justice (JK). "I was so sick I couldn't go to work. I'm in that database. I've done everything to get back to a normal life and have been holding a job for the past two years. If someone searches for me, I'll lose my job immediately. You have to do something about this. My neighbors can see my house on a map with a red dot. I'm totally stressed out now by this! I feel so damn bad I might kill myself," one man writes to JK ." (https://www.expressen.se/nyheter/deras-liv-forstors-av-lexbase/)

While the information provided by the database in theory was accessible through public records even before 2014, the process was laborious, necessitating visits to multiple courts to potentially find a matching name. Just as in the case of most other online criminal databases, the introduction of this particular platform thus represented a significant reduction in the costs of accessing criminal background information, especially for the general public.

2.3 Mapping to empirical design

The company behind the platform was officially established as a business entity in 2012. During the two years preceding the public launch of the database, the company began acquiring criminal background data through direct engagement with individual courts, requesting access to criminal charges stored in the digital administrative system known as Vera. A government decree mandated courts to grant access to Vera to anyone seeking information on criminal charges for a given individual.¹⁷ However, after five years, courts should expunge the information.¹⁸ The courts had never before processed this type of large scale request for the universe of criminal cases. Together with some uncertainties surrounding the exact details regarding the expungement procedures, this created some variation among courts in the way they were able to respond to the request. For instance, while some courts were not able to strictly adhere to the exact expungement period, other courts not only delivered criminal cases but also civil law suits.¹⁹ In the end, the request resulted in distinct cutoff dates for each court, marking the point at which criminal charges were integrated into the database. The specific cutoff date was determined by both the timing of the company's request for information and the length of time it took for each court to process the request and then deliver the records.

To identify the cutoff dates, we obtained data from the company detailing the number of cases added to the database for each court and month. Utilizing our comprehensive micro data covering all criminal charges in Sweden since 1986, we calculated the share of cases included in the database. Our analysis comprises the 46 courts that operated during the period preceding the database's launch (i.e., 2006-2013). For each court j and month t, we computed the change in the share of cases by calculating the difference between the average share of cases in the preceding 12 months and that of the subsequent 12 months. The cutoff date was identified as the month exhibiting the

¹⁷See decree 2001:639)

¹⁸The courts were still obliged to preserve cases in printed form. In order to search for a specific case a person had to visit the court to manually search for the case number in a volume of books that compiles all the cases that have been processed by the court.

¹⁹For instance, in 2013 the Swedish Authority for Privacy Protection criticized DC Västmanland for not following the stipulated timeline (DNR 1319-2013). A similar statement was made for DC Attunda (DNR 1316-2013). Report 2012:1 by the Swedish Authority for Privacy Protection provide a detailed discussion about these issues.

largest increase in the share of cases over the entire period. In the empirical analysis we focus on the 32 courts with observed cutoff dates before 2010 to ensure the representation of all courts in our sample within the database at its launch.²⁰ Additionally, we excluded four courts with no clear evidence of cutoff.²¹

While approximately 5-6 years passed between the initial charge and the release of the platform, the open access version of the database did not include the date of the charge. This omission meant that most employers, land-lords, and ordinary citizens could not distinguish between charges based on their timing unless they actively registered on the platform and purchased the sentencing information. Consequently, from the perspective of most public agents, the launch of the platform provided new information about the individuals listed in the database.



Figure 1. Change in the weighted share of cases around the cutoff

Notes: The figure plots the weighted average share of Lexbase cases across all district courts in our sample within a 12 month window around the court-specific cutoff date

Figure 1 shows the average share of cases included in the database around the cutoff for the courts in our sample. Consistent with not all courts strictly adhering to the stipulated timeline for expungement, the cutoff is not sharp. However, there is a clear discontinuity in the share of cases with the an estimated size of the cutoff of 0.64 (0.13).²² ²³ When interpreting this figure it should be noted that the estimates are affected by measurement error. First, our register based micro data do not include the complete number of cases that

²⁰It is unclear whether all courts were actually included in the database at the time it was launched. Thus, it is possible that some courts were added at a later point in time.

²¹These courts, which represent 7.5 percent of the total number of cases, are DC Umeå, DC Gällivare, DC Haparanda, DC Värmland.

 $^{^{22}}$ The estimated size of the cutoff is 0.68 (0.12) when not weighting by the size of the courts as measured by the number of cases.

²³Figure A8 show the cutoff for each court in our sample.

exist in Lexbase since we miss information on individuals that are lacking a personal id number. Most of these individuals are asylum seekers. Second, in some cases, courts accidentally delivered not only criminal cases but also civil cases that were not made searchable on the platform.²⁴

3 Data and sample construction

3.1 Administrative data

Our empirical analyses rely on micro data that originate from various administrative registers managed by Statistics Sweden. These registers contain information on the entire Swedish population aged 16 to 65, spanning the years from 1990 to 2016. The data holds demographic and socioeconomic information such as age, place at birth, gender, educational level, employment and income measures. To ensure comprehensive tracking of labor market outcomes for all individuals in our sample, we restrict the sample to individuals younger than 55 in the year of the initial charge.

These data have been linked to the Swedish Conviction Register maintained by the National Council for Crime Prevention (*Brottsförebyggande rådet* -BRÅ). Within these records, we have access to comprehensive details concerning criminal charges during this period. The data include information on the date of the conviction as well as the sentence. This information pertains to criminal charges in Swedish district courts, which is the court of first instance. A single criminal charge may encompass multiple crimes, and we observe all crimes within a given charge.²⁵ The data also include information on acquittals.

Appendix B describes how the variables in our analysis are constructed.

3.2 Classifying offenders

Utilizing the court-specific cutoff dates, we categorize defendants by exposure statuses employing a 12-month window. Defendants charged within 12 months after this date are classified as exposed, while those charged up to 12 months prior to this are categorized as non-exposed. We focus on first time defendants to ensure that each individual will only appear once in our estimation sample. In addition, this group is particularly important from a policy perspective as non-disclosure interventions are likely to have a greater chance to succeed for individuals before they have accumulated significant criminal

²⁴Data provided by the company behind the database show that 20.81 percent of the total number of records delivered were civil cases.

²⁵The data exclude minor offenses such as speeding tickets but include offenses such as driving without a license and DUI.

experience. This is also the reason why non-disclosure programs typically target first-time defendants.

Table A1 columns (1) and (2) shows descriptive statistics for our sample separately for exposed and non-exposed defendants. These baseline characteristics are measured in one year before the initial trial. As expected, the background characteristics of the individuals are very similar in the two groups. We can also see that there are striking similarities in terms of the case characteristics and the sentence. This does not only show that the covariates in the two groups balance but also that our research design should be able to identify the role of stigma net of that from formal sanctions.

4 Research design

In our main analysis, we estimate the following conventional difference-indifferences model where we compare the outcomes of exposed and non-exposed individuals before and after the database was introduced

$$Outcome_{it} = \alpha + \beta \text{exposed}_i * \text{post}_t + \text{exposed}_i + \gamma_t + \lambda_c + \delta X_{it} + \varepsilon_{it}$$
(1)

where $Outcome_{igt}$ is the relevant outcome of defendant *i* in calendar year *t*. $exposed_i$ is an indicator variable set to one for individuals who were charged after the cutoff date and zero otherwise; $post_t$ is an indicator variable set to one for years when the database was in place and zero otherwise. $exposed_i$ and court (λ_c) fixed effects control for persistent unobserved heterogeneity between exposed and non-exposed individuals as well as between courts. X_{it} is a vector of controls for baseline characteristics such as age, gender and immigrant. It also controls for time since the charge and seasonality in criminal charges by including calendar month fixed effects.²⁶ Since the reform affected everyone at the same time and remained in effect, the concerns typically associated with staggered difference-in-differences research designs (Roth et al., 2023) do not apply to the current setting.

This strategy relies on the standard assumption of parallel trends: the outcomes should not have evolved differently between exposed and non-exposed offenders in the absence of the database. We assess this by inspecting the estimated β coefficients for the years leading up to the launch of the platform.

The other main assumption is that of no anticipatory effects, which is violated if, for instance, courts or offenders somehow were able to foresee the introduction of the database and the exact cutoff date in each court. In this case, we would expect to see a change in the frequency of court cases around the cutoff date. For instance, courts might choose to have the trial after the cutoff date to ensure that an offender gets exposed in the database. We might

²⁶There are typically fewer cases processed by courts in the summer and around Christmas.

also expect to see changes in the characteristics of offenders around the cutoff if some offenders, for instance, more highly educated ones, are better able to predict the future arrival of the database. In this case, there should be an increase in the share of highly educated offenders just before the cutoff.

We already saw in Table A1 that the characteristics of exposed and nonexposed offenders are very similar. To examine this in more detail, we estimated the change in the frequency of court cases and in the characteristics of the offenders within a one-year window around the court-specific cutoff date. Figure A1 shows no evidence of such changes for the average number of cases in the courts and the predicted risk of recidivism.²⁷ These findings are, of course, as expected, as it is hard to think of a scenario where courts could predict the (timing of) arrival of a currently non-existing platform of this type, less which the cutoff dates would be.²⁸

5 Results

This section presents the results from our empirical analysis. We start by providing the main results and robustness. We then show results for relevant subgroups of the population. We end by examining some of the underlying mechanisms.

5.1 Main results

Table 1 shows the difference-in-differences estimates of the effect of exposure on criminal behavior and labor supply.

We can see that exposure has no statistically significant exposure effect on the probability that the individual is (re-)convicted. We can also see that there is no significant effect on employment. The estimates are relatively precise, ruling out larger increases in the probability of conviction than 0.9 percent

$$Convicted_i = \alpha + \beta X_i + \varepsilon_i$$

where X_i is a vector of the following baseline defendant characteristics, age, immigrant, educational attainment, employment, log earnings. The risk score is then defined as

$$\hat{\alpha} + X_i \hat{\beta}$$

²⁸Consistent with this, Appendix Figure A2 shows that there was no mentioning of Lexbase in Swedish media before Jan 2014.

²⁷We calculate our measure of recidivism risk from the predicted dependent variables for each individual from the following OLS regression

and captures a defendant's risk of recidivism over a five year period. Appendix Figure A10 shows similar results when estimating the model separately for each individual characteristic included in the prediction.

	Convicted		Employed		Log earnings	
	(1)	(2)	(3)	(4)	(5)	(6)
Exposed*Post	0.003	0.003	-0.007	-0.008	-0.093**	-0.093**
	(0.003)	(0.003)	(0.008)	(0.008)	(0.042)	(0.042)
Sample mean	0.088	0.088	0.666	0.666	5.835	5.835
Year FE	Yes	Yes	Yes	Yes	Yes	Yes
Court FE	Yes	Yes	Yes	Yes	Yes	Yes
Time since charge FE	Yes	Yes	Yes	Yes	Yes	Yes
Calendar month FE	Yes	Yes	Yes	Yes	Yes	Yes
Baseline controls	No	Yes	No	Yes	No	Yes
Observations	186,840	186,840	186,840	186,840	186,840	186,840

Notes: This table reports difference-in-differences estimates based on model (1) for crime and labor market outcomes. Exposed is an indicator variable set to one for defendants sentenced up to 12 months after the court-specific cutoff date; zero otherwise. Post is an indicator variable set to one for the years the database was in place (i.e. 2014 and onwards); zero otherwise. The regressions control for age, gender, immigrant and the main effect of being exposed. Numbers in bracket show the mean of the dependent variable. Standard errors in parenthesis are clustered at the individual and court level. *** = significant at 1 percent level, ** = significant at 5 percent level, * = significant at 10 percent level.

points and larger decreases in the probability of being employed by 2.4 percentage points with 95 percent confidence. This corresponds to about one third of the causal effect of incarceration on the probability of being employed.²⁹ We do, however, find a that exposure leads to a statistically significant decrease in log earnings by about 9.3 percent.³⁰ While we cannot rule out adjustments in terms of hours worked, the absence of an effect on labor supply at the extensive margin provide suggestive evidence that open access to online criminal databases may primarily hurt the outcomes of defendants by reducing wages.

Consistent with the strong similarity in baseline characteristics shown in Table A1 the estimates are almost completely invariant to adding baseline controls to the regressions. Figure A11 plots dynamic difference-in-differences

Table 1. Main Results

²⁹Dobbie et al. (2018) estimate the causal effect of incarceration on the probability of being employed in Sweden to be -7.6 percentage points.

³⁰We replace zero earnings with 1 before taking logs. The results are similar when instead estimating the model using the inverse hyperbolic sine.
estimates that mimics those in Table 1 by year relative to 2013.³¹ We find no evidence of differential pre-trends.

To further explore how exposure impacts labor market outcomes, Appendix Figure A12 plots the difference-in-differences estimates based on model (1) and corresponding 90 percent confidence intervals of the effect of exposure on the probability of a defendant's earnings falling above various thresholds. The impact of on earnings is concentrated in the left tail of the earnings distribution, with little to no effect on the probability of earning above higher thresholds than the 80th percentile. These results suggest that exposure primarily affects earnings at the low to mid-end of the earnings distribution.

The exposure effects on earnings are smaller than those shown by Mueller-Smith and Schnepel (2021) who find that diverted felony defendants in Harris County, TX experiences increases in total earnings over the ten-year follow up period grow by 93 percent and improvements in quarterly employment rates by 49 percent. The effect is also smaller than those reported by Finlay (2008) who show that ex-prisoners in US states with open records to online criminal databases experience 18.7 percent reductions in earnings. On the other hand, Agan et al. (2024) find no significant benefits among individuals whose records were reduced proactively in San Joaquin County, CA.

To which extent can differences in the results be explained by the specific Swedish context? We examine this question by following Rose (2021) and estimate the following event study specification to examine the labor market impact of a criminal conviction

$$Y_{it} = \alpha_i + \gamma_t + \beta X_{it} + \sum_{s \in [-5, 5]} \delta_s \gamma D_{ist} + \varepsilon_{it}, \qquad (2)$$

where Y_{it} is employment or log earnings for individual *i* and year *t*, α_i is an individual fixed effect, γ_t is year fixed effects, X_{it} is a vector of time-varying annual age dummies, an D_{ist} is 1 when individual *i* is *s* quarters from his first conviction. We use dummies for $s \in [-5,5]$ to estimate 5 years of dynamic effects focusing on individuals convicted for the first time between 1997 and 2010 so that the outcomes are observed for at least 5 years before and after conviction.³² We take advantage of the fact that our data provides information on the full population of individuals charged with a crime. While the

$$Outcome_{i,g,s} = \alpha + \sum_{\substack{k=2011\\k\neq 2013}}^{2016} \beta_k \cdot 1\{exposed\} \cdot 1\{s=k\} + \gamma \cdot 1\{exposed\} + \theta_s + \beta X_{i,g,s} + \varepsilon_{i,l,s},$$

where the outcome $Outcome_{i,g,s}$ is observed for individual *i*, belonging to exposure group *g* in year *s*.

³¹We estinate the following dynamic difference-in-differences model

 $^{^{32}}$ We use +/- 7 years and bin the end points after +/- 5 years to be able to identify both the age and year fixed effects.

results should primarily be interpreted as a descriptive comparative exercise, the model controls for permanent unobserved heterogeneity using individual fixed effects. The results for the two-way fixed effects model are reported in Appendix Figure A13 and the results using the procedure by Sun and Abraham (2021) are reported in Appendix Figure A14.

The results show a sharp reduction in both employment and earnings after the initial charge. The drop in earnings is about 38 percent one year after the charge. This effect size is similar to that reported by Rose (2021) who finds earnings losses by 44 percent for individuals convicted of a felony and about 26 percent for individuals convicted for a misdemeanor.³³

5.2 Further robustness checks

Table A2 provides results from further robustness checks. It is possible that unobserved seasonal breaks that potentially coincide with the timing of our cutoffs could confound with our results.³⁴ While we did not see any evidence of differences in the composition of crime types or sentences between exposed and non-exposed defendants in Table A1, Table A2 provides additional support for the absence of seasonal confounders. The table presents the estimated effects of a placebo experiment where we adjust the cutoff dates to one year earlier or one year later than the actual court-specific cutoff dates. Panels A and B show that there are no significant discontinuities in these placebo exercises.

Panels C and D show the sensitivity of our results for decreasing the bandwidth for individuals included in the treatment and control group to +/-9 months and +/-6 months. We can see that, while the sample size decreases considerably, the estimates for log earnings are similar to baseline albeit slightly larger. The other estimates are also similar and not significantly different from baseline.

There are at least two reasons to believe that our exposure effects represent a lower bound of the actual effect of being listed in the database. First, it is possible that some of the defendants that we classify as non-exposed re-offended in the period between the initial charge and the introduction of the platform and therefore also were exposed. Second, society may place less emphasis on offenses that occurred in the distant past, as is the case in our setting. However, given that the recidivism rate within our sample—comprised of first-time offenders—is relatively low, and that the date of the charge is not revealed unless the full verdict is purchased, we anticipate that such bias is limited.³⁵ We

³³As another comparison, Grenet et al. (2024) show that avoiding a prison term by substituting from prison to electronic monitoring in Sweden increases log earnings by 22.1 percent.

³⁴For instance, during holiday seasons courts may choose to process more urgent cases and postpone less serious ones.

³⁵While the one-year recidivism rate is 8.8 percent for first-time defendants it is 26.1 percent among multiple-time defendants.

empirically assess the role of this "bias" by taking advantage of the fact that the database excluded confession cases in the first two years after its launch. Confession cases generally involve less severe crimes and entail a defendant's admission of guilt, with the prosecutor, rather than a court, overseeing these cases. This institutional feature provides a way to examine whether the immediate exposure effects are different from the intermediate exposure effects. Panel E shows estimates from this alternative difference-in-differences model where the "exposed" group includes defendants whose cases where processed in court, and the "non-exposed" group comprises defendants with a confession case. In this analysis we measure the outcomes one-year post-conviction and the repeated cross-sectional dataset runs from 2010 to 2015. All regressions control for the same set of covariates as in model (1). While the results in Panel E are similar to our baseline specification, these results should, be interpreted with some caution as confession cases represent special types of crimes.^{36 37}

5.3 Heterogeneity

Having established that our results are not sensitive to changes in the specification or sample we next examine heterogeneity in the effects of exposure based on two different criteria: (i) defendant baseline socioeconomic and demographic characteristics, (ii) case characteristics. The first criterion is motivated by the observation that the effect of criminal justice interactions often vary based on labor market attachment and demographics; e.g. Dobbie et al. (2018), Garin et al. (2023), Jordan et al. (2023), Rose (2021), and Doleac and Hansen (2020). The second criterion is motivated by providing additional insights about the role of stigma.

Defendant characteristics

Table 2 presents results from a variation of model (1) where we interact exposed*post with baseline defendant characteristics measured in the year before the initial trial. In this exercise we use indicators for male, above age 25 (the median age in our sample), immigrant and whether the defendant holds a high school (HS) degree. Each panel presents the results for a different dependent variable. Column (1) recapitulates our baseline estimates from Table 1.

³⁶This analysis is potentially also affected by general deterrence effects if the existence of the database reduced the likelihood of engaging in crime in the general population.

³⁷As an alternative to model (1) we also estimated a regression discontinuity in time design (RDiT) where the running variable was time to the court-specific cutoff using the same dependent variables observed after the introduction of the platform. The estimates were, however, very imprecise to inform about the effect of interest. While a 95 percent confidence interval overlapped with our baseline estimates, the interval was not informative about the causal effect of interest. For instance, the exposure effects including 20 percent reductions in earnings could not be ruled out with 95 percent confidence.

In column (2), we find that the average exposure effect on convictions masks significant differences by gender. While the effect of exposure is close to zero for males, exposure increases the risk of conviction for females by 1.2 percentage points. We also find significant stronger increases in the probability of conviction for defendants above age 25, and for those with at least a HS degree. Turning to labor market outcomes in Panels B and C we find significantly stronger adverse effects on both employment and earnings for males, defendants above age 25 and those holding a HS degree. We also find that the effect on earnings is significantly weaker for migrants compared to natives -3.6 and -11.4 percent, respectively. In summary, we find stronger adverse exposure effects in groups where the opportunity cost of engaging in crime is higher.

Case characteristics

Table 3 presents estimates from regressions where we investigate potential heterogeneous exposure effects by case characteristics. Such effects are interesting in themselves as they may be informative about what type of information is relevant to include in criminal background checks. They potentially also contribute to shed light on the role of informal sanction costs. The labeling theory is one of the cornerstone models in criminology and posits that revealing one's criminal background can exacerbate criminal behavior. In this model, interactions with the criminal justice system labels the offender and cause him to adopt stigmatizing beliefs about himself and adopt a deviant identity (e.g. Becker, 1968; Lemert, 1967). Because of the severe difficulties involved in distinguishing informal sanction costs attached to the stigma of holding a criminal record from the imposed formal ones such as fines, community service or prison, we are not aware of any study where the empirical strategy by design is able to identify informal sanctions net of formal ones (Chalfin and McCrary, 2017).³⁸ Isolating the role of informal sanction costs is important as these mechanisms may operate in different direction and vary depending on the context. For instance, in places where the spatial concentration of criminals is higher, the role of stigma could be reduced.³⁹

The fact that all individuals in our sample are charged with a crime but public exposure of criminal records is quasi-randomly determined presents a rare opportunity to identify the impact of informal sanctions independently of formal ones. While this setting is sufficient to identify the role of informal

³⁸Empirical studies in criminology tend to compare offenders either convicted or incarcerated with those not formally sanctioned e.g. Bernburg and Krohn (2003), Farrington et al. (1978), and Murray et al. (2017).

³⁹It can be useful to consider the (dis)utility of the offender, U, as a function of formal sanctions, F, and informal sanctions, I, so that U = f(F,I). If informal sanctions lead to worse utility $\partial U/\partial I < 0$, while formal sanctions improve utility $\partial U/\partial F > 0$, for instance if prisons provide adequate rehabilitation programs, then the reduced form effect depends on the relative magnitudes of these partial derivatives and their interaction within the utility function.

Panel A: Convicted Exposed x Post (1) (2) (3) (4) (5) Exposed x Post 0.003 0.012*** -0.000 0.003 -0.002 x Male -0.011*** (0.003) (0.004) (0.003) (0.004) x Male -0.011*** (0.003) (0.003) (0.004) x Above age 25 0.007** (0.003) (0.003) x Immigrant - - -0.001 x HS degree 0.088 0.088 0.088 0.008 Sample mean 0.088 0.088 0.088 0.088 0.008 Panel B: Employed (0.003) (0.007) (0.008) (0.007) (0.008) x Male -0.017* (0.007) (0.008) (0.007) (0.008) x Immigrant -0.666 0.666 0.666 0.666 0.666 Sample mean 0.666 0.666 0.666 0.666 0.666 Panel C: Log Earnings -0.010* -0.017* -0.010* -0.010*	<i>v</i>					
Exposed x Post 0.003 0.012^{***} -0.000 0.003 -0.002 x Male -0.011^{***} (0.003) (0.003) (0.003) (0.004) x Above age 25 0.007^{**} (0.003) -0.001 x Immigrant -0.001 (0.003) x HS degree 0.088 0.088 0.088 Panel B: Employed -0.0017^* (0.003) Exposed x Post -0.008 0.005 0.016^{**} -0.010 (0.008) (0.010) (0.007) (0.007) (0.007) x Male -0.017^* (0.007) (0.007) (0.007) x Male -0.017^* (0.006) (0.007) (0.007) x Male -0.017^* (0.006) (0.007) (0.008) x Immigrant 0.066 0.666 0.666 0.666 0.666 Panel C: Log Earnings (0.042) (0.064) (0.047) (0.042) x Male -0.093^{**} -0.010 (0.047) (0.042) x Male -0.093^{**}	Panel A: Convicted	(1)	(2)	(3)	(4)	(5)
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	Exposed x Post	0.003	0.012***	-0.000	0.003	-0.002
x Male -0.011*** 0.007** x Above age 25 0.007** (0.003) x Immigrant -0.001 (0.003) x Immigrant -0.008 0.088 0.088 0.088 x HS degree -0.008 0.088 0.088 0.088 0.088 Panel B: Employed -0.008 0.005 0.016** -0.010 0.006 x Male -0.017* (0.007) (0.007) (0.007) x Male -0.017* (0.007) (0.007) x Male -0.017* (0.007) (0.007) x Male -0.017* (0.007) (0.008) x Immigrant -0.066 0.666 0.666 Sample mean 0.666 0.666 0.666 Panel C: Log Earnings -0.010 (0.042) (0.044) x Male -0.093** -0.010 (0.047) (0.042) x Male -0.093** -0.010 (0.047) (0.042) x Male -0.010* (0.046) (0.047) (0.042) x Male -0.010* (0.046) (0.047) <td< td=""><td></td><td>(0.003)</td><td>(0.003)</td><td>(0.004)</td><td>(0.003)</td><td>(0.004)</td></td<>		(0.003)	(0.003)	(0.004)	(0.003)	(0.004)
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	x Male		-0.011***	. ,	. ,	
x Above age 25 0.007^{**} (0.003) x Immigrant -0.001 (0.003) x HS degree 0.007^{**} (0.003) Sample mean 0.088 0.088 0.088 Panel B: Employed Exposed x Post -0.008 0.005 0.016^{**} 			(0.003)			
x Immigrant .0.003) x HS degree .0.007*** Sample mean 0.088 0.088 0.088 0.088 Panel B: Employed	x Above age 25			0.007**		
x Inmigrant -0.001 x HS degree 0.007** Sample mean 0.088 0.088 0.088 0.088 Panel B: Employed -0.008 0.005 0.016** -0.010 0.006 Exposed x Post -0.008 0.005 0.016** -0.010 0.006 (0.008) (0.010) (0.007) (0.008) (0.007) x Male -0.017* (0.007) (0.008) (0.007) x Above age 25 -0.047*** (0.007) -0.027*** x Immigrant 0.066 0.666 0.666 0.666 x HS degree -0.027*** (0.007) (0.008) x HS degree -0.093** -0.010 (0.027) -0.114** -0.010 x Male -0.093** -0.010 (0.047) (0.042) (0.046) (0.047) (0.042) x Male -0.107* (0.046) (0.046) (0.047) (0.042) x Male -0.107* (0.048) (0.048) (0.048) (0.048) x Immigrant -0.237*** (0.048) (0.048) (0.048)	e			(0.003)		
x HS degree (0.003) Sample mean 0.088 0.088 0.088 0.088 Panel B: Employed -0.008 0.005 0.016** -0.010 0.006 Exposed x Post -0.008 0.005 0.016** -0.010 0.006 x Male -0.017* (0.009) (0.007) (0.008) (0.007) x Male -0.017* (0.006) - - - x Immigrant -0.047*** (0.007) - - - x Immigrant 0.666 0.666 0.666 0.666 0.666 Panel C: Log Earnings -0.093** -0.010 (0.047) (0.042) x Male -0.107* (0.046) (0.047) (0.042) x Male -0.093** -0.010 (0.047) (0.042) x Male -0.107* (0.048) -0.107 (0.048) x Above age 25 -0.237*** (0.046) -0.167*** (0.048) x Male -0.107* (0.048) -0.167*** (0.048) x Immigrant 0.078* (0.048) -0.167**	x Immigrant				-0.001	
x HS degree 0.007** (0.003) Sample mean 0.088 0.088 0.088 0.088 Panel B: Employed -0.008 0.005 0.016** -0.010 0.006 Exposed x Post -0.008 0.005 0.016** -0.010 0.006 x Male -0.017* (0.009) (0.007) (0.008) (0.007) x Male -0.017* (0.007) (0.007) (0.007) x Immigrant 0.006 0.007 (0.007) x HS degree -0.027*** (0.008) Sample mean 0.666 0.666 0.666 0.666 Panel C: Log Earnings -0.027 -0.114** -0.010 (0.042) (0.064) (0.046) (0.047) (0.042) x Male -0.107* (0.046) (0.047) (0.042) x Above age 25 -0.237*** (0.046) -0.167** (0.048) x Immigrant (0.069) -0.237*** (0.046) -0.167*** x Male -0.107* (0.048) -0.167*** (0.048) x HS degree -0.167***<	8				(0.003)	
Sample mean 0.088 0.088 0.088 0.088 0.088 Panel B: Employed -0.008 0.005 0.016*** -0.010 0.006 x Male -0.017* (0.009) (0.007) (0.008) (0.007) x Male -0.017* (0.009) (0.007) (0.008) (0.007) x Male -0.017* (0.006) -0.047*** (0.007) x Male -0.017* (0.007) (0.007) x HS degree -0.027**** (0.007) x HS degree -0.027*** (0.008) Sample mean 0.666 0.666 0.666 0.666 Panel C: Log Earnings -0.093** -0.010 (0.027) -0.114*** -0.010 x Male -0.093** -0.010 (0.042) (0.046) (0.042) x Male -0.107* - - - - x Male -0.107* - - - - x Male -0.0237*** (0.048) - -	x HS degree				()	0.007**
Sample mean 0.088 0.088 0.088 0.088 0.088 Panel B: Employed -0.008 0.005 0.016^{**} -0.010 0.006 x Male -0.017^* (0.009) (0.007) (0.008) (0.007) x Male -0.017^* (0.009) -0.047^{***} (0.007) (0.008) x Above age 25 -0.047^{***} (0.007) -0.027^{***} (0.007) x HS degree -0.0666 0.666 0.666 0.666 0.666 0.666 0.007 Sample mean 0.666 0.666 0.666 0.666 0.666 0.666 Panel C: Log Earnings -0.093^{**} -0.010 0.027 -0.114^{**} -0.010 x Male -0.093^{**} -0.010^* (0.042) (0.048) (0.042) x Male -0.107^* (0.048) (0.048) (0.048) (0.048) x Immigrant (0.060) (0.048) (0.048) (0.048) (0.048) <td></td> <td></td> <td></td> <td></td> <td></td> <td>(0.003)</td>						(0.003)
Panel B: Employed Exposed x Post -0.008 0.005 0.016** -0.010 0.006 (0.008) (0.010) (0.007) (0.008) (0.007) x Male -0.017* (0.009) (0.006) (0.007) x Above age 25 -0.047*** (0.006) (0.007) x Immigrant 0.007 (0.007) (0.007) x HS degree -0.027*** (0.008) Sample mean 0.666 0.666 0.666 0.666 Panel C: Log Earnings -0.010* (0.042) (0.064) (0.046) (0.047) x Male -0.107* (0.048) -0.017* (0.042) (0.048) x Male -0.107* (0.048) -0.167*** (0.048) x Male -0.107* (0.048) -0.167*** (0.048) x Immigrant 0.078* (0.048) -0.167*** (0.048) x Immigrant -0.167*** (0.048) -0.167*** (0.048) -0.167*** x HS degree -0.167*** (0.048) -0.167*** (0.048) -0.167*** <t< td=""><td>Sample mean</td><td>0.088</td><td>0.088</td><td>0.088</td><td>0.088</td><td>0.088</td></t<>	Sample mean	0.088	0.088	0.088	0.088	0.088
Panel B: Employed -0.008 0.005 0.016** -0.010 0.006 kx posed x Post -0.008 (0.001) (0.007) (0.008) (0.007) x Male -0.017* (0.009) (0.007) (0.008) (0.007) x Above age 25 -0.047*** (0.006) - - - x Immigrant - - 0.007 (0.007) (0.008) x HS degree - - - - - - - - - 0.027*** (0.008) Sample mean 0.666 0.666 0.666 0.666 0.666 0.666 0.666 0.666 0.047) (0.042) - - - - - 0.027*** (0.008) 3 - - - - - - - - - - - - - - - 0.010 (0.047) (0.042) - - - - - - - <td< td=""><td></td><td></td><td></td><td></td><td></td><td></td></td<>						
Exposed x Post-0.008 0.005 0.016^{-4} -0.010 0.006 (0.008)(0.001)(0.007)(0.008)(0.007)x Male -0.017^* (0.009)(0.007)x Above age 25 -0.047^{***} (0.006)x Immigrant -0.047^{***} (0.007)x HS degree -0.027^{***} Sample mean 0.666 0.666 Panel C: Log Earnings -0.093^{**} -0.010 Exposed x Post -0.093^{**} -0.010 (0.042) (0.064) (0.046) (0.042) (0.064) (0.046) (0.042) (0.064) (0.046) x Male -0.107^* (0.048) -0.107^* x Immigrant (0.078^*) (0.048) -0.167^{***} (0.048) -0.167^{***} (0.048) -0.167^{***} (0.048) 5.835 5.835 Sample mean <t< td=""><td>Panel B: Employed</td><td>0.000</td><td>0.005</td><td>0.01/**</td><td>0.010</td><td>0.000</td></t<>	Panel B: Employed	0.000	0.005	0.01/**	0.010	0.000
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	Exposed x Post	-0.008	0.005	0.016**	-0.010	0.006
x Male -0.017^{*} (0.009)x Above age 25 -0.047^{***} (0.006)x Immigrant 0.007 (0.007)x HS degree -0.027^{***} (0.008)Sample mean 0.666 0.666 0.666 Panel C: Log Earnings Exposed x Post -0.093^{**} (0.042)Male -0.093^{**} (0.042)x Male -0.107^{*} (0.060)x Male -0.107^{*} (0.042)x Male -0.107^{***} (0.048)x Immigrant 0.078^{*} (0.048)x Immigrant 0.078^{*} (0.048)x HS degree -0.167^{***} (0.048)x HS degree -0.167^{***} (0.048)Year FEYesYear SYesYear SYes	26.1	(0.008)	(0.010)	(0.007)	(0.008)	(0.007)
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	x Male		-0.017*			
x Above age 25 $-0.04/^{***}$ x Immigrant (0.006) x Immigrant 0.007 x HS degree -0.027^{***} Sample mean 0.666 0.666 0.666 Panel C: Log Earnings -0.093^{**} -0.010 0.027 -0.114^{***} Exposed x Post -0.093^{**} -0.010 0.027 -0.114^{***} -0.010 x Male -0.107^* (0.046) (0.047) (0.042) x Male -0.107^* (0.048) -0.078^* x Immigrant 0.078^* (0.048) x HS degree -0.167^{***} (0.048) Sample mean 5.835 5.835 5.835 5.835 Year FE Yes Yes Y			(0.009)			
x Immigrant 0.006 x HS degree -0.027^{***} (0.008) Sample mean 0.666 0.666 0.666 0.666 0.666 Panel C: Log Earnings Exposed x Post -0.093^{**} -0.010 0.027 -0.114^{**} -0.010 (0.042) (0.064) (0.046) (0.047) (0.042) x Male -0.107^{*} (0.060) x Above age 25 -0.237^{***} (0.048) x Immigrant 0.078^{*} (0.048) x Immigrant 0.078^{*} (0.048) x HS degree -0.167^{***} (0.048) X HS degree -0.167^{**} (0.048) X HS degree -0.167^{**} (0.048) X HS degree -0.167^{**} (0.048) X	x Above age 25			-0.047***		
x Immigrant 0.007 x HS degree -0.027^{***} x HS degree 0.666 Sample mean 0.666 0.666 0.666 Panel C: Log Earnings 0.007 -0.114^{**} -0.003 Exposed x Post -0.093^{**} -0.010 0.027 -0.114^{**} -0.010 x Male -0.007^{***} (0.042) (0.064) (0.046) (0.047) (0.042) x Male -0.107^{*} (0.048) -0.237^{***} (0.048) x Immigrant 0.078^{*} (0.048) -0.167^{***} x HS degree -0.167^{***} (0.048) -0.167^{***}	* •			(0.006)	0.007	
x HS degree-0.027*** (0.008)Sample mean0.6660.6660.6660.666Panel C: Log Earnings Exposed x Post-0.093** (0.042)-0.010 (0.042)0.027 (0.046)-0.114** (0.046)-0.010 (0.047)x Male-0.107* (0.060)-0.237*** (0.048)-0.078* (0.048)-0.078* (0.048)x Immigrant0.078* (0.048)-0.167*** (0.048)-0.167*** (0.048)x HS degree-0.167*** (0.048)-0.167*** (0.048)Sample mean5.8355.8355.835Year FEYesYesYesYesYear FEYesYesYesYesSingle Charge FEYacYacYacYac	x Immigrant				0.007	
x HS degree-0.027*** (0.008)Sample mean0.6660.6660.6660.666Panel C: Log EarningsExposed x Post-0.093**-0.0100.027-0.114**-0.010(0.042)(0.064)(0.046)(0.047)(0.042)x Male-0.107*(0.060)x Above age 25 -0.237^{***} (0.048)x Immigrant0.078*(0.046)x HS degree -0.167^{***} (0.046)x HS degree -0.167^{***} (0.048)Sample mean5.8355.8355.8355.835Year FEYesYesYesYesYear FEYesYesYesYesSince Charge EEYesYesYesYesYear Since Charge EEYesYesYesYes					(0.007)	
Sample mean 0.666 0.666 0.666 0.666 0.666 Panel C: Log Earnings-0.093** -0.010 0.027 -0.114^{**} -0.010 Exposed x Post -0.093^{**} -0.010 (0.046) (0.047) (0.042) x Male -0.107^* (0.060) (0.046) (0.047) (0.042) x Above age 25 -0.237^{***} (0.048) -0.167^* x Immigrant -0.107^* (0.048) -0.167^{***} x HS degree -0.167^* (0.048) -0.167^{***} Sample mean 5.835 5.835 5.835 5.835 Year FEYesYesYesYesCourt FEYesYesYesYesYarYarYarYarYar	x HS degree					-0.027***
Sample mean 0.666 0.660 0.010 0.027 -0.114** -0.010 (0.042) x Male -0.010 (0.042) x 0.042) x 0.043) x 0.046) x 0.046) x 0.046) x 0.046) x 0.0						(0.008)
$\begin{array}{c c c c c c c } Panel C: Log Earnings \\ Exposed x Post & -0.093^{**} & -0.010 & 0.027 & -0.114^{**} & -0.010 \\ & & & & & & & & & & & & & & & & & & $	Sample mean	0.666	0.666	0.666	0.666	0.666
Exposed x Post -0.093** -0.010 0.027 -0.114** -0.010 x Male -0.107* (0.046) (0.047) (0.042) x Male -0.107* (0.060) -0.237*** -0.107* x Above age 25 -0.237*** (0.048) -0.167*** x Immigrant 0.078* -0.167*** (0.048) x HS degree -0.167*** (0.048) -0.167*** Sample mean 5.835 5.835 5.835 5.835 Year FE Yes Yes Yes Yes Court FE Yes Yes Yes Yes Times Charge EE Yes Yes Yes Yes	Panel C: Log Earnings					
Image: constraint of the system of the sy	Exposed x Post	-0.093**	-0.010	0.027	-0.114**	-0.010
x Male -0.107* (0.060) x Above age 25 -0.237*** (0.048) x Immigrant 0.078* (0.046) x HS degree -0.167*** (0.046) x HS degree -0.167*** (0.048) Sample mean 5.835 5.835 5.835 5.835 5.835 Year FE Yes Yes Yes Yes Yes Yes Court FE Yes Yes Yes Yes Yes Yes	1	(0.042)	(0.064)	(0.046)	(0.047)	(0.042)
x Above age 25 -0.237*** x Immigrant (0.048) x Immigrant 0.078* x HS degree -0.167*** x HS degree -0.167*** Sample mean 5.835 5.835 5.835 Year FE Yes Yes Yes Court FE Yes Yes Yes Times Since Charge EE Yes Yes Yes	x Male	· · · ·	-0.107*		· /	
x Above age 25 x Immigrant x HS degree Sample mean 5.835 Year FE Court FE Yes Yes Yes Yes Yes Yes Yes Yes			(0.060)			
x Immigrant (0.048) x HS degree (0.048) x HS degree -0.167*** (0.048) (0.048) Sample mean 5.835 5.835 Sample mean 5.835 5.835 5.835 Year FE Yes Yes Yes Court FE Yes Yes Yes Yes Times Change EE Yes Yes Yes Yes	x Above age 25		()	-0.237***		
x Immigrant 0.078* (0.046) x HS degree -0.167*** (0.048) Sample mean 5.835 5.835 5.835 5.835 5.835 Year FE Yes Yes Yes Yes Yes Court FE Yes Yes Yes Yes Yes				(0.048)		
x HS degree (0.046) x HS degree -0.167*** (0.048) (0.048) Sample mean 5.835 5.835 5.835 Year FE Yes Yes Yes Yes Court FE Yes Yes Yes Yes Times Since Charge EE Yes Yes Yes Yes	x Immigrant			(01010)	0.078*	
x HS degree -0.167*** Sample mean 5.835 5.835 5.835 5.835 Year FE Yes Yes Yes Yes Court FE Yes Yes Yes Yes Times Since Change EE Yes Yes Yes Yes	n mingrand				(0.046)	
Sample mean5.8355.8355.8355.835(0.048)Year FEYesYesYesYesYesCourt FEYesYesYesYesYesTime Since Charge EEYesYesYesYesYes	x HS degree				(01010)	-0.167***
Sample mean5.8355.8355.8355.835Year FEYesYesYesYesCourt FEYesYesYesYesTime Since Charge FEYesYesYesYes						(0.048)
Year FE Yes Yes Yes Yes Court FE Yes Yes Yes Yes Time Since Change FE Yes Yes Yes	Sample mean	5.835	5.835	5.835	5.835	5.835
Court FE Yes Yes Yes Yes Time Since Charge FE Yes Yes Yes Yes	Year FE	Yes	Yes	Yes	Yes	Yes
Time Cinese Charge EE Veg Veg Veg Veg Veg	Court FE	Yes	Yes	Yes	Yes	Yes
Three Since Charge FE Tes Tes Tes Tes Tes Tes	Time Since Charge FE	Yes	Yes	Yes	Yes	Yes
Calendar Month FE Yes Yes Yes Yes Yes	Calendar Month FE	Yes	Yes	Yes	Yes	Yes
Baseline Controls Yes Yes Yes Yes Yes	Baseline Controls	Yes	Yes	Yes	Yes	Yes
	Observations	186.840	186.840	186.840	186.840	186.840
	Observations	186,840	186,840	186,840	186,840	186,840

 Table 2. Results by Demographic and Socioeconomic Characteristics

Note: This table reports difference-in-differences estimates based on model (1) of the effect of exposure on crime and labor market outcomes. The regressions control for age, gender, immigrant and the main effect of being exposed. Standard errors in parenthesis are clustered at the individual and court level. *** = significant at 1 percent level, ** = significant at 5 percent level, * = significant at 10 percent level.

Panel A: Convicted	(1)	(2)	(3)	(4)
Exposed x Post	0.003	0.002	0.001	0.002
	(0.003)	(0.003)	(0.004)	(0.003)
x Acquitted		0.007**		
		(0.004)		
x Misdemeanor			0.003	
			(0.003)	0.001
x Neighborhood offenders				0.001
< median	0.000	0.000	0.000	(0.003)
Sample mean	0.088	0.088	0.088	0.088
Panel B: Employed				
Exposed x Post	-0.008	-0.004	0.001	-0.001
Exposed x 1 ost	(0.008)	(0.004)	(0.001)	(0.001)
x Acquitted	(0.000)	-0.029**	(0.007)	(0.000)
A requited		(0.011)		
x Misdemeanor		(0.00-2)	-0.036***	
			(0.009)	
x Neighborhood offenders			× /	-0.017***
< median				(0.006)
Sample mean	0.666	0.666	0.666	0.666
Panel C: Log Earnings				
Exposed x Post	-0.093**	-0.065	-0.051	-0.054
	(0.042)	(0.046)	(0.043)	(0.044)
x Acquitted		-0.194**		
		(0.076)	0 4 4 4 * *	
x Misdemeanor			-0.141**	
NT . 1.1 1 1 . CC 1			(0.061)	0.000**
x Neighborhood offenders				-0.098
< median Sample mean	5 8 2 5	5 8 2 5	5 8 2 5	(0.039)
Sample mean	5.055	5.055	5.055	5.055
Year FE	Yes	Yes	Yes	Yes
Court FE	Yes	Yes	Yes	Yes
Baseline Controls	Yes	Yes	Yes	Yes
Observations	186,840	186,840	186,840	186,840

 Table 3. Case characteristics and labeling

Note: This table reports difference-in-differences estimates based on model (1) of the effect of exposure on crime and labor market outcomes. The regressions control for the same variables as in Table 2. Numbers in brackets show mean of the dependent variable. Standard errors in parenthesis are clustered at the individual and court level. *** = significant at 1 percent level, ** = significant at 5 percent level, * = significant at 10 percent level.

sanctions, an analysis of whether the exposure effect varies by case characteristics provides an additional way to investigate the importance of stigma.

The results in column (2) of Panel A show that the exposure effects on recidivism are significantly higher for defendants who were acquitted in court compared to defendants who were convicted. The estimates suggest that the probability of being convicted increases by almost 1 percentage point for those acquitted.

Panels B and C show that there is also a significantly larger labor market penalty for acquitted defendants compared to convicted defendants. We can also see that the labor market penalty of exposure is significantly higher for defendants who were convicted of a misdemeanor compared to those convicted with a non-misdemeanor. Many misdemeanor offenses do not involve violence or firearms; rather, they stem from the criminalization of everyday behaviors like possessing small amounts of banned substances for personal use, vandalism, disturbing the peace, minor theft, and trespassing.⁴⁰

Recall that the open access version of the database did not differentiate between severe crimes and misdemeanors. Additionally, it included acquitted defendants without indicating their not-guilty verdicts. Therefore, one possible interpretation for the stronger treatment effects is that the platform's information may have signaled that an individual was a more prolific criminal than they actually were.

Several theoretical models predict that, all else equal, the degree of stigma in a given area is declining in the number of individuals with criminal records (e.g. Chalfin and McCrary, 2017; Weibull and Villa, 2005; Funk, 2004; Rose, 2021). To examine this we used our data to calculate the number of individuals convicted of a crime in the past five years living in each municipality. We then interacted *exposed* x *post* in model (1) with an indicator for the defendant living in an area with less than median criminals. The results show that the adverse labor market effects indeed are concentrated in the subsample of individuals who lives in neighborhoods with a low share of the population having a criminal conviction. For instance, while the decrease in earnings is 15.2 percent for those living in places with relatively few ex-convicts, it is 5.4 percent for defendants in neighborhoods with a higher concentration of ex-convicts.

5.4 Interpretation and additional results

Our main findings so far is that the massive use of the database is connected to significant adverse exposure effects on earnings but not on employment or criminal recidivism. However, there are significantly stronger detrimental

⁴⁰Misdemeanors are minor offenses, usually resulting in incarceration of one year or less. Felonies are more serious, such as robbery, theft over USD 1,000, and assault with a deadly weapon, carrying harsher penalties. While the grouping of offenses into misdemeanor and felonies is not used in Sweden, we classify misdemeanors as crimes for which no individual was sentenced to prison in our data.

effects on both labor market outcomes and recidivism in specific defendant subgroups such as those with at least a high school degree, acquitted individuals, and those living in areas with few criminals. These results are consistent with stigma attached to being marked as a criminal playing a key role in explaining the effects of interactions with the criminal justice system. In this section we provide results from additional analyses that aims to assess various explanations for our findings.

Dynamic scarring effects

Several studies show that criminal conviction is associated with a large drop in labor market outcomes (Rose, 2021). In Section 5.1 we showed that this is also the case in Sweden. If a conviction causes individuals to quickly accumulate labor market scars, such as lengthy employment gaps, employers may remain hesitant to hire these individuals, potentially making it more difficult to identify exposure effects (Agan et al., 2024). However, we do find a significant labor market penalty even net of this dynamic scaring effect. We also estimated our event study model (model (2)) focusing on the effect on labor market outcomes around the time of acquittal and misdemeanor charges. While imprecise, the point estimates are much closer to zero compared to that of convictions.⁴¹ Since we find significantly stronger exposure effects for acquittal and misdemeanor cases, it means that for those results, any dynamic scaring effects are likely to be small. The results in Panel E in Appendix Table A2 where we examine the immediate effect of the introduction of the platform using confession cases as an alternative control group finding similar results as in baseline also seem to rule out that dynamic scarring effects matters, at least in the context of this paper.

Behavioral responses

To what extent can our results be explained by behavioral adjustments taken by the defendants listed in the database? We examine this by estimating exposure effects for several potentially important behavioral markers. Previous studies have shown that criminal justice interactions has meaningful effects on earnings through changes in family structure (Charles and Luoh, 2010; Chetty et al., 2020; Svaver, 2011) and human capital investments (Finlay et al., 2023). Table A3 shows the results for exposure effects on the probability of enrolling in college and the likelihood of starting a labor market training program. These outcomes could proxy for human capital investments that defendant may potentially take in order to avoid the potential harmful effects of having ones criminal records exposed. We also examine the probability of moving to a different neighborhood. Finally, we present estimates for family formation outcomes. With the exception for the probability of enrolling in college, we find no significant exposure effects. The results for college en-

⁴¹The results are available on request.

rollment should also be interpreted with some caution since the effect size is small (0.4 percentage points) and only significant at the 10 percent level. One potential reason for why individuals not seem to respond to exposure may be that lack of knowledge or perceived low rewards. It also provide suggestive evidence that the reaction of employers to exposure is what explains the labor market penalty.

Industry sorting

Individuals could also respond to exposure by sorting into jobs that are more lenient towards hiring individuals with criminal records. To examine this we follow Mueller-Smith and Schnepel, 2021 and classify each two digit SNI (NAICS) industrial sector as having low or high employment penetration among convicted individuals using the regression adjusted ratios of the sector-specific employment rate for those individuals with convictions versus those without in our analysis samples (see Table A4). Sectors with ratios of 1.5 or less are high penetration industries relatively more accessible to those with criminal histories and the opposite is true for low penetration industries with ratios greater than 1.5. The results in Table A5 show small and imprecise estimates but no strong evidence in favor of differential sorting across industries for any of the groups. For sectors with employment penetration ratios of 1.2 or higher we find some evidence that individuals are sorting away from high penetration industries but the estimates are small in magnitude and imprecise.

AKM sorting

Job ladder models propose that workers engage in on-the-job search throughout their careers, gradually moving to higher-paying employers. In these models, pay disparities among firms reflect differences in how the surplus is divided between workers and employers (Burdett and Mortensen, 1998; Haltiwanger et al., 2018). Within this framework, exposed workers may experience earnings losses due to employer effects as they move down the job ladder to employers that offer lower pay. We examine to what extent the exposure effects can be explained by differential sorting of workers to firms that vary in their estimated earnings premia by following (Lachowska et al., 2020). We estimate the earnings premia in different firms using the methods suggested by Abowd et al. (1999) (AKM). In this exercise we define a firm as an establishment.

The AKM model can be written

$$logY_{itj} = \alpha_i + \beta_{i(i,t)} + \phi_t + X_{itj} + \varepsilon_{itj}$$
(3)

where $logY_{itj}$ denotes earnings of worker i with employer j in year t; α_i is a worker-specific fixed effect; $\beta_{j(i,t)}$ is an employer-specific fixed effect; ϕ_t is a vector of calendar year indicators; X_{itj} consist of education-gender-specific age dummies (following Engbom et al., 2023), and ε_{itj} is the error component. The function j(i,t) indexes the employer j effect for worker i in year t. Due to the problem of multicollinearity between age and year-fixed effects, we assume that the age effects are constant between ages 45 and 54.

This model controls for sorting of workers based on fixed differences in their underlying characteristics. The firm fixed effects capture firm characteristics that result in above- or below-average earnings for all workers at employer j and can be interpreted as representing the advantages derived from being employed by a given employer. We estimate the AKM model for firms in the connected set over the years 2005-2016, including workers aged 18-65. We exclude establishments with fewer than 5 movers when estimating the model but note that our results are robust to also including all establishments in the connected set.

We use the estimated $\hat{\beta}_{j(i,t)}$ as outcomes when we estimate exposure effects using model (1). The results are presented in Appendix Table A6. In general, we find significantly stronger adverse exposure effects on the earnings premia in the subgroups of defendants where the labor market responses were concentrated. This result is consistent with the predictions in job ladder models that exposed workers will experience earnings losses due to employer effects, as they moves down the job ladder to a lower-paying employer.

In order to better understand whether a move down the job ladder is due to workers changing employer or workers staying in the same firm we estimated model (1) using as dependent variable an indicator set to one if the individual has changed firm from one year to another and zero if (s)he stayed in the firm. Individuals who are non-employed are missing in the data. Overall, the results shown in Appendix Table A7 shows that workers at least age 25 are significantly more likely to change firm compared to younger workers, that acquitted workers are significantly more likely to change firm relative to convicted workers, and that workers charged with a misdemeanor are significantly more likely to move to a different firm relative to those charged with a more severe crime.

Taken together, it seems as if at least part of the reduction in the AKM earnings premium is due to workers transiting to lower-paying employers. While we cannot distinguish between voluntary and involuntary job losses, it is possible that exposed workers are fired or do not get their temporary contracts renewed. These workers may be forced to adjust their reservation wage downwards and in the process moving down the job ladder to a lower-paying employer.

Effects on children

In this section we investigate whether the introduction of the database affected the outcomes of the children of exposed defendants. The potential for significant yet previously overlooked spillover effects of parental criminal justice involvement is suggested by recent studies estimating the causal effect of parental incarceration (Arteaga, 2022; Bhuller et al., 2020; Dobbie et al., 2018; Norris et al., 2021), but it is unclear how children are affected by increased public access to criminal records of their parents.⁴² In this analysis, we focus on children who finishes grade 9 (typically at age 16) in the period 2010-2016 and link them to their parents. We also add information on court convictions for these children. We start by verifying in Appendix Table A8 that the effects in the parental subsample are relatively similar to that in the full population, although the precision of the point estimates is lower for log earnings and higher for employment. Table A9 shows the difference-in-differences estimates from model (1) of the exposure effects on children. We find significantly stronger adverse exposure effects on compulsory school GPA for children belonging to the subgroups of defendants with the largest adverse exposure effect on children could be driven by both a direct effect of the children experiencing the stigma from having a parent in the database or by a reduction in the family's resources.

6 Concluding Remarks

We study the unexpected launch of Sweden's first online criminal database in 2014 that provided unrestricted, anonymous, and cost-free public access to the criminal records of every individual charged with a crime. To estimate the causal effect of public access to this database, we leverage rich register data and the fact that administrative rules only allowed the company behind the database to identify offenders sentenced before certain dates in the past.

We find significant adverse effects of exposure on earnings but not on employment or criminal recidivism. However, there are significantly stronger detrimental effects on both labor market outcomes and recidivism in defendant subgroups with a higher opportunity cost of engaging in criminal activity, acquitted individuals, those charged with a misdemeanor and those living in areas with fewer criminals. Our results highlight that the stigma associated with being marked as a criminal is a potentially important but previously unappreciated mechanism explaining behavioral responses to criminal justice interactions.

⁴²While the findings are mixed, many of the estimates lack precision, resulting in overlapping confidence intervals for comparable outcomes. Using a random-judge design, Dobbie et al. (2018) report that parental incarceration leads to increased teen crime, lower school performance, and negative impacts on employment and earnings. Consistent with this, Grenet et al. (2024) show quasi-experimental evidence that a reform were parents were able to substitute a prison sentence to instead serve the full sentence using electronic monitoring significantly improved child outcomes. Conversely, Norris et al. (2021) find that in the United States, parental incarceration reduces teen crime, does not affect teen parenthood, and increases the likelihood of children living in affluent neighborhoods as adults. Arteaga (2022) observes positive effects on children's educational attainment in Colombia. Bhuller et al. (2020) find no significant effects on school performance or children's propensity for criminal behavior.

While we are unable to directly measure the long-run impact of open access criminal databases on defendants using our data, we place the magnitude of our estimates in perspective by considering a partial back-of-the-envelope calculation that accounts for the direct economic impacts on the defendant himself. To calculate the implied effect we make the conservative assumption that the treatment effect on log earnings is constant for a period of three years then completely fades away. We multiply the treatment effect with baseline average earnings (SEK 148,134) and estimate the three year earnings loss for the defendants in our sample to be SEK 44,436. Since 2014, about 47,000 first time defendants are added to the database every year. For instance, in 2018 48,183 first time defendants were convicted in court and consequently were listed in the database.⁴³ The combined earnings losses this year thus amounts to SEK 2.14 billion, or 23 % of total government spending on prison and rehabilitation that year.⁴⁴ It is, however, important to note that we are unable to determine the full welfare consequences of open access to online criminal records using our research design. It is possible that these databases have a general deterrence effect in the full population. Our analysis will, therefore, overstate the social costs of these databases if the threat of being included in the database decreases crime rates. The existing literature has struggled to provide definitive explanations for why the impact of stricter sentencing policies on offenders' outcomes appears to vary across different study contexts. Our findings potentially offer a partial explanation. In environments where stigma carries less weight, such as places with a high spatial concentration of criminals, informal sanctions may play a diminished role in explaining the reduced form effect of sanction policy relative to that of formal sanctions. On the other hand, in contexts where informal sanction costs matter more, the beneficial effects of more prison staff or rehabilitation programs could be offset. Our findings directly speak to the debate over what type of information on cases should be disclosed in criminal background checks. Our results suggest that reforms that increase leniency in the conviction dimension (e.g. higher evidentiary standards or not prosecuting marginal cases) potentially has social benefits. In many countries, the information documented in background checks has vastly expanded along with their public availability so that many record checks now include non-criminal details that presumptively mark individuals as risky (Zedner and Ashworth, 2019). However, hardly any research has investigated the consequences of shifts in the content and governance of these records.

⁴³https://bra.se/statistik/kriminalstatistik/personer-lagforda-for-brott.html

⁴⁴Total government spending on prison and rehabilitation was SEK 9.3 billion in 2018. https://www.kriminalvarden.se/globalassets/publikationer/ekonomi/kriminalvardensarsredovisning-2018.pdf

References

- Abowd, J., F. Kramarz, and N. Margolis (1999). "High wage workers and high wage firms". *Econometrica* 67, 251–333.
- Abrams, David S (2012). "Estimating the deterrent effect of incarceration using sentencing enhancements". *American Economic Journal: Applied Economics* 4.4, 32–56.
- Agan, A., J. Doleac, and A. Harvey (2023). "Misdemeanor prosecution". *Quarterly Journal of Economics* 138.3, 1453–1505.
- Agan, Amanda, Andrew Garin, Dimitri Koustas, Alexandre Max, and Crystal Yang (2024). "The Labor Market Impacts of Reducing Felony Convictions". *American Economic Review: Insights*.
- Agan, Amanda and Sonja Starr (2018). "Ban the box, criminal records, and racial discrimination: A field experiment". *The Quarterly Journal of Economics* 133.1, 191–235.
- Aizer, Anna and Joseph J Doyle Jr (2015). "Juvenile incarceration, human capital, and future crime: Evidence from randomly assigned judges". *The Quarterly Journal of Economics* 130.2, 759–803.
- Anker, Anne-Sofie, Jennifer Doleac, and Rasmus Landersø (2021). "The Effects of DNA Databases on the Deterrence and Detection of Offenders". *American Economic Journal: Applied Economics* 13.4, 194–225.
- Arteaga, C. (2022). "Parental Incarceration and Children's Educational Attainment". *Review of Economics and Statistics*.
- Backman, Cristel (2012). "Criminal Records in Sweden: Regulation of Access to Criminal Records and the Use of Criminal Background Checks by Employers".
- Barbarino, Alessandro and Giovanni Mastrobuoni (2014). "The incapacitation effect of incarceration: Evidence from several Italian collective pardons". *American Economic Journal: Economic Policy* 6.1, 1–37.
- Becker, Gary S (1968). "Crime and punishment: An economic approach". *Journal of political economy* 76.2, 169–217.
- Bernburg, Jön Gunnar and Marvin D Krohn (2003). "Labeling, life chances, and adult crime: The direct and indirect effects of official intervention in adolescence on crime in early adulthood". *Criminology* 41.4, 1287–1318.
- Bhuller, Manudeep, Gordon B Dahl, Katrine V Løken, and Magne Mogstad (2020). "Incarceration, recidivism, and employment". *Journal of Political Economy* 128.4, 1269–1324.
- Buonanno, Paolo and Steven Raphael (2013). "Incarceration and incapacitation: Evidence from the 2006 Italian collective pardon". American Economic Review 103.6, 2437–2465.

- Burdett, K. and D. Mortensen (1998). "Wage Differentials, Employer Size, and Unemployment". *International Economic Review* 39.2, 257–273.
- Bushway, S., I. Cabreros, J. Welburn Paige, D. Schwam, and J. Wenger (2022). "Barred from employment: More than half of unemployed men in their 30s had a criminal history of arrest". *Science Advances* 8.7.
- Bushway, Shawn D and Gary Sweeten (2007). "Abolish lifetime bans for exfelons". *Criminology & Pub. Pol'y* 6, 697.
- Chalfin, A. and J. McCrary (2017). "Criminal Deterrence: A Review of the Literature". *Journal of Economic Literature* 55.1, 5–48.
- Charles, K. and M. Luoh (2010). "Male incarceration, the marriage market, and female outcomes". *The Review of Economics and Statistics* 92.3, 614–627.
- Chetty, R., N. Hendren, and S. Porter (2020). "Race and economic opportunity in the United States: An intergenerational perspective". *The Quarterly Journal of Economics* 135.2, 711–783.
- Cullen, Zoe, Will Dobbie, and Michael Hoffman (2024). "Increasing the Demand for Workers with a Criminal Record". *Quarterly Journal of Economics*.
- Dobbie, Will, Jacob Goldin, and Crystal S Yang (2018). "The effects of pretrial detention on conviction, future crime, and employment: Evidence from randomly assigned judges". *American Economic Review* 108.2, 201–240.
- Dobbie, Will, Hans Grönqvist, Susan Niknami, Mårten Palme, and Mikael Priks (2018). "The intergenerational effects of parental incarceration". Tech. rep. National Bureau of Economic Research.
- Doleac, Jennifer L and Benjamin Hansen (2020). "The unintended consequences of "ban the box": Statistical discrimination and employment outcomes when criminal histories are hidden". *Journal of Labor Economics* 38.2, 321–374.
- Drago, Francesco, Roberto Galbiati, and Pietro Vertova (2009). "The deterrent effects of prison: Evidence from a natural experiment". *Journal of political Economy* 117.2, 257–280.
- Engbom, N., C. Moser, and J. Sauerman (2023). "Firm pay dynamics". *Journal of Econometrics* 233, 396–423.
- Farrington, David P, Stan G Osborn, and Donald J West (1978). "The persistence of labelling effects". *Brit. J. Criminology* 18, 277.
- Finlay, Keith (2008). "Effect of employer access to criminal history data on the labor market outcomes of ex-offenders and non-offenders". Tech. rep. National Bureau of Economic Research.
- Finlay, Keith, Michael Mueller-Smith, and Brittany Street (2023). "Children's Indirect Exposure to the U.S. Justice System: Evidence From Longitudinal

Links between Survey and Administrative Data*". *The Quarterly Journal of Economics* 138.4, 2181–2224.

- Freeman, Richard B. (1992). "Crime and the Employment of Disadvantaged Youths". *Urban Labor Markets and Job Opportunity*. Washington DC: The Urban Institute.
- Funk, Patricia (2004). "On the Effective Use of Stigma as a Crime Deterrent". *European Economic Review* 48.4, 715–725.
- Garin, A., D. Koustas, C. McPherson, S. Norris, M. Pecenco, E. Rose, Y. Shem-Tov, and J. Weaver (2023). "The Impact of Incarceration on Employment, Earnings, and Tax Filing." University of Chicago, Becker Friedman Institute for Economics Working Paper.
- Grenet, Julien, Hans Grönqvist, and Susan Niknami (2024). "The Effects of Electronic Monitoring on Offendes and Their Families". *Journal of Public Economics*.
- Grogger, Jeffrey (1995). "The effect of arrests on the employment and earnings of young men". *The Quarterly Journal of Economics* 110.1, 51–71.
- Grönqvist, Hans, Julien Grenet, and Susan Niknami (2022). "The Effects of Electronic Monitoring on Offenders and Their Family Members". *Available at SSRN 4181510*.
- Haltiwanger, J., H. Hyatt, L. Kahn, and E. McEntarfer (2018). "Cyclical Job Ladders by Firm Size and Wages". *American Economic Journal: Macroeconomics* 10.2, 52–85.
- Helland, Eric and Alexander Tabarrok (2007). "Does three strikes deter? A nonparametric estimation". *Journal of human resources* 42.2, 309–330.
- Hjalmarsson, R. and M. Lindquist (2022). "The Health Effects of Prison". *American Economic Journal: Applied Economics* 14, 234–270.
- Holzer, Harry J, Steven Raphael, and Michael A Stoll (2006). "Perceived criminality, criminal background checks, and the racial hiring practices of employers". *The Journal of Law and Economics* 49.2, 451–480.
- Jackson, Osborne and Bo Zhao (2017). "The effect of changing employers' access to criminal histories on ex-offenders' labor market outcomes: evidence from the 2010–2012 Massachusetts CORI Reform".
- Jordan, A., E. Kager, and D. Neal (2023). "Heterogenous impacts of sentencing decisions". *NBER Working Paper*.
- Kahan, Dan M and Eric A Posner (1999). "Shaming white-collar criminals: A proposal for reform of the federal sentencing guidelines". *The Journal of Law and Economics* 42.S1, 365–392.
- Kamat, V., S. Norris, and M. Pecenoc (2024). "Conviction, Incarceration and Policy Effects in the Criminal Justice System". *SSRN Working Paper*.

- Kessler, Daniel and Steven D Levitt (1999). "Using sentence enhancements to distinguish between deterrence and incapacitation". *The Journal of Law and Economics* 42.S1, 343–364.
- Kling, Jeffrey R (2006). "Incarceration length, employment, and earnings". *American Economic Review* 96.3, 863–876.
- Kuziemko, Ilyana (2013). "How should inmates be released from prison? An assessment of parole versus fixed-sentence regimes". *The Quarterly Journal of Economics* 128.1, 371–424.
- Lachowska, Marta, Alexandre Mas, and Stephen A. Woodbury (Oct. 2020). "Sources of Displaced Workers' Long-Term Earnings Losses". *American Economic Review* 110.10, 3231–66.
- Lee, David S and Justin McCrary (2017). "The deterrence effect of prison: Dynamic theory and evidence". *Regression discontinuity designs*. Vol. 38. Emerald Publishing Limited, 73–146.
- Lemert, Edwin McCarthy (1967). *Human deviance, social problems, and social control.* Englewood Cliffs, NJ, Prentice-Hall.
- Mueller-Smith, Michael (2015). "The criminal and labor market impacts of incarceration". *Unpublished working paper* 18.
- Mueller-Smith, Michael and Kevin Schnepel (2021). "Diversion in the Criminal Justice System". *Review of Economic Studies* 88.2, 883–936.
- Mungan, Murat C (2016). "A generalized model for reputational sanctions and the (ir) relevance of the interactions between legal and reputational sanctions". *International Review of Law and Economics* 46, 86–92.
- Murray, Joseph, Arjan Blokland, David P Farrington, and Delphine Theobald (2017). "Long-term effects of conviction and incarceration on men in the Cambridge Study in Delinquent Development". *Labeling theory*. Routledge, 209–235.
- Nagin, Daniel S (2013). "Deterrence: A review of the evidence by a criminologist for economists". *Annu. Rev. Econ.* 5.1, 83–105.
- Nagin, Daniel S and Greg Pogarsky (2003). "An experimental investigation of deterrence: Cheating, self-serving bias, and impulsivity". *Criminology* 41.1, 167–194.
- Norris, S., M. Pecenco, and J. Weaver (2021). "The Effects of Parental and Sibling Incarceration: Evidence from Ohio". *American Economic Review* 111.9, 2926–2963.
- Ouss, Aurelie., John-Eric. Humphries, Kamelia. Stareva, Megan. Stevenson, and Winnie van Dijk (2023). "Conviction, Incarceration and Recidivism: Understanding the Revolving Door".

- Owens, Emily G (2009). "More time, less crime? Estimating the incapacitative effect of sentence enhancements". *The Journal of Law and Economics* 52.3, 551–579.
- Pager, Devah (2003). "The mark of a criminal record". *American journal of sociology* 108.5, 937–975.
- Polinsky, A Mitchell and Steven Shavell (2007). "The theory of public enforcement of law". *Handbook of law and economics* 1, 403–454.
- Prescott, J. and S. Starr (2020). "Expungement of Criminal Convictions: An Empirical Study". *Harvard Law Review* 133.8, 2460–2465.
- Raphael, S. (2014). "The New Scarlet Letter? Negotiating the US Labor Market with a Criminal Record". Tech. rep. W E Upjohn institute.
- Rasmusen, Eric (1996). "Stigma and self-fulfilling expectations of criminality". *The Journal of Law and Economics* 39.2, 519–543.
- Rose, E. and S-Y Yotam (2021). "How Does Incarceration Affect Reoffending? Estimating the Dose-Response Function". *Journal of Political Economy* 129.12, 3302–3356.
- Rose, Evan K (2021). "Does banning the box help ex-offenders get jobs? Evaluating the effects of a prominent example". *Journal of Labor Economics* 39.1, 79–113.
- Roth, J., P. Sant'Anna, A. Bilinski, and J. Poe (2023). "What's Trending in Difference-in-Differences? A Synthesis of the Recent Econometrics Literature". *Journal of Econometrics* 235.2, 2218–2244.
- Selbin, J., J. McCrary, and J. Epstein (2018). "Unmarked? Criminal record Clearing and Employment Outcomes". *Journal of Criminal Law and Criminology* 108.1.
- Shoagii, Daniel and Stan Veugeriii (2016). "No Woman No Crime Ban the Box, Employment, and Upskillingi".
- Sun, Liyang and Sarah Abraham (2021). "Estimating dynamic treatment effects in event studies with heterogeneous treatment effects". *Journal of Econometrics* 225.2. Themed Issue: Treatment Effect 1, 175–199.
- Svaver, M. (2011). "Crime and Partnerships". Review of Economics of the Household 9, 307–325.
- Uggen, C., M. Vuolo, S. Lageson, E. Ruhland, and K. Witham (2014). "The edge of stigma: An experimental audit of the effects of low-level criminal records on employment". *Criminology* 38.52, 627–654.
- Weibull, Jörgen W and Edgar Villa (2005). "Crime, punishment and social norms". Tech. rep. SSE/EFI Working Paper Series in Economics and Finance.

- Western, B. (2002). "The Impact of Incarceration on Wage Mobility and Inequality". *American Sociological Review* 67.4.
- Zedner, L. and A. Ashworth (2019). "The Rise and Restraint of the Preventive State". *Annual Review of Criminology* 2, 429–450.
- Zimring, Franklin E, Gordon Hawkins, and James Vorenberg (1973). "Deterrence: The legal threat in crime control".

Appendix A: Additional figures and tables



Notes: Figure (a) shows the average number cases for the courts in our sample by month relative to the cutoff. Figure (b) shows the average predicted risk of recidivism of defendants by month relative to cutoff. The prediction is done using the baseline individual characteristics of the defendant listed in Table 1.

Panel A: Baseline Characteristics	Exposed defendants	Non-exposed defendants
Male	0.779	0.777
	(0.415)	(0.416)
Age	29.627	29.133
	(11.966)	(12.031)
Immigrant	0.295	0.282
	(0.456)	(0.450)
High school degree	0.618	0.607
	(0.486)	(0.488)
Log earnings	5.068	4.968
	(3.375)	(3.363)
Employed	0.544	0.528
	(0.498)	(0.499)
Panel B: Case characteristics		
Acquitted	0.144	0.144
-	(0.351)	(0.351)
Misdemeanor	0.171	0.157
	(0.377)	(0.363)
Felony	0.829	0.843
	(0.377)	(0.363)
DUI	0.086	0.081
	(0.280)	(0.273)
Drug offense	0.057	0.052
-	(0.231)	(0.222)
Property offense	0.148	0.148
	(0.355)	(0.355)
Violent offense	0.173	0.182
	(0.378)	(0.385)
Traffic offense	0.157	0.147
	(0.363)	(0.354)
Other offense	0.381	0.390
	(0.486)	(0.488)
Panel C: Sentence		
Fines	0.499	0.488
	(0.500)	(0.500)
Probation	0.176	0.172
	(0.381)	(0.378)
Community service	0.214	0.217
-	(0.410)	(0.412)
Prison	0.074	0.075
	(0.262)	(0.263)
Other	0.037	0.048
	(0.189)	(0.214)
Observations	97,374	89,466

Table A1. Descriptive Statistics

Notes: This table provides summary statistics for baseline characteristics of the estimation sample for exposed and non-exposed defendants. Exposed defendants were sentenced up to 12 months after the court-specific cutoff date, making them visible in the database, and non-exposed defendants were sentenced in a 12 month window before the cutoff. Baseline characteristics are measured one year before the initial trial.

_



Notes: This figure plots the number of official criminal background check inquires for personal use (likely to be submitted to a potential employer) per registered firm submitted to the Swedish police over the period 2010-2017 using data from the Swedish police and Statistics Sweden. The figure also plots the share of ads posted at the Public Employment Office that mentions criminal background checks using data from the Public Employment Office.



Figure A3. Background checks using the platform

Notes: This figure plots the cumulative number of searches for criminal background using the platform. The figure also plots the number of registered users per background check conducted. The figure uses data provided by the company behind the platform



Figure A4. Homepage Lexbase

Notes: This image shows the homepage of Lexbase and was retrieved on April 4, 2024.

How to use Lexbase

With a few single button presses, you can obtain useful decision-making information. You start by choosing a search method. The guide below explains how the service works.

1. Select search method

A) Search person

In this search function Search person you can specify your search using four fields; First name, last name, social security number and city. If you get a hit on a person, it means that he or she has been subject to a court hearing. If you do not get a hit, it is probably because the person does not appear in any legal process.

Example: Shows results for A	nders Andersson in	Stockholm	
För- och efternamn		Or	1 Annual
Anders Andersson ×	#	× 📳 Stockholm ×	SÖK

In the search fields, you can use asterisk (*) in your search text to replace part of a search term. You also don't need to search for an exact name of a person or company. Parts of the name in question are sufficient.

If you do not get a hit on a certain person, it is probably because they are not present in any legal process.

B) Search address

This search function helps you who want information about people living in a certain area or in your immediate area. The search tool sorts by Address, Postcode and City. The more fields you fill in, the narrower the hit picture you get.

Example: Showing results on Storgatan 21 in Gothenburg.

Gata					100	A Maria
9	Storgatan 21 ×	#	٠	Göteborg		SŎK

2. Search results are displayed

When you have made a search, the search result is presented in a list. To read a report, click on the button on the right which looks like this and is called **CORECOMMAND**

Namn		Ort	
Carl HERROR Residences	30.00	Rentilled	FÖREKOMMANDE
CROPTER Automy, Anderson	14.8	84	FÖREKOMMANDE
Earths (2007) Anderson Assamt	(E) (b)	mender	FÖREKOMMANDE
KONON AND AND AND AND AND AND AND AND AND AN	31.8	Visionite	FÖREKOMMANDE
im Ovisian 10800 kolesen		Gittelong	FÖREKOMMANDE
Tataa Inderaan		Panados	FÖREKOMMANDE
KC Related Vari Redenatori		(jumpline)	FÖREKOMMANDE
Kel KOUND Indexes	66.84	United and	FÖREKOMMANDE
SHUD Herry Anderson	38.8	Robust	FÖREKOMMANDE
San Rear WOODS, Anderson		Manage .	FÖREKOMMANDE

3. Content of the report

In the judgment report, detailed information is given about the case such as the parties, the course of events and the court's examination of liability and verdict. The verdict is also available in its entirety in PDF format.

DOM				
I denoscopporten pro utilicity i pribering arrowende anovar oci	domater kring Sendet sässen domatet. Domen erhålta även i a	parter, Minder	withripp cant donishters -formut.	
- Dogleopus krott / Salum - Parter / YBRatad	- Aktuall foliookilisingualeess - Adresskarta	- Denskill - Rokground	- Bala daman i PDF-format	129 SEK
 Skraft J Politikjal Dorsonoppgilline 	- Augineraledations - Doersited	- Aklagare - Hirssarare	2	BETALA MED SALDO

4. Payment options

In order to be able to shop at Lexbase, you need to become a member with us. You become one by pressing "Become a member" on the site. We offer you several secure payment options. With us, you can choose between buying information via card, PayPal or against an invoice, or you can shop via a membership account (free) with us. If you choose to open a member account, you will receive a bonus based on how much money you deposit. You can top up your account with us at any time. Remember that you only pay when you want a report.

nce the payment is

Belopp: 12	29kr			
Valj betalningssätt:	_=	Estantido: 52 -		
Kreditkort	Bankkort	Saler becatting	~	Choose the payment method that suits you best. Once the
Swish	() swish			made, you get access to the legal information immediately.
PayPal	PayPal			
factors or inicia al testa acceptore d	web's Kimizon Alber	BETALA	TĂNG	
The repor	t is dis	played		
Once you have	paid, the	report is pre	sented i	mmediately.
1 Person				

						_	
ATTSINFORMATIC	IN IS						
,							
1. DOM 815079	2. DOM 813179	3. DOM 817769	4. DOM 83917	5. DOM 810504			
Datum domslut: 2011-10							
Domstol:					A		
DOMSLUT					LADDA NED HELA DOMAN		
Begångna brott 1. Rattfylleri 4 § 2 st tr	afikbrottslagen (1951:64	90					
2. Narkotikabrott 1 § 1 (1968:64) 3. Brott mot koldanen	st 6 p och 2 § narkotika 1 6 1 st och 4 6 1 st laor	straffagen					
förbud beträffande knir farliga föremål	var och andra						
Päföljd m.m.							
Dagsböter 100 å 50 kr							
Förverkande och besl I beslag taget knogjärn (2011-0201-8G8826-)	ng förklaras förverkat. Besl 13.	aget ska bestå					
Brottsofferfond Den tilltalade åläggs at	t betala en avgift på 500	kr enligt lagen (1994:4)	(9) om				
istory							
u can acces	s vour previo	ous reports a	at any time a	t no extra co	st. You will find	the reports in you	ir history unde
ofile.	o your provid	au reporta (acting time to	ie no extra ce	Ser roa win mia	and reports in you	in motory unde
0.11.01							

Figure A5. How the Search-Engine at Lexbase Works

Notes: This image is from the homepage of Lexbase and provides the guide from the platform on how to use the search engine. The image was retrieved on April 4, 2024.



Figure A6. Cumulative number of media articles covering Lexbase 2013-2016

Notes: This figure plots the cumulative number of media articles mentioning Lexbase in 2013-2016. Data on media coverage is from the Retriever/Mediearkivet database, which is a Nordic news archive with material from mainly major Swedish newspapers. The information was collected on Jan 20th, 2024.



Figure A7. Relative number of searches on Google

Notes: This figure plots the relative number of searches on Google for the terms "Lexbase", "Best restaurants" and "New car". The data from Google Trends was collected on Jan 21st, 2024. Google Trends normalizes search data to make comparisons between terms easier. Search results are normalized to the time and location of a query by dividing each data point by the total searches of the geography and time range it represents. The resulting numbers are then scaled on a range of 0 to 100 based on a topic's proportion to all searches on all topics. To make the figure more easy to read we drop the days immediately following the launch of the database when the searches skyrocketed and focus on the period starting one week after the introduction and up until the last week of 2015. The information was collected on Jan 20th, 2024.





Notes: The figure plots the share of Lexbase cases in our register data within a 12 month window around the cutoff separately for the district courts (DC) in our sample.



Figure A10. Case characteristics around the cutoff (*a*) Age

(b) Males

(c) Immigrants



(d) High school degree









Notes: The figure shows the average case and defendant characteristics by time relative to cutoff.



Figure A11. Annual DD estimates of the exposure effects (*a*) Conviction (*b*) Employment

Notes: This figure plots DD-estimates from model (1) by year relative to 2013. All regressions control for year, court, time since charge, calendar month, age, gender, immigrant and the main effect of being exposed. The bars show the 95 percent confidence interval based on standard errors clustered at the individual and court level. Panel (a) shows conviction between 2011-2016, Panels (b)-(c) shows labor market outcomes in 2011-2016.

Figure A12. Effect of exposure on position in the earnings distribution



Notes: This figure plots the estimated exposure effects on (solid line) along with the 90 percent confidence interval (dashed line) based on model (1) where the dependent variable is an indicator for the individual earning more than the i:th percentile in the earnings distribution. The regressions control for age, gender, immigrant and the main effect of being exposed. Numbers in brackets show mean of the dependent variable. Standard errors in parenthesis are clustered at the individual and court level.



Notes: This figure estimates event study coefficients +/- 7 years around the charge, where the end points after +/- 5 year are binned. Panel (a) gives the coefficients for employment, and Panel (b) gives the coefficients for log earnings. The sample is restricted to first-time offenders between 1997-2010 for those aged 25-57 at the time of the charge. All regressions control for age, year and individual fixed effects. The bars show the 95 percent confidence interval based on standard errors clustered at the individual level.





Notes: This figure estimates event study coefficients +/- 7 years around the charge, where the end points after +/- 5 year are binned. The procedure from Sun and Abraham (2021) is employed, using the last treated unit as control, excluding time periods after the last unit received treatment. Panel (a) gives the coefficients for employment, and Panel (b) gives the coefficients for log earnings. The sample is restricted to first-time offenders between 1997-2010 for those aged 25-57 at the time of the charge. All regressions control for age, year and individual fixed effects. The bars show the 95 percent confidence interval based on standard errors clustered at the individual level.

Table	A2.	Robustness

	Convicted	Employed	Log Earnings
Panel A: Cutoff -1 (N=172,548)	(1)	(2)	(3)
Exposed*Post	-0.000	0.003	0.023
	(0.003)	(0.006)	(0.045)
Outcome mean	0.084	0.674	5.875
<i>Panel B: Cutoff</i> +1 (<i>N</i> =206,364)			
Exposed*Post	-0.004	-0.003	-0.048
	(0.003)	(0.005)	(0.030)
Outcome mean	0.101	0.648	5.728
Panel C: Bandwidth 9 months (N	=142,764)		
Exposed*Post	0.002	-0.011*	-0.120***
	(0.003)	(0.006)	(0.031)
Outcome mean	0.088	0.665	5.825
Panel D: Bandwidth 6 months (N	=95,724)		
Exposed*Post	0.001	-0.010*	-0.135***
	(0.004)	(0.006)	(0.030)
Outcome mean	0.087	0.667	5.847
Panel E: Short-term effects $(N=1)$	36,656)		
Exposed*Post	0.004	-0.007	-0.098***
-	(0.004)	(0.006)	(0.035)
Outcome mean	0.101	0.532	5.118
Year FE	Yes	Yes	Yes
Court FE	Yes	Yes	Yes
Time since charge FE	Yes	Yes	Yes
Calendar month FE	Yes	Yes	Yes
Baseline controls	Yes	Yes	Yes

Note: This table reports difference-in-differences estimates based on model (1) of the effect of exposure on crime and labor market outcomes. Panels A and B show results from placebo cutoff dates one year before and one year after the actual cutoff date in a given court, respectively. Panels C and D show results using alternative bandwidths around the court-specific cutoff date. Panel E shows estimates from an alternative difference-in-differences model where the "exposed" group includes defendants who underwent a trial, while the "non-exposed" group comprises defendants who were directly convicted by a prosecutor for confession cases. These confession cases were not included in the database in 2014 and 2015. Outcomes are measured one year post-conviction and the repeated cross-sectional dataset runs from 2010 to 2015. All regressions control for age, gender, immigrant and the main effect of being exposed. Numbers in brackets show mean of the dependent variable. Standard errors in parenthesis are clustered at the individual and court level. *** = significant at 1 percent level, ** = significant at 5 percent level, * = significant at 10 percent level.

	All
	(1)
In college	0.004*
	(0.002)
	[0.029]
Labor market training	0.001
	(0.002)
	[0.011]
Moving	0.004
	(0.006)
	[0.227]
Single household	0.001
	(0.005)
	[0.703]
Get pregnant	-0.001
	(0.003)
	[0.052]
Year FE	Yes
Court FE	Yes
Time since charge FE	Yes
Calendar month FE	Yes
Baseline Controls	Yes
Observations	186,840

Table A3. Additional outcomes

Note: This table reports difference-in-differences estimates based on model (1) for alternative outcomes. The regressions control for age, gender, immigrant and the main effect of being exposed. Numbers in brackets show the mean of the dependent variables. Standard errors are clustered at the individual and court level. *** = significant at 1 percent level, ** = significant at 5 percent level, * = significant at 10 percent level.

	Ratio	Penetration classification
	(1)	(2)
Banking	2.2	Low
Construction	0.4	High
Culture	1.2	High
Education	2.7	Low
Electricity	1.9	Low
Farming	1.5	High
Healthcare	1.6	Low
Hotel and Restaurants	0.6	High
Information	1.5	High
Law and Economics	1.3	High
Letting	0.6	High
Manufacturing	1.0	High
Mining	0.9	High
Other services	1.4	High
Public Administration	2.6	Low
Real estate	1.2	High
Retail	0.9	High
Transport	0.5	High
Water & Sanitation	0.6	High

Table A4. Employment penetration by industry

Note: This table presents the ratio of employment in a particular industry for individuals without a court conviction relative to those with a court conviction in year 2013. No adjustments for baseline characteristics are done. Equal employment penetration among these two groups is expressed as a ratio equal to 1. We classify industries as low penetration if the ratio exceeds 1.5.

Table A5.	Industry	effects

_

	All
	(1)
Empl. in high penetration industries 1.5	-0.010*
	(0.006)
	[0.584]
Empl. in low penetration industries 1.5	-0.003
	(0.003)
	[0.150]
Empl. in high penetration industries 1.2	-0.008*
	(0.004)
	[0.489]
Empl. in low penetration industries 1.2	-0.004
	(0.005)
	[0.245]
Empl. in high penetration industries 1	-0.005
	(0.005)
	[0.398]
Empl. in low penetration industries 1	-0.00
	(0.005)
	[0.336]
Year FE	Yes
Court FE	Yes
Time since charge FE	Yes
Calendar month FE	Yes
Baseline Controls	Yes
Observations	186,840

Note: This table reports difference-in-differences estimates based on model (1) of the effect of exposure on sorting across industries. The regressions control for age, gender, immigrant and the main effect of being exposed. Numbers in brackets show mean of the dependent variable. Standard errors are clustered at the individual level. *** = significant at 1 percent level, ** = significant at 5 percent level, * = significant at 10 percent level.

	(1)	(7)	$\hat{\boldsymbol{c}}$	~ ~				0.016^{**}
Exposed x Post	0.002	0.016	0.014^{*}	-0.005	0.021*	0.006	0.011	
	(0.007)	(0.012)	(0.007)	(0.006)	(0000)	(0.006)	(0.008)	(0.008)
x Male		-0.017* (0.009)						
x At least age 25			-0.022** (0.006)					
x Immigrant				0.033^{***} (0.009)				
x HS degree					-0.031*** (0.008)			
x Acquitted						-0.019** (0.009)		
x Misdemaneor							-0.036^{***} (0.009)	
x Nbh offenders < med.								-0.032*** (0.007)
Year FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Court FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Baseline Controls	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Observations	100,152	100,152	100,152	100,152	100,152	100,152	100,152	100,152
	(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)
--------------------------	---------	---------	-----------	---------	---------------------	--------------	---------------	---------
Exposed x Post		-0.002	-0.029***	-0.008	-0.023**	-0.010	-0.013*	-0.011
4		(0.013)	(600.0)	(0.010)	(0.010)	(0.007)	(0.008)	(0.008)
x Male		-0.007						
		(0.011)						
x At least age 25			0.040***					
x Immigrant			(000.0)	0.003				
)				(0.001)				
x HS degree					0.024*** (0.006)			
x Acquitted						0.017^{**}		
						(0.007)		
x Misdemaneor							0.025^{***}	
							(0.008)	
x Nbh offenders < med.								0.007
								(0.006)
Year FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Court FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Baseline Controls	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Observations	103,908	103,908	103,908	103,908	103,908	103,908	103,908	103,908

firm from one year to another and zero if (s)he stayed in the firm. Individuals who are non-employed are missing in the data. The regressions control for the same variables as in Table 2. Standard errors are clustered at the individual and court level. *** = significant at 1 percent level, ** = significant at 5 percent level, * = significant at 10 percent level. Note: Th

Table A7. Firm switching

									,
g disp. inc.	(8)	-0.036	(0.034)	8.084	Yes	Yes	Yes	23,388	,
Family lo	(2)	-0.036	(0.035)	8.084	Yes	Yes	No	23,388	.
rnings	(9)	-0.104	(0.074)	6.221	Yes	Yes	Yes	23,388	
Log ea	(5)	-0.104	(0.074)	6.221	Yes	Yes	N_0	23,388	
loyed	(4)	-0.023*	(0.013)	0.735	Yes	Yes	Yes	23,388	•
Emp	(3)	-0.023*	(0.013)	0.735	Yes	Yes	No	23,388	
icted	(2)	0.004	(0.006)	0.049	Yes	Yes	Yes	23,388	
Convi	(1)	0.004	(0.006)	0.049	Yes	Yes	No	23,388	
		Exposed*Post		Sample mean	Year FE	Court FE	Baseline controls	Observations	

Table A8. Results for parental subsample

Note: This table reports difference-in-differences estimates based on model (1) for crime and labor market outcomes restricted to the subsample of individuals in our main sample that were parents of children belonging to the grade 9 graduation cohorts 2010-2016. Exposed is an indicator variable set to one for defendants sentenced up to 12 months after the court-specific cutoff date; zero otherwise. Post is an indicator variable set to one for the years the database was in place (i.e. 2014 and onwards); zero otherwise. The regressions control for the same variables as in Table 2. Numbers in bracket show the mean of the dependent variable. Standard errors in parenthesis are clustered at the individual and court level. *** = significant at 1 percent level, ** = significant at 5 percent level, * = significant at 10 percent level.

Table A9.	Effects	on	children	's	outcomes
-----------	---------	----	----------	----	----------

Panel A: GPA (pct. rank)	(1)	(2)	(3)	(4)	(5)	(6)
Exposed x Post	-2.203	0.190	-1.043	-2.170	-0.799	-0.709
	(2.192)	(2.394)	(2.638)	(2.241)	(2.234)	(2.025)
x Defendant is the father		-3.368**				
		(1.671)				
x Parent HS degree			-1.828			
			(1.763)			
x Parent aqcuitted				-0.736		
				(1.493)		
x Parent misdemaneor					-3.827*	
					(1.972)	
x Nbh offenders < med						-3.706**
						(1.572)
Sample mean	42.572	42.572	42.572	42.572	42.572	42.572
Panel B: Court conviction						
Exposed x Post	0.009	-0.013	-0.000	0.009	-0.000	0.008
	(0.010)	(0.009)	(0.011)	(0.010)	(0.016)	(0.010)
x Defendant is the father		0.032***				
		(0.009)				
x Parent HS degree			0.012			
			(0.011)			
x Parent aqcuitted				0.0053		
				(0.013)		
x Parent misdemaneor					0.003	
					(0.011)	
x Nbh offenders < med						0.003
						(0.010)
Sample mean	0.023	0.023	0.023	0.023	0.023 2	0.023
Year FE	Yes	Yes	Yes	Yes	Yes	Yes
Court FE	Yes	Yes	Yes	Yes	Yes	Yes
Baseline Controls	Yes	Yes	Yes	Yes	Yes	Yes
Observations	5,119	5,119	5,119	5,119	5,119	5,119

Note: This table reports difference-in-differences estimates based on model (1) of the effect of exposure on compulsory school GPA (percentile rank) in grade 9. The sample consist of children to defendants in our main sample belonging to the cohorts who graduated from compulsory school 2010-2016. The regressions control for the same characteristics of the parents as in Table 2. Numbers in brackets show mean of the dependent variable. Standard errors in parenthesis are clustered at the individual and court level. *** = significant at 1 percent level, ** = significant at 5 percent level, * = significant at 10 percent level.

Appendix B: Data

Exposed: An indicator equal to one if the individual was sentenced within 12 months after the court-specific cutoff month.

Non-Exposed: An indicator equal to one if the individual was sentenced within 12 months before the court-specific cutoff month.

Male: An indicator equal to one if the individual is a male.

Immigrant: An indicator equal to one if the individual was born outside of Sweden.

Age at Initial Trial: We calculate age at initial trial as the year of the trial minus the calendar year when the individual is born.

High School Degree: An indicator for whether an individual has a high school degree or a higher education degree (e.g., college, university).

Convicted: An indicator set to one if an individual was charged and convicted in the initial trial.

Acquitted: An indicator set to one if an individual was charged and acquitted in the initial trial.

Misdemeanor: An indicator set to one if the the offender committed a crime that no defendant has ever been sentenced to prison for in our data.

Convictions per 10,000 inhabitants: The number of convictions per 10,000 inhabitants in the municipality of residence.

Criminal Conviction: An indicator for whether the charged individual is convicted of new crime in a given year.

Employment: Measures the employment status of the charged individual in a given year. A binary employment indicator set equal to one if the charged individual is registered as employed on Nov 1st.

Earnings (\$1,000s)): Annual labor earnings. Earnings are deflated to 2015 and represented in U.S. dollars using the exchange rate SEK/\$ = 9.25.

Mobility: An indicator set to one if the individual has changed place of residence in a given year.

In college: An indicator set to one if the individual was enrolled in college education in a given year.

Labor market training: An indicator set to one if the individual was enrolled in labor market training in a given year.

Economic Studies

- 1987:1 Haraldson, Marty. To Care and To Cure. A linear programming approach to national health planning in developing countries. 98 pp.
- 1989:1 Chryssanthou, Nikos. The Portfolio Demand for the ECU. A Transaction Cost Approach. 42 pp.
- 1989:2 Hansson, Bengt. Construction of Swedish Capital Stocks, 1963-87. An Application of the Hulten-Wykoff Studies. 37 pp.
- 1989:3 Choe, Byung-Tae. Some Notes on Utility Functions Demand and Aggregation. 39 pp.
- 1989:4 Skedinger, Per. Studies of Wage and Employment Determination in the Swedish Wood Industry. 89 pp.
- 1990:1 Gustafson, Claes-Håkan. Inventory Investment in Manufacturing Firms. Theory and Evidence. 98 pp.
- 1990:2 Bantekas, Apostolos. The Demand for Male and Female Workers in Swedish Manufacturing. 56 pp.
- 1991:1 Lundholm, Michael. Compulsory Social Insurance. A Critical Review. 109 pp.
- 1992:1 Sundberg, Gun. The Demand for Health and Medical Care in Sweden. 58 pp.
- 1992:2 Gustavsson, Thomas. No Arbitrage Pricing and the Term Structure of Interest Rates. 47 pp.
- 1992:3 Elvander, Nils. Labour Market Relations in Sweden and Great Britain. A Comparative Study of Local Wage Formation in the Private Sector during the 1980s. 43 pp.
- 12 Dillén, Mats. Studies in Optimal Taxation, Stabilization, and Imperfect Competition. 1993. 143 pp.
- 13 Banks, Ferdinand E., A Modern Introduction to International Money, Banking and Finance. 1993. 303 pp.
- 14 Mellander, Erik. Measuring Productivity and Inefficiency Without Quantitative Output Data. 1993. 140 pp.
- 15 Ackum Agell. Susanne. Essays on Work and Pay. 1993. 116 pp.
- 16 Eriksson, Claes. Essays on Growth and Distribution. 1994. 129 pp.
- 17 Banks, Ferdinand E., A Modern Introduction to International Money, Banking and Finance. 2nd version, 1994. 313 pp.

- 18 Apel, Mikael. Essays on Taxation and Economic Behavior. 1994. 144 pp.
- 19 Dillén, Hans. Asset Prices in Open Monetary Economies. A Contingent Claims Approach. 1994. 100 pp.
- 20 Jansson, Per. Essays on Empirical Macroeconomics. 1994. 146 pp.
- 21 Banks, Ferdinand E., A Modern Introduction to International Money, Banking, and Finance. 3rd version, 1995. 313 pp.
- 22 Dufwenberg, Martin. On Rationality and Belief Formation in Games. 1995. 93 pp.
- 23 Lindén, Johan. Job Search and Wage Bargaining. 1995. 127 pp.
- 24 Shahnazarian, Hovick. Three Essays on Corporate Taxation. 1996. 112 pp.
- Svensson, Roger. Foreign Activities of Swedish Multinational Corporations. 1996.
 166 pp.
- 26 Sundberg, Gun. Essays on Health Economics. 1996. 174 pp.
- 27 Sacklén, Hans. Essays on Empirical Models of Labor Supply. 1996. 168 pp.
- 28 Fredriksson, Peter. Education, Migration and Active Labor Market Policy. 1997. 106 pp.
- 29 Ekman, Erik. Household and Corporate Behaviour under Uncertainty. 1997. 160 pp.
- 30 Stoltz, Bo. Essays on Portfolio Behavior and Asset Pricing. 1997. 122 pp.
- 31 Dahlberg, Matz. Essays on Estimation Methods and Local Public Economics. 1997. 179 pp.
- 32 Kolm, Ann-Sofie. Taxation, Wage Formation, Unemployment and Welfare. 1997. 162 pp.
- Boije, Robert. Capitalisation, Efficiency and the Demand for Local Public Services. 1997.148 pp.
- 34 Hort, Katinka. On Price Formation and Quantity Adjustment in Swedish Housing Markets. 1997. 185 pp.
- 35 Lindström, Thomas. Studies in Empirical Macroeconomics. 1998. 113 pp.
- 36 Hemström, Maria. Salary Determination in Professional Labour Markets. 1998. 127 pp.
- 37 Forsling, Gunnar. Utilization of Tax Allowances and Corporate Borrowing. 1998. 96 pp.
- 38 Nydahl, Stefan. Essays on Stock Prices and Exchange Rates. 1998. 133 pp.
- 39 Bergström, Pål. Essays on Labour Economics and Econometrics. 1998. 163 pp.

- 40 Heiborn, Marie. Essays on Demographic Factors and Housing Markets. 1998. 138 pp.
- 41 Åsberg, Per. Four Essays in Housing Economics. 1998. 166 pp.
- 42 Hokkanen, Jyry. Interpreting Budget Deficits and Productivity Fluctuations. 1998. 146 pp.
- 43 Lunander, Anders. Bids and Values. 1999. 127 pp.
- 44 Eklöf, Matias. Studies in Empirical Microeconomics. 1999. 213 pp.
- Johansson, Eva. Essays on Local Public Finance and Intergovernmental Grants. 1999.
 156 pp.
- 46 Lundin, Douglas. Studies in Empirical Public Economics. 1999. 97 pp.
- 47 Hansen, Sten. Essays on Finance, Taxation and Corporate Investment. 1999. 140 pp.
- 48 Widmalm, Frida. Studies in Growth and Household Allocation. 2000. 100 pp.
- 49 Arslanogullari, Sebastian. Household Adjustment to Unemployment. 2000. 153 pp.
- 50 Lindberg, Sara. Studies in Credit Constraints and Economic Behavior. 2000. 135 pp.
- 51 Nordblom, Katarina. Essays on Fiscal Policy, Growth, and the Importance of Family Altruism. 2000. 105 pp.
- 52 Andersson, Björn. Growth, Saving, and Demography. 2000. 99 pp.
- 53 Åslund, Olof. Health, Immigration, and Settlement Policies. 2000. 224 pp.
- 54 Bali Swain, Ranjula. Demand, Segmentation and Rationing in the Rural Credit Markets of Puri. 2001. 160 pp.
- 55 Löfqvist, Richard. Tax Avoidance, Dividend Signaling and Shareholder Taxation in an Open Economy. 2001. 145 pp.
- 56 Vejsiu, Altin. Essays on Labor Market Dynamics. 2001. 209 pp.
- 57 Zetterström, Erik. Residential Mobility and Tenure Choice in the Swedish Housing Market. 2001. 125 pp.
- 58 Grahn, Sofia. Topics in Cooperative Game Theory. 2001. 106 pp.
- 59 Laséen, Stefan. Macroeconomic Fluctuations and Microeconomic Adjustments. Wages, Capital, and Labor Market Policy. 2001. 142 pp.
- 60 Arnek, Magnus. Empirical Essays on Procurement and Regulation. 2002. 155 pp.
- Jordahl, Henrik. Essays on Voting Behavior, Labor Market Policy, and Taxation. 2002.
 172 pp.

- 62 Lindhe, Tobias. Corporate Tax Integration and the Cost of Capital. 2002. 102 pp.
- 63 Hallberg, Daniel. Essays on Household Behavior and Time-Use. 2002. 170 pp.
- 64 Larsson, Laura. Evaluating Social Programs: Active Labor Market Policies and Social Insurance. 2002. 126 pp.
- 65 Bergvall, Anders. Essays on Exchange Rates and Macroeconomic Stability. 2002. 122 pp.
- 66 Nordström Skans, Oskar. Labour Market Effects of Working Time Reductions and Demographic Changes. 2002. 118 pp.
- 67 Jansson, Joakim. Empirical Studies in Corporate Finance, Taxation and Investment. 2002. 132 pp.
- 68 Carlsson, Mikael. Macroeconomic Fluctuations and Firm Dynamics: Technology, Production and Capital Formation. 2002. 149 pp.
- 69 Eriksson, Stefan. The Persistence of Unemployment: Does Competition between Employed and Unemployed Job Applicants Matter? 2002. 154 pp.
- 70 Huitfeldt, Henrik. Labour Market Behaviour in a Transition Economy: The Czech Experience. 2003. 110 pp.
- 71 Johnsson, Richard. Transport Tax Policy Simulations and Satellite Accounting within a CGE Framework. 2003. 84 pp.
- 72 Öberg, Ann. Essays on Capital Income Taxation in the Corporate and Housing Sectors. 2003. 183 pp.
- 73 Andersson, Fredrik. Causes and Labor Market Consequences of Producer Heterogeneity. 2003. 197 pp.
- 74 Engström, Per. Optimal Taxation in Search Equilibrium. 2003. 127 pp.
- 75 Lundin, Magnus. The Dynamic Behavior of Prices and Investment: Financial Constraints and Customer Markets. 2003. 125 pp.
- 76 Ekström, Erika. Essays on Inequality and Education. 2003. 166 pp.
- 77 Barot, Bharat. Empirical Studies in Consumption, House Prices and the Accuracy of European Growth and Inflation Forecasts. 2003. 137 pp.
- 78 Österholm, Pär. Time Series and Macroeconomics: Studies in Demography and Monetary Policy. 2004. 116 pp.
- 79 Bruér, Mattias. Empirical Studies in Demography and Macroeconomics. 2004. 113 pp.
- 80 Gustavsson, Magnus. Empirical Essays on Earnings Inequality. 2004. 154 pp.

- 81 Toll, Stefan. Studies in Mortgage Pricing and Finance Theory. 2004. 100 pp.
- 82 Hesselius, Patrik. Sickness Absence and Labour Market Outcomes. 2004. 109 pp.
- 83 Häkkinen, Iida. Essays on School Resources, Academic Achievement and Student Employment. 2004. 123 pp.
- 84 Armelius, Hanna. Distributional Side Effects of Tax Policies: An Analysis of Tax Avoidance and Congestion Tolls. 2004. 96 pp.
- 85 Ahlin, Åsa. Compulsory Schooling in a Decentralized Setting: Studies of the Swedish Case. 2004. 148 pp.
- Heldt, Tobias. Sustainable Nature Tourism and the Nature of Tourists' Cooperative Behavior: Recreation Conflicts, Conditional Cooperation and the Public Good Problem.
 2005. 148 pp.
- 87 Holmberg, Pär. Modelling Bidding Behaviour in Electricity Auctions: Supply Function Equilibria with Uncertain Demand and Capacity Constraints. 2005. 43 pp.
- 88 Welz, Peter. Quantitative new Keynesian macroeconomics and monetary policy 2005. 128 pp.
- 89 Ågren, Hanna. Essays on Political Representation, Electoral Accountability and Strategic Interactions. 2005. 147 pp.
- 90 Budh, Erika. Essays on environmental economics. 2005. 115 pp.
- 91 Chen, Jie. Empirical Essays on Housing Allowances, Housing Wealth and Aggregate Consumption. 2005. 192 pp.
- 92 Angelov, Nikolay. Essays on Unit-Root Testing and on Discrete-Response Modelling of Firm Mergers. 2006. 127 pp.
- 93 Savvidou, Eleni. Technology, Human Capital and Labor Demand. 2006. 151 pp.
- 94 Lindvall, Lars. Public Expenditures and Youth Crime. 2006. 112 pp.
- 95 Söderström, Martin. Evaluating Institutional Changes in Education and Wage Policy. 2006. 131 pp.
- 96 Lagerström, Jonas. Discrimination, Sickness Absence, and Labor Market Policy. 2006. 105 pp.
- 97 Johansson, Kerstin. Empirical essays on labor-force participation, matching, and trade. 2006. 168 pp.
- 98 Ågren, Martin. Essays on Prospect Theory and the Statistical Modeling of Financial Returns. 2006. 105 pp.

- 99 Nahum, Ruth-Aïda. Studies on the Determinants and Effects of Health, Inequality and Labour Supply: Micro and Macro Evidence. 2006. 153 pp.
- 100 Žamac, Jovan. Education, Pensions, and Demography. 2007. 105 pp.
- 101 Post, Erik. Macroeconomic Uncertainty and Exchange Rate Policy. 2007. 129 pp.
- 102 Nordberg, Mikael. Allies Yet Rivals: Input Joint Ventures and Their Competitive Effects. 2007. 122 pp.
- 103 Johansson, Fredrik. Essays on Measurement Error and Nonresponse. 2007. 130 pp.
- 104 Haraldsson, Mattias. Essays on Transport Economics. 2007. 104 pp.
- 105 Edmark, Karin. Strategic Interactions among Swedish Local Governments. 2007. 141 pp.
- 106 Oreland, Carl. Family Control in Swedish Public Companies. Implications for Firm Performance, Dividends and CEO Cash Compensation. 2007. 121 pp.
- 107 Andersson, Christian. Teachers and Student Outcomes: Evidence using Swedish Data. 2007. 154 pp.
- 108 Kjellberg, David. Expectations, Uncertainty, and Monetary Policy. 2007. 132 pp.
- 109 Nykvist, Jenny. Self-employment Entry and Survival Evidence from Sweden. 2008. 94 pp.
- 110 Selin, Håkan. Four Empirical Essays on Responses to Income Taxation. 2008. 133 pp.
- 111 Lindahl, Erica. Empirical studies of public policies within the primary school and the sickness insurance. 2008. 143 pp.
- 112 Liang, Che-Yuan. Essays in Political Economics and Public Finance. 2008. 125 pp.
- 113 Elinder, Mikael. Essays on Economic Voting, Cognitive Dissonance, and Trust. 2008. 120 pp.
- 114 Grönqvist, Hans. Essays in Labor and Demographic Economics. 2009. 120 pp.
- 115 Bengtsson, Niklas. Essays in Development and Labor Economics. 2009. 93 pp.
- 116 Vikström, Johan. Incentives and Norms in Social Insurance: Applications, Identification and Inference. 2009. 205 pp.
- 117 Liu, Qian. Essays on Labor Economics: Education, Employment, and Gender. 2009. 133 pp.
- 118 Glans, Erik. Pension reforms and retirement behaviour. 2009. 126 pp.
- 119 Douhan, Robin. Development, Education and Entrepreneurship. 2009.

- 120 Nilsson, Peter. Essays on Social Interactions and the Long-term Effects of Early-life Conditions. 2009. 180 pp.
- 121 Johansson, Elly-Ann. Essays on schooling, gender, and parental leave. 2010. 131 pp.
- 122 Hall, Caroline. Empirical Essays on Education and Social Insurance Policies. 2010. 147 pp.
- 123 Enström-Öst, Cecilia. Housing policy and family formation. 2010. 98 pp.
- 124 Winstrand, Jakob. Essays on Valuation of Environmental Attributes. 2010. 96 pp.
- 125 Söderberg, Johan. Price Setting, Inflation Dynamics, and Monetary Policy. 2010. 102 pp.
- 126 Rickne, Johanna. Essays in Development, Institutions and Gender. 2011. 138 pp.
- Hensvik, Lena. The effects of markets, managers and peers on worker outcomes. 2011.179 pp.
- 128 Lundqvist, Heléne. Empirical Essays in Political and Public. 2011. 157 pp.
- 129 Bastani, Spencer. Essays on the Economics of Income Taxation. 2012. 257 pp.
- 130 Corbo, Vesna. Monetary Policy, Trade Dynamics, and Labor Markets in Open Economies. 2012. 262 pp.
- 131 Nordin, Mattias. Information, Voting Behavior and Electoral Accountability. 2012. 187 pp.
- 132 Vikman, Ulrika. Benefits or Work? Social Programs and Labor Supply. 2013. 161 pp.
- 133 Ek, Susanne. Essays on unemployment insurance design. 2013. 136 pp.
- 134 Österholm, Göran. Essays on Managerial Compensation. 2013. 143 pp.
- 135 Adermon, Adrian. Essays on the transmission of human capital and the impact of technological change. 2013. 138 pp.
- 136 Kolsrud, Jonas. Insuring Against Unemployment 2013. 140 pp.
- 137 Hanspers, Kajsa. Essays on Welfare Dependency and the Privatization of Welfare Services. 2013. 208 pp.
- 138 Persson, Anna. Activation Programs, Benefit Take-Up, and Labor Market Attachment. 2013. 164 pp.
- 139 Engdahl, Mattias. International Mobility and the Labor Market. 2013. 216 pp.
- 140 Krzysztof Karbownik. Essays in education and family economics. 2013. 182 pp.

- 141 Oscar Erixson. Economic Decisions and Social Norms in Life and Death Situations. 2013.183 pp.
- 142 Pia Fromlet. Essays on Inflation Targeting and Export Price Dynamics. 2013. 145 pp.
- 143 Daniel Avdic. Microeconometric Analyses of Individual Behavior in Public Welfare Systems. Applications in Health and Education Economics. 2014. 176 pp.
- 144 Arizo Karimi. Impacts of Policies, Peers and Parenthood on Labor Market Outcomes. 2014. 221 pp.
- 145 Karolina Stadin. Employment Dynamics. 2014. 134 pp.
- 146 Haishan Yu. Essays on Environmental and Energy Economics. 132 pp.
- 147 Martin Nilsson. Essays on Health Shocks and Social Insurance. 139 pp.
- 148 Tove Eliasson. Empirical Essays on Wage Setting and Immigrant Labor Market Opportunities. 2014. 144 pp.
- 149 Erik Spector. Financial Frictions and Firm Dynamics. 2014. 129 pp.
- 150 Michihito Ando. Essays on the Evaluation of Public Policies. 2015. 193 pp.
- 151 Selva Bahar Baziki. Firms, International Competition, and the Labor Market. 2015. 183 pp.
- 152 Fredrik Sävje. What would have happened? Four essays investigating causality. 2015. 229 pp.
- 153 Ina Blind. Essays on Urban Economics. 2015. 197 pp.
- 154 Jonas Poulsen. Essays on Development and Politics in Sub-Saharan Africa. 2015. 240 pp.
- 155 Lovisa Persson. Essays on Politics, Fiscal Institutions, and Public Finance. 2015. 137 pp.
- 156 Gabriella Chirico Willstedt. Demand, Competition and Redistribution in Swedish Dental Care. 2015. 119 pp.
- 157 Yuwei Zhao de Gosson de Varennes. Benefit Design, Retirement Decisions and Welfare Within and Across Generations in Defined Contribution Pension Schemes. 2016. 148 pp.
- 158 Johannes Hagen. Essays on Pensions, Retirement and Tax Evasion. 2016. 195 pp.
- 159 Rachatar Nilavongse. Housing, Banking and the Macro Economy. 2016. 156 pp.
- 160 Linna Martén. Essays on Politics, Law, and Economics. 2016. 150 pp.
- 161 Olof Rosenqvist. Essays on Determinants of Individual Performance and Labor Market Outcomes. 2016. 151 pp.
- 162 Linuz Aggeborn. Essays on Politics and Health Economics. 2016. 203 pp.

- 163 Glenn Mickelsson. DSGE Model Estimation and Labor Market Dynamics. 2016. 166 pp.
- 164 Sebastian Axbard. Crime, Corruption and Development. 2016. 150 pp.
- 165 Mattias Öhman. Essays on Cognitive Development and Medical Care. 2016. 181 pp.
- 166 Jon Frank. Essays on Corporate Finance and Asset Pricing. 2017. 160 pp.
- 167 Ylva Moberg. Gender, Incentives, and the Division of Labor. 2017. 220 pp.
- 168 Sebastian Escobar. Essays on inheritance, small businesses and energy consumption. 2017. 194 pp.
- 169 Evelina Björkegren. Family, Neighborhoods, and Health. 2017. 226 pp.
- 170 Jenny Jans. Causes and Consequences of Early-life Conditions. Alcohol, Pollution and Parental Leave Policies. 2017. 209 pp.
- 171 Josefine Andersson. Insurances against job loss and disability. Private and public interventions and their effects on job search and labor supply. 2017. 175 pp.
- 172 Jacob Lundberg. Essays on Income Taxation and Wealth Inequality. 2017. 173 pp.
- 173 Anna Norén. Caring, Sharing, and Childbearing. Essays on Labor Supply, Infant Health, and Family Policies. 2017. 206 pp.
- 174 Irina Andone. Exchange Rates, Exports, Inflation, and International Monetary Cooperation. 2018. 174 pp.
- 175 Henrik Andersson. Immigration and the Neighborhood. Essays on the Causes and Consequences of International Migration. 2018. 181 pp.
- 176 Aino-Maija Aalto. Incentives and Inequalities in Family and Working Life. 2018. 131 pp.
- 177 Gunnar Brandén. Understanding Intergenerational Mobility. Inequality, Student Aid and Nature-Nurture Interactions. 2018. 125 pp.
- 178 Mohammad H. Sepahvand. Essays on Risk Attitudes in Sub-Saharan Africa. 2019. 215 pp.
- 179 Mathias von Buxhoeveden. Partial and General Equilibrium Effects of Unemployment Insurance. Identification, Estimation and Inference. 2019. 89 pp.
- 180 Stefano Lombardi. Essays on Event History Analysis and the Effects of Social Programs on Individuals and Firms. 2019. 150 pp.
- 181 Arnaldur Stefansson. Essays in Public Finance and Behavioral Economics. 2019. 191 pp.
- 182 Cristina Bratu. Immigration: Policies, Mobility and Integration. 2019. 173 pp.
- 183 Tamás Vasi. Banks, Shocks and Monetary Policy. 2020. 148 pp.

- 184 Jonas Cederlöf. Job Loss: Consequences and Labor Market Policy. 2020. 213 pp.
- 185 Dmytro Stoyko. Expectations, Financial Markets and Monetary Policy. 2020. 153 pp.
- 186 Paula Roth. Essays on Inequality, Insolvency and Innovation. 2020. 191 pp.
- 187 Fredrik Hansson. Consequences of Poor Housing, Essays on Urban and Health Economics. 2020. 143 pp.
- 188 Maria Olsson. Essays on Macroeconomics: Wage Rigidity and Aggregate Fluctuations. 2020. 130 pp.
- 189 Dagmar Müller. Social Networks and the School-to-Work Transition. 2020. 146 pp.
- 190 Maria Sandström. Essays on Savings and Intangible Capital. 2020. 129 pp.
- Anna Thoresson. Wages and Their Impact on Individuals, Households and Firms. 2020.
 220 pp.
- 192. Jonas Klarin. Empirical Essays in Public and Political Economics. 2020. 129 pp.
- 193. André Reslow. Electoral Incentives and Information Content in Macroeconomic Forecasts. 2021. 184 pp.
- Davide Cipullo. Political Careers, Government Stability, and Electoral Cycles. 2021.
 308 pp.
- 195. Olle Hammar. The Mystery of Inequality: Essays on Culture, Development, and Distributions. 2021. 210 pp.
- 196. J. Lucas Tilley. Inputs and Incentives in Education. 2021. 184 pp.
- 197. Sebastian Jävervall. Corruption, Distortions and Development. 2021. 215 pp.
- Vivika Halapuu. Upper Secondary Education: Access, Choices and Graduation. 2021. 141 pp.
- Charlotte Paulie. Essays on the Distribution of Production, Prices and Wealth. 2021. 141 pp.
- Kerstin Westergren. Essays on Inflation Expectations, Monetary Policy and Tax Reform. 2021. 124 pp.
- 201. Melinda Süveg. Finance, Shocks, Competition and Price Setting. 2021. 137 pp.
- Adrian Poignant. Gold, Coal and Iron. Essays on Industrialization and Economic Development. 2022. 214 pp.
- 203. Daniel Bougt. A Sequence of Essays on Sequences of Auctions. 2022. 188 pp.

- 204. Lillit Ottosson. From Welfare to Work. Financial Incentives, Active Labor Market Policies, and Integration Programs. 2022. 219 pp.
- 205. Yaroslav Yakymovych. Workers and Occupations in a Changing Labour Market. The Heterogeneous Effects of Mass Layoffs and Social Safety Nets. 2022. 212 pp.
- 206. Raoul van Maarseveen. Urbanization and Education. The Effect of Childhood Urban Residency on Educational Attainment. 2022. 210 pp.
- 207. Sofia Hernnäs. Automation and the Consequences of Occupational Decline. 2022. 229 pp.
- 208. Simon Ek, Structural Change, Match Quality, and Integration in the Labor Market. 2023. 176 pp.
- 209. Maximiliano Sosa Andres, A Risky and Polarized World: Essays on Uncertainty, Ideology and Foreign Policy. 2023. 226 pp.
- 210. Alice Hallman, Hypocrites, Devil's Advocates, and Bandwagoneers. Essays on Costly Signaling. 2023. 176 pp.
- 211. Edvin Hertegård, Essays on Families, Health Policy, and the Determinants of Children's Long-Term Outcomes. 2023. 236 pp.
- 212. Josefin Videnord, Trade, Innovation, and Gender. 2023. 156 pp.
- 213. Ao Fei, Essays on Taxation, Externalities, and Poverty Traps. 2023. 127 pp.
- 214. Anton Sundberg, Essays in Labor Economics. Parenthood, Immigration, and Education. 2024. 209 pp.
- 215. Gabriella Kindström, Urban Dynamics and Contemporary Challenges. Essays on Housing and Neighborhood Amenities. 2024. 161 pp.
- 216. Akib Khan, Essays in Human Capital and Immigration. 2024. 308 pp.
- 217. Erika Forsberg, Labor-market inequality. Essays on the roles of families, firms, location, and criminal records. 2024. 218 pp.